### DEVELOPMENT OF A SLURRY MONITORING SYSTEM

. .

12

.

second<sup>2</sup>

Michael Petrus Clegg

4

Å,

1.

di.

18

i.

å.

A Project Report Submitted to the Faculty of Engineering, University of the Witmatersrand, Johannesburg, in partial fulfilment of the requirements for the Degree of Master of Science

Johannesburg 1985

PREFACE

Å.

### DECLARATION

I declare that this project report is my own, unaided work. It is being submitted for the degree of Master of Science in the University of the Witwatersrand, Johannesburg. It has not been submitted before for any degree or examination in any other University.

Nelezza Michael Petrus Clegg

Twonty eight day of June 1985

Twenty-eight aldoy of June 1985

Preface

2

کہ ہے ک



## ACKNOWLEDGEMENT

The Author would like to acknowledge the following:

The University Senior Bursary, the Bernard Price Scholarship and the CSIR for funding received during the project.

UWTec, the University of the Witwatersrand Technology Centre for making the project possible.

Prof. Rodd, my supervisor, for the support and advice given throughout the duration of the project.

Niels Verkerk, for his patience, co-operation, and the use of his slurry test rig.

Krohne Measurement and Control (Pty) Ltd for offering a full mass flow system for evaluation.

Preface

5. R

ું કે વ

÷.,

( 1.,9,

### ABSTRACT

This project report can be dyided into three main sections. The original objective of the project was the development of slurry monitoring instrumentation. However, during the literature survey for the project, it was found that several manufacturers had reachly or were about to, release integrated, intelligent instrumentation which set many of the project requirements. At one of the initial project requirements was, in fact, that as much commercially available instrumentation as possible should be used, with the rememining hardware only being developed if necessary, it was dacided to evalues the presently available products rather than continue with the development of futher hardware.

A point which gross out of the literature survey ues the increasing use of nuclear based instrumentation. Many end-users still shy way from these instruments, largely due to misunderstanding, unfamiliarity, the health hazard and the associated legal requirements for their use. It are also found that faw users (and product representives) understand the principles, operation and application of unclear instrument. As muclear density gauges are a integral pert of the proposed solution, an investigation into the principles, operation, health hazard'. signil requirements and applications of nuclear based instrumentsion was understaken. The objecdevalop an understanding of how to choose and apply such instruments. haphasis was placed on gamme-ray based instruments which are prevalent in the mining and process control industries.

The report thus serves two purposes. Firstly, it is a survey and evaluation of instruments which existly the project requirements. Secondly, it is a guide to the application of magnetic flowmeters and the theory and application of muchaer density gauges in the monitoring of slurrins.

The evaluation of the recommended instrumentation, was performed on a full mass flow system, which was made available Messurement and Control (Pty) Ltd. It was found that the system met all the project requirements, but was inverse limited in floxibility.

The conclusion highlights the major points of the report, and discusses the success of the project. Some pointers are also given to possible future developments in instrumentation.

Abstract

# TABLE OF CONTENTS

1.0 BACKGROUND TO THE PROJECT	1				
	4				
1.2 Advantages of a high Density Slurry	2				
1.4 Instrumentation Description	2				
1.5 Conclusion	4				
	4				
2. B. SLUBBY CONCENTRATION MEASUREMENT	3				
2.1 Introduction	Ř				
7.2 Moisture Measurement	Ř				
2.2.1 Canacitance Method	ĕ				
2.2.7 Conductivity Method	g				
7.7.3 Microwave Method	ğ				
2.2.4 Infrared Guacing Method	10				
2.2.5 Neutron Moderation Method	11				
2.3 Concentration or Percent Solids Measurement	11				
2.3.1 Gravimetric Method	13				
2.3.2 Differential Pressure Method	13				
2.3.3 Nuclear Radiation Method	13				
2.4 Moisture conten'/ Percent solids evaluation	16				
3.0 MASS METERING EQUIPMENT	17				
3.1 Introduction	17				
3.2 Mass Flow Meter	17				
3.3 Mass Flow Computer	19				
3.4 Mass Flow Evaluation	20				
4.0 FLOW MEASUREMENT	21				
4.1 Introduction	21				
4.2 Ultrasonic Meters	21				
4.3 Magnetic Flowmeters	23				
4.3.1 Characteristics	24				
4.3.2 Sinewave or Pulsed DC Excitation	25				
4.3.3 Selection of Insulating Liner	27				
4.3.4 Selection of Electrodes	27				
4.3.5 Installation Recommendations	28				
4.4 Flowmeter Evaluation	28				
5.0 NUCLEAR RADIATION APPLICABLE TO DENSITY MEASUREMENT	30				
5.1 Introduction	30				
5.2 Atoms	30				
	31				
5.4 Nuclear Energy Levels	34				
5.5 Nuclear Disintegrations	35				
5.5.1 Gamma Decay	35				
5,5.2 Alpha Decay	31				
5.5.5 Deta decay	37				
5.5.4 Complex Decay Schemes	31				
5.6 The Radioactive Decay Law	38				
D.7 Conclusion	40				
6.0 ENERGY LOSS AND PENETRATION OF RADIATION THEOLOGY					
MATTER					
	41				
	41				
	41				

22

÷

а<sub>с</sub>

8

6.1 Introduction   6.2 Machanisms of Charged-Particle Energy Loss   6.2.1 Coulomb Interactions   6.2.2 Emission of Electromagnetic Radiation   6.3.3 Energy Loss   6.3.4 Totaping Power Due to lonization and Excitation   6.3.1 Theragens of harma and X-rays with Matter   6.3.1 The Photoelectric Effact   6.3.2 Compton Scattering or Compton Effect   6.3.3 Pair Production   6.3.4 Total Attenuation Coefficient   6.3.5 Photon Energy Absorption Coefficient   6.4.1 Types of Neutron Interactions	41 42 42 43 44 45 56 78 99
6.4.2 Neutron Cross Sections	50 50 51
7.0 GAS-FILLED DETECTORS   7.1 Introduction   7.2 Relationship Between High Voltage and Charge Collected   7.3 Different Types of Gas-filled Counters   7.3.1 Construction   7.3.2 Ionization Chambers   7.3.3.6 Geiger-Muller (GM) Counters   7.3.3.1 Operation and Quanching of the Discharge   7.3.3.2 The Pulse Shape and the Dead Time of a GM Counter   7.4 Conclusion	52 53 55 55 57 58 60 61
8.0 SCINTILLATION DETECTORS   8.1 Introduction   8.2 Inorganic (Crystal) Scintillators   8.2.1 The Mechaniam of the Scintillation Process   8.2.2 Time Dependance of Photon Emmission   8.3.3 The Photomultiplier Tube   8.3.1 General Description   8.3.2 Electron Multiplication in a Photomultiplier   8.4 Dead Time of Scintillation Counters   8.5 Sources of Backround in Scintillation Counters   8.6 Industrial Scintillation Detectors	63 64 64 66 68 68 68 68 68 70 71 71 72
9.0 STATISTICS AND ERRORS IN NUCLEAR MEASUREMENTS   9.1 Introduction   9.2 The Errors, Accuracy and Precision of Measurements   9.3 The Normal Distribution   9.4 The Poisson Distribution   9.5 The Chi-Square Test of Goodness of Fit   9.6 Criterion for Rejecting a Reading   9.7 Propagation of Errors   9.8 Dead Time Losses   9.10 The Case of the Decaying source   9.11 Conclusion	73 73 75 76 77 78 79 79 79 80

Table of Contents

÷.,

.

vii

 æ

ŝ

10.1   Introduction   2     10.2   Health Physics   2     10.3   Effect of Exposure   2     10.4   Principles of Protection   2     10.4.1   Basic Concepts   2     10.4.2   What an Employer Should Do   2     10.4.3   What an Employer Should Do   2     10.4.4   What an Supplier Should Do   2	11233455
70.5 Legal Requirements 5   10.6 Emergency Procedures 5   10.7 Conclusion 5	36 36 37
11.0 GAMMA RAY INSTRUMENTS IN AN INDUSTRIAL ENVIROMENT E   11.1 Introduction   11.2 Industrial instruments   11.3 Factors Influencing Density Measurement   11.4 The Nuclear Density Gauge   11.4.1 The Nuclear Density Gauge   11.4.2 The Nuclear Density Gauge   11.4.3 Tandogue or Digital Signal Conditioning   11.4.3 Danologue or Digital Signal Conditioning   11.4.5.1 Analogue or Digital Signal Conditioning   11.4.5.2 Digital 'Ingal Conditioning   11.4.5.3 Digital 'Ingal Conditioning   11.4.5.1 Analogue or Digital Signal Conditioning   11.4.5.2 Digital 'Ingal Conditioning   11.5 Considerations in Density Gauge Applications   11.6 Consultation in Consistional Constructions	38 38 39 31 33 33 35 30 11 23 33 35 30 11
12.0   EVALUATION OF KROHNE MASS FLOW SYSTEM   10     12.1   Introduction   11     12.2   Operating Principles   10     12.3   The Test Rig   10     12.4   Test Nethodology   10     12.5   Mass Flow System Evaluation   10     12.6   Mass Flow System Evaluation   10     12.7   Conclusion   10	13 13 13 15 17 18 19 14
13.0 CONCLUSION   11     APPENDIX A. GLOSSARY OF NUCLEAR TERMS   11	15 17
APPENDIX B. MASS ATTENUATION COEFFICIENTS 11   APPENDIX C. MASS FLOW EVALUATION TEST RESULTS. 12   C.1 Computed Results. 12   C.2 Readings for Evaluation of Magnetic Flowmater. 12   C.3 Readings for Evaluation of Nuclear Density Gauge. 13	19 20 23 35
REFERENCES	41

Table of Contents

STATES OF THE OWNER OF THE OWNER

viii

æ

. . .

.

# LIST OF ILLUSTRATIONS

Figure	1.	System Configuration
Figure	2.	System Characteristics and Requirements
Figure	3.	Neu.ron Moisture Measurement
Figure	4.	Gravimetric Density Meter
Figure	5.	Differential Pressure Counterflow Mater
Figure	б.	Micro Motion Sensor Tube
Figure	7.	Micro Motion Operating Principle
Figure	8.	Doppler Flowmeter Schematic
Figure	9.	Magnetic Flowmeter Schematic
Figure	10.	Magnetic Flowmeter Liner Characteristics
Figure	11.	Neutron-Proton Properties
Figure	12.	Bound and virtual nuclear energy levels
Figure	13.	Radioactive lastopes
Figure	14.	Maximum Energy Calculated
Figure	15.	The relative importance of the three major gamma inter-
-		actions
Figure	16.	A typical gas-filled detector
Figure	17.	The relationship between applied voltage and charge 54
Figure	18.	External quenching circuit for GM counter
Figure	19.	Dead time and recovery time for a GM counter 61
Figure	20.	Allowed and forbidden energy bands of a crystal 65
Figure	21.	Schematic diagram of photomultiplier interior 69
Figure	22.	Example of Standard Deviation Computation
Figure	23.	Krohne DH80 Block Wiring Disgram
Figure	24.	Mass Flow System Block Diagram
Figure	25.	Schematic of Test Rig
Figure	26.	Measured vs. Actual Flowrate
Figure	27.	Measured vs. Actual Specific Gravity
Figure	28.	Slurry Pressure Recordings
Figure	29.	Slurry Velocity Profile

List of Illustrations

and a subscription of the subscription of the

8. ~

1. No. 1. 1.

Į

#### 1.0 BACKGROUND TO THE PROJECT

### 1.1 GENERAL

In deep South African gold mines, the gold bearing reaf occurs as a thin sheet or table of rock which is often only 75mm to 100mm thick. Gullies are driven into the rock and the ore is then removed from stopes which cross-connect the gullies.

As the ors is mined, the stope converges under the overburden pressures (up to  $100~{\rm MPs}$  in mines of 3500m depth). A typical stope closure rate iv 100mm a month.

The prevention of closure has traditionally been controlled by means of timber supports. This system has several disadvantages :

- 1. The handling of the timber is labour intensive.
- 2. The large quantities of timber underground constitutes a fire hazard.
- Ventilation is difficult to control because of the large open spaces between the packs.

The hydraulic backfilling of stopes in deep gold mines is an attractive proposition as a means of controlling convergence of the stopes and absorbing energy released from the surrounding rock.

The advantages and benefits of using a hydraulic backfill system rather than a system of timber supports include:

- 1. Expensive timber need no longer be used.
- 2. Reduced labour requirements.
- 3. Reduced fire hazards.
- Ventilation control is facile in ed as a result of the alimination of open spaces.
- Mine safety is increased. The compression of the fill absorbs energy released from the rock in the vi inity of the filled stope.
- The problem of disposing of large quantities of mining waste can be partially alleviated by underground stowage.

The use of hydraulic stowage is not limited to gold mines. In South Africa, the large production of flysh and bottom shi nthe coal-burning power stations, provides a ready source of a fill material for backfilling of the coal mines. Flysh has pozzolanic properties and 1. has been shown that flysh/cesant mirtures can give similar shear properties to those produce by other shown in the coal shown of the set of the se

1

Background to the Project

Flyash is readily available from the power stations and can normally be obtained for only the cost of its rransport. As most of the power stations are located close to the coal mines, this cost is minimal. The use of flyash as a fill has several advantages :

- 1. Power stations need not purchase land for the formation of dumps.
- 2. Elimination of the dumps leads to a more sesthetic environment.
- The ash can provide a base for further workings in thick seams and hence increase the percentage extraction, which extends the life of the power station.
- The support to the coal pillars increases the safety factor of the mines.

A futher application of high concentration paste pupping is the reclamation of unburnt flyash from allows daws. Investigations of the chemical analysis of the flyash from one particular power plant revealed that about 45% carbon remaind unburnt.

### 1.2 ADVANTAGES OF A HIGH DENSITY SLURRY

Nydraulic backfilling has not, despite its potential applications and advantages in gold and coal mines, been fully accepted in South Africa. The reasons for this include:

- The fine gold tailings are slow draining. The stope could thus converge before the tailings have solidified.
- 2. The fill r mes retaining while still in the liquid state.
- The excess. ter that is introduced by conventional slurry techniques must be pumped back to the surface.

Research at the Materials Handling Research Unit (NHRU) of the University of the Vitaversrend has land to a proposed system, in which it is intranded to cramsport the Slurry in a high density mate form with as little as the summary of the summary of the summary of the slurry assumes a homogeneous form and minimum use is madeed as a care resource, namely

Futher advantages in applications to gole closes include :

 Less drainage is required, thus less drawergence takes place while the tailings solidify.

Background to the Project

2

- Retaining of the fill is not required, simplifying underground operstion.
- 3. There is no excess water to be pumped back to the surface.

For flyash transportation the advantages are :

- Utilization of the pozzolanic characteristic which is moisture dependant.
- 2. The average moisture content of the flyash as removed from the dem is about 45%. By pumping a high concentration parts back to the power plant the drying requirements can be minimized. (This is an important economic aspect as the drying plant of a short pipeline often makes the system uncenomeical).

### 1.3 PROJECT OBJECTIVES

In order to implement a practical high density slurry pump system in coal and gold mines, instrumentation is required to enable control of the backfill process.

The project objective is the choice and/or development of mutable instrumentation to wessure the mass flowtate and moisture content of the flyssiggoldside slorry (sing an inite mixer, the slorry is mixed with pneumatically conveyed dry coment, to strengthen the hydraulle fill. The ement/slurry mixing ratio is controlled by measuring the mass flowrate. The moisture content measurement is used to control the water content of the slurry.

The overall system configuration is shown in Figure 1 on page 4.

#### 1.4 INSTUMENTATION REQUIREMENTS

The transportation of high density slurries or "pastes" is associated with high head losses resulting from the high viscosities inherent in high concentration slurries. High pressure reniprocating pumps are used to overcome the high head losses incurred in the pasts system.

The high pressure and head losses impose three important criteria on the instrumentation.

- The instrumentation must be capable of withstanding the high pressures involved (up to 100 bar/10NPa).
- 2. The instrumentation must be abrasion resistive.

Background to the Project

- **-**

1.05

90

160

.4



#### Figure 1. System Configuration

The instrumentation must be non-intrusive to minimize abrasion and the pressure drop.

A summary of the remaining slurry characteristics and instrumentation requirements is given in Figure 2 on page 7.

#### 1.5 CONCLUSION

The basis of this project is the choice and/or devalopment of instrumentation suitable for a high density slurry system. The remainder of this report outlines the procedure undertaken to choose and avaluate the meessary instrumentation. The choice/devalopment and application of instrumentation is in many cases still much of a black art. The overiding feature of a linerrumentation is the principle upon which the instrument

Background to the Project

4

 $\sim$ 

is based. There are easy ways to obtain a measurement, but usually only one, or very few, ways suitable to a particular application. Once a suitable principle hat been found, the user can survey the market for a suitable instrument based on thet principle. If a commercial instrument is not available the user will then have to consider developing his own instrument.

If an analogy is sought, use principle represents the design and the instrument the implementation. No matter how good the implementation, a poor design will always lead to poor results.

Therefore, a successful instrumentation system is heavily dependent on the principles on which the instruments are based, and the users understanding of these principles. Because of the importance of the measuring principle, much of this report covers the different measuring principles, with the aim of providing a clear understanding thereof. Particular emphasis is placed on aspectio flyw measurement and nuclear damity measurement, which form the basis of the instrumentation suitable for a high eignsity solution; works and the solution of the solut

Chapters Two to Four covers the slurry monitoring instrumentation. Much of the material in these chapters was derived from instrumentation jourmals and manufacturers literature and as such represents the users point of view. Chapter Two consists of an extensive survay into various principles for neasuring demsity and moisture content. Chapter Three discusses techniques for measuring mass flowrate. Chapter Four discusses techniques relating to slurry flow measurement.

Chapters Five to Eleven deals with the theoretical side of nuclear physfics. The material in chapter Five to Eight was taken with slight modification from Tsoulfanidis[1983]. (This is an excellent text for an in-depth study of all agreets of nuclear physics.) The exphasis is on nuclear physics as it relates to process control instrumentation. Chapter Five reviews the concepts of atomic and nuclear physics relavant to radiation measurements. Chapter Six discusses the mechanisms by which insizing radiation interacts and loses morgs as it nows through matter. Chapter Saven discusses gas-filled detectors with scintilistic and theore sensiti in any radiation measurement. Chapter Tar relates to the bashth heards and legal regulations associated with the use of nuclear sources. Chapter Eleven discusses 'gamma ray instruments in an industrial environent. Emphasis is so their properties and dosign tredeoffs.

Chapter 12 discusses the evaluation of a full mass flow system which was obtained on loan from Krohme Measurement and Control (Pty) Ltd. A brief description is given of the system, followed by a detailed description of the test rig and methodology employed in evaluating the system. The test results are then evaluated in datail.

5

Background to the Project

đ

a start of start and start of a

Ϋ́,

6

Chapter 13 concludes the report. The major woints of the report are highlighted and the success of the project artlined. Some pointers to future developments in instrumentation are 41.0 given.

Background to the Project

No. of Concession, Name

FLYASH and GOLDSLIME SLURRY CHARACTERISTICS.

 $\sum_{i=1}^n \frac{1}{i} \frac{\partial g_i}{\partial x_i} \sum_{i=1}^n \frac{\partial g_i}{\partial x_i} \sum_{i=1}^$ 

7

Particle size	:	0 - 3 m <sub>i</sub> m,
Constituency	:	Homogeneous Paste
pH	:	Alkali - acidic
Flow Profile	;	Flat
Temperature	:	Ambient
Solid Specific Gravity	:	2,2 - 2,7
Slurry Specific Gravity	:	1,5 - 2,0
SYSTEM PUMP and PIPING.		

Pressure Range	: 0 - 100 ber (10 MPa)
Flowrate Range	: 0 - 1,0 m/s
Moisture Content Range	: 20 ~ 35.%
Pipe Diameter	; 1,25 mm
Pressure Drop	: 0 - 100 bar

INSTRUMENT REQUIREMENTS.

Intruding	Meters	:	Not allowed
Accuracy	- Flow	;	5 %
	- Moisture	;	1 %
Output	- Masa Flow	:	4 - 20 mÅ
	- Moisture Content	;	4 - 20 mÅ

# Figure 2. System Characteristics and Requirements

Background to the Project

### 2.0 SLURRY CONCENTRATION MEASUREMENT

#### 2.1 INTRODUCTION

The project objectives involve the choice and/or development of instrumentation to measure the mass flowrate and mojsture content of a high concentration slurry. The most important factor for a successful instrument is the measuring principle on which the instrument is besad.

This chapter is an extensive literature survey into the principles for measuring the moisture content of high concentration slurries. The measurement of mass flowrate is discussed in the following two chapters. The survey corsisted of reviewing instrument journels, conference papers and memufactures literature.

### 2.2 MOISTURE MEASUREMENT

An important point arising from the literature survey is the confusing use of terminology survey for the survey of the terminology is relative to three areas of interest, notant of survey in a gas, moisture in a moist bulk solid, or instruct, notant of a turry.

Several techniques are available as measuring moisture content of bulk severals. These are reviewed they could possibly be used for measurement of concentration of slurries. These techniques aither measure and the data of water directly or infer is. These techniques aither measure the material. The various techniques and their limitations are discussed below.

### 2.2.1 CAPACITANCE METHOD

This method is based on the principle that the dielectric constant of the material varies with the moisture content[Gumpert,1981]. The major problem is that the constant also varies with the mount of material be-

a



The capacitance method has a moisture measurement range of <0.01-30% with m accuracy of 0.25% and a repeatement of the standard of the standa

### 2.2.2 CONDUCTIVITY METHOD

This mathed relies on the variation of conductivity of the sample material with the variation of molisture concent. The conductivity is however dependent on several factors including density, temperature, dirt build-up on the containar walls and constituency of the material itself. The method although often proposed has not proven reliable, gives inconsistent rasults and has poor accuracy.

#### 2.2.3 MICROWAVE METHOD

This method is based on the use of microweves with frequencies from 1 to 100 GHz[hyser,1984]. The method works on the principie of comparing the affective propagation of felectromegnotic waves in a specified material, to the propagation in free space. The wetten: of modification depends on the complex value dislectric constant of the metarial. The real part of the constant determines the velocity of the propagating wave in the material while the complex part, called the loss factor, determines the attemation of the wave applitude as it punctrates the material. The real and complex components of the constant depend on the material density, in the fact, that the dislectric constant for veter is much greater than that of matrials commonly measured. The water idelectric constant therefore largely determines the effective dielectric constant of the moist material.

5 a

a

A horn extense is used to couple the microwave signals to and from the sample. The ratio of the phase shift and amplitude, which is reasonably independent of density, has a fixed relationship to the moisture content of a given material. The device can thus be calibrated to read the moisture content of the material.

The advantage of this method is that it is non-contacting, the microwaves pass right through the material, and density is autometically compensated for. The method would however be unsuitable for slurries, as it would not be feasible to mount the horn antennae onto the pipe, thus making it virtually impossible to couple the signal into the pipe. The dielectric



constant for the slurry could also be a limiting factor. The method is most suited to bulk solids.

The microwave method has a moisture measurement range of <0.01-50% with an accuracy of 0.0% and a repeatedbility of 0.03% depending on application. It is however slighty affected by density, fats, and particle size variation.

### 2.2.4 INFRARED GUAGING METHOD

The motion is based on the fact that weter exhibits discrete selective absorption banks in the near inferred region at 2,95 microns, 1,94 microns and 1,45 microns (decreasing sensitivicy) [Crebtree,1975]. In practise a filter wheel having two options interference filters is rotated in front of an energy source (balogen lemp). The two beams of energy are imaged equentially onto the surface lemph.

The wavelengths of the beams, as dotarnined by the filters, are selected such that one is at the chosen measuring wavelength for wavelength and the other at a wavelength adjacent to the absorption feature and is not absorbed by the esterial itself, say 1,8 microns.

A detector collects the energy bursts reflected back from the material surface, only one of which has been absorbed by the water in the material. A directly proportioned reading is obtained by forming the logarithm of the ratio of the two intensities.

The advantage of this method is that it is non-contacting and the use of a reference beam eliminates most errors resulting from time-drift, wear and dirt buildup.

The anglor limitation of this method is that the beams reflact off the surface of the material and do not pass through: Thus it is not suitable for any material where the water concentrian within the material differs from that on the surface. Another limitation is due to strong surface reflactions and thus it can not be used on shiny material. The factor which eliminates it from contention for slurry monitoring is that transparent windows would have to be provided in the pipeline for the beams to pass through (a pressure and abrasion limitation) and the fact that it is a backcarter method.

The same technique may also be applied to measurements of film thickness, in-situ gas analysis and trace element determination.

The infrared method has a moisture measurement range of <0.01-90% with an accuracy of 1% and a repeatability of 0.1% depending on application.

It is however affected by density, fsts, colour, particle size and temperature variation.

### 2.2.5 NEUTRON MODERATION METHOD

This technique works on the principle that the speed of fast neutrons in strongly moderated by hydrogen nuclei, while the speed is barely affected by atomic nuclei of higher atomic weight[Pulse,1983]. Thus, in the vincinity of a fast neutron source (Baryllium or Americium), a cloud of alow neutrons is formed whose density depends largely on the hydrogen content of any moisture which is prevent. By combining the fast neutron source with a slow neutron measuring system, a non-contact moisture measuring guege is obtained.

Such a system is not affected by temperature, pil-value or pressure. The sensing unit can be mounted over conveyors, on bins, or on pipelines. The sample volume should be fairly large, 400mm wide by 50mm thick as a minimum for conveyors, or 200mm minimum pipe dimenter.

A block diagram for a conveyor belt system is shown in Figure 3 on page 12.

The system does however have several limitations. The measurement is dependent on the number of heaviers and some present in the sample and thus not only water is measured, but all substances containing hydrogen (e.g. hydrocarbons). This can be comparated for if the number of non-water bonded hydrogen storms is relatively constant. The measurement is also dependent on the density of the substances and thus density compensation is required. A possible saving can however be oblieved by using the same detector for the neutrons and gamma args. A further limitation is that the source and detector are usually combined in one housing and thus the neutrons do not pass through the sample under measurement.

### 2.3 CONCENTRATION OR PERCENT SOLIDS MEASUREMENT

Apart from the conductivity and capacitance methods proposed above which can be used for measuring the concentration of slurries the remainder of the methods all rely on the measurement of the specific gravity of the slurry. Density gauges can be calibrated in percent solids, only if a two-component solirry exists where one component's density (such as water  $S\xi = 1$ ) is fixed. Some small error may be introduced if the solids' mass attenmention coefficient to density varies vides.

Using the above assumption the percent solids is calculated as follows:

CABLE TO MEASURING B RECORDING INSTRUMENTS B RECORDING INSTRUMENTS DETECTOR THERMAL NEUTRON SCRAPER PLOW DETECTOR FAST NEUTRON DETECTOR DAMMA RAY DETECTOR DETECTOR DETECTOR DAMMA RAY DETECTOR DET

Figure 3. Neutron Moisture Measurement

Percent Solids (by whight) = 5% Solids (SGU Slurry - 1) SGU Slurry (SGU Solids - 1)

The method for monauring the concentration then boils down to a method for mensuring density or specific gravity. Note that this effectively excludes the above muchods since density componention is required for most of them, whereas the concentration of a Surry can be obtained singly from the density measurement alone. As will be seen later, density measurement is also required for measurement.

Note: The terms density and specific gravity are used interchangeably and this can lead to some confusion. For clarity the terms are defined below.

Density is the mass per unit volume and has units kg/m<sup>3</sup>

Specific Gravity (SG) is is the mass ratio of equal volumes of the material and water at standard conditions and is dimensionless.

The techniques for measuring density fall into three categories, newely fundamental mass measurement, nearly direct mass measurement and informatial mass measurement. The first two are likely to give the most accurate and consistent results and the discussion will centre on these categories. It is seen that all tachniques measure the mass of the metarial. To derive the density the assumption is sliveys made that the volume is fixed and known (full contains?).

Slurry Concentration Measurement

12

\*\* \*

The various techniques are discussed below:

### 2.3.1 GRAVIMETRIC ME HOD

A diagram of this device is shown in Figure 4 on page 14. The method is very simple and consists simply of taking a saction of pipe of known volves, filled with the slurry and weighing if. The method is thus one of fundamental mass measurement. The social of pipe which is weighed is usually formed into a U-tube, but there has been a proposal to use a streight section of pipe. [Baker, 1979]

Despite its simplicity the method has several limitations. Early devices were ansatzing to pressure, and devices are normally, limited in their working pressure. Problems can also arise with the flexible couplings. The method also does not offer high accuracies and in practical implementation is not always put in-line, but, a sample of the sleary meterial stapped off and flows through the matcr. It is however to keep the velocity above the slurry setting velocity to prevent deposition of solids in the mater causing an error in the measurement.

#### 2.3.2 DIFFERENTIAL PRESSURE METHOD

An example of a seter based on this principle is shown in Figure 5 on page 15. The metre bas an inverted U'-loop in the slurry line. By measuring the pressure difference between the top and bottom of the two legs and making certain assumptions, the concentration of the slurry can be obratined[Muller,1395;3]somain,1979). The method is cons.'kreat to be useful but not expected to yield high accuracy. Unlike all the other maters in he literature survey, it is not known if there is a conserving meater based on this principle. The principle is one of nearly direct mass measurement.

#### 2.3.3 NUCLEAR RADIATION METHOD

Unlike the nuclear radiation mater which uses noutron radiation for monisture measurement (discoussed earlier), this method uses games rays and measures the density of the material in the pipe. Nuclear density meaters are rapidly being excepted as the standard for density measurement, are commonly available and give accurate and consistent results.[C and 1,1982;Springer,1979]



U-tube density meter (plan view)



#### Straight-through density meter (side view)

Figure 4. Gravimetric Density Meter

The messurement technique is as follows. A nuclear source (Ceasium-137 with 30 yr. half life, or Cobalt-60 with 5 yr. half life) and a detector (sither an ionisation chamber or scintillation counter) are placed diametrically opposite the pipe. High energy games rays are beamed through the pipe and onto the detector. When games rays are beamed through the sate absorbed in proportion to the metarial density. An increase in the sate site density results is a reduced detector output.

Thus, the measurement of the absorption of the rays by the material is a diract measure of the density of the material. For acceptable stability and elimination of excessive drift, it must be ensured that at least 5% of the gamma rays are absorbed. Conversity, if sipe diameter and density are too great, not enough radiation can penetrate the detector.

Slurry Concentration Nessurement

14

.8



#### Figure 5. Differential Pressure Counterflow Meter

For the anjority of density applications Cassium-137 is used as the radiaisotope. Source sizes normally vary batwasen 10 and 2000 mCi (milicurics) as a function of pipe diameter and specific gravity span. Gauge sensitivity is increased by the use of collimated beam geometry which restricts radiation in all directions except for a direct path to the detector. This minimizes scatter and pormits the use of larger sources with increased measuring sensitivity. Next designs are such that radiation integration and from the gauge surface in any direction, will not exceed 5 mM/h (milliruntgams par hour). This is such that radiper yeak. The source holder is provided with a shutter mechanism to close the radiation beam part during installation, when the power fails or, where required, from a remote nontrol bard.

The minimum encayuring full scale span is about 0.05 SG units with a corresponding accuracy of 0.0005 or better. When measuring small spans, the zero drift due to source decay becomes an important consideration. The source decay compensator unit is essential for such installations. For wider renges, it is essential only if the source is Gobalt-60. The source decay effect with Gassian-137 is only about 3% por year. Froceass materials with high temperature expansion coefficients are provided with temperature structer of the source of the sourc

Slurry Concentration Measurement

15

The disadvantage of a nuclear density gauge is that a parait is needed to operate it, and suitably qualified personnel must b employed to accept responsibility for the nuclear source. Some operators may also be reluctant to operate the gauge and extra precentions must be taken to minimize the health hazard. However, it must be stressed that the safety requirements are no more onerous than these needed for any damgerous material or chemical.

### 2.4 MOISTURE CONTENT/ PERCENT SOLIDS EVALUATION

It should be clear that the density maters are the best solution since not only are they often required for compensation in direct molsture measurement but are also required to calculate the solids mass flow. In choosing a density metar it is clear that the nuclear meter offres the best possibility of success. It is not affected by external affects, is accurate, commonly available and has undergone extensive development by several manufacturers.

This concludes the review of concentration and moisture content meters. The following chapter discusses the other aspect of the project, namely, techniques for measuring the mass flowrate of high concentration slurries.

Slurry Concentration Measurement

.

16

#### 3.0 MASS METERING EQUIPMENT

### 3.1 INTRODUCTION

To meet the project objectives instrumentation is required to measure the slurry moisture content and mass flowarts. This chapter discusses instrumentation to measure the mass flow rate of high concentration slurries.

Two basic techniques are used. The first technique uses coriolis forces to measure mass flow directly, and the second technique adopted is to calculate the mass flow. Driginally this was done using analogue computers but more and more manufacturers are introducing microprocessor based systems.

### 3.2 MASS FLOW METER

The Mass Flow Mator measures a very small force generated by the process fluid as it moves through the sensor tube. This force results from the acceleration or deceleration of the fluid particles as the tube vibrates perpendicular to the direction of fluw. The force is analogous to the coriolis force which causes sir currents to circulate around the rotating earth, and to gyroscopic forces amployed in navigation systems.

Meters based on this principle were initially available only in small sizes, but increasingly larger meters (up to 5 tons/hour) are becoming available. The mass flew meter is also suitable for measuring gas mass flowrates, and this has resulted in meters based on this principle becoming available from seven1 manufacturers.

The discussion in the remainder of this section is based on the Micro Motion Mass Flow Meter[Micro Motion, 1983].

The forces induced by the fluid flow on the sansor tube are the coriolis/gyroscopic forces. Figure 6 on page 18 shows the fluid with mass (m) and velocity (V) moving through the sensor which is rotating with angular velocity (W) about axis 0-0.

Forces exerted by the fluid on each leg (F1 and F2) are 180 degrees out of phase. The angular volocity of the sensor tube is not required to be constant 1: t can oscillate. The associated force is also oscillatory and is proportional to the fluid mass (m) and valocity (V). As the tube vibrates about exis 0-0, the forces or scate an oscillating moment about the axis R-R. The total moment (M) is the integral of the moments of all the resents tube. The soments H causes an angular

Mass Metering Equipment

12.9.

i de



deflection or twist of the sensor about axis R-R, which is at its maximum at the midpoint of the vibrating tube travel. (see Figure 7 on page 19).

The mess flow rate can be derived by measuring the deflection angle using the two sensors shown in Figure 7 on page 19. This measurement is accouplished by measuring the relative times that each sensor detects the midpoint crossing of the respective lag. The tume difference at zero flow is nulled by calibration of the signal conditioning electronics.

The mass flow rate (Q) is proportional only to the time interval and geometric constants and is independent of the vibration frequency of the sensor tubé. The meter is also capable of measuring reverse flow direction.

It is also possible to obtain a density output from the meter. The density output is however dependant on the spring constants and tube mass of the sensor, and thus subject to drift as metal fatigue takes place. The output is suitable only for relative density approximations.

The major limitation of using this meter in the present project is that it induces a large pressure drop which is dependent on the fluid viscosity, density and velocity.



#### SENSOR TUBE WITH AXES OF ROTATION AND MOMENT & M

Figure 6. Micro Motion Sensor Tube

Mass Metering Equipment

18

`e



### COMPLETE CYCLE OF SENSOR TUBE: MEASUREMENT OF TWIST ANGLE O



#### 3.3 MASS FLOW COMPUTER

The mass flow is derived from the product of the volumetric flow rate, the material density and the solids concentration. The product of the volumetric flow rate and density give the total mass flow rate. Multiplying this by the percent solids (concentration) gives the mass flow rate of the solids, i.e.

SF = F.PS.SG

Where

F = vclume flow of slurry in m<sup>3</sup>/hr (from flowmeter)

PS = percent solids (concentration) by weight

SG = slurry specific gravity

The volumetric flow rate is obtained from a flowmeter, while the density is obtained from a density gauge and the arctant solids from a contentration meter. It was shown in the review of concentration measurement in section "Concentration or Percent Solids Measurement" on page 11 that

Mass Metering Equipment

19

. مرجع



a density gauge can be used to measure both the specific gravity and the concentration of a slurry.

The mass flow computer (usually a stand alone unit) has an input for volumetric flow rate, SS and percent solids, and provides signal outputs of both solids concentration and mass flow rete.

#### 3.4 MASS FLOW EVALUATION

One of the main constraints on the instrumentation is that it must not introduce a pressure drop. Thus, it is not possible to use a Nease Plaw Neter. Therefore, the mass fire is calculated by means of a mass flow computer. This has advantages as a density gauge is already required to measure the slurry concentration. In this project a nuclear density gauge is used to determine the density and slurry concentration.

To complete the mass flow calculation, the volumetric flowrate must be determined. Different flow measuring principles are discussed in the next chapter.

Mass Metering Equipment

, d

der.

p.

### 4.0 FLOW MEASUREMENT

#### 4.1 INTRODUCTION

In the previous chapter it was shown that a mass flow computer can be used to calculate the mass flowrate of a slurry based on the SG, volumetric flowrate and solids concentration. This chapter discusses techniques for measuring the volumetric flowrate.

Although there are several methods for measuring flow, such as orifice plates, vortex meters and turbine meters, only two meters warrant serious consideration for the measurement of slurries. These are ultrasonic meters and meganetic flowmeters. In general, flowmaters measure the velocity of the fluid in a pipe. The volumetric flowrate is determined from the pipe disensions and the assumption that the pipe is foll.

#### 4.2 ULTRASONIC METERS

Ultrasonic maters [Merrit,1982;Mintek,1983;] are usually based on one of two principles, measurement of transit time or measurement of doppler frequency shift. Of the two, doppler is the most commonly used and only doppler meters will be discussed.

When doppler meters were introduced, they sere heralded as the first ideal and universal flowmeters. The meters were non-intrusive and could in fact be simply clamped onto the pipe conveying the material. The meters did, however, not live up to their promise, largely due to poor quality and, possibly, incorrect application.

More recently the quality of the meters has improved, and the meters are being recommended in instrumentation journals for measuring the flowrate of slurries.

The prime advantage of such instruments, are, that they are non-intrusive, - they can also be fixed onto pipework without interrupting the process. Their cost does not rise with pipe diameter and they can also seasusilurise with magnetic proporties. It is further claimed that they are inexpensive, accurate and when properly applied, give reproducible results.

The meter operates by transmitting an ultrasonic beam through the wall of the pipe. The beam is backsatured, or reflected from particles in the slurry or ore, or from air bubbles. If these particles are moving their velocities are added to (or subtracted from) the velocity of the reflected ultrasonic beam in the medium. The resulting change in frequency between the incident and reflected beam is known as the Doppler shift, and is directly proportional to the velocity of the particles. A combi-

Flow measurement

b,

nation of the doppler equation and Snell's law shows that the change in frequency is :

Frequency change =  $f1-f2 = (2V, \cos x)/C$ 

Where:

f1 and f2 are the transmitted and reflected beams respective frequencies;

 $\boldsymbol{x}$  is the angle between the axis of the pipe and the incident ultrasonic beam;

C is the velocity of sound in the medium; and

 $\boldsymbol{V}$  is the velocity of the reflecting particles, and is much smaller than  $\boldsymbol{C}.$ 

The mode of operation is shown in Figure 8.



Figure 8. Dopplar Flowmeter Schematic

For transmission of the ultrasonic beam, there must be an acoustic bond between the transducer and the pipe, and the vall of the pipe must transmit the ultrasound into the sivery. Further, since Rayloigh beckscattering depends on the the six of the particles (a relation to the third power) the slurry must contain particles of a cartain size and in a certain contentration.

Flow measurement

J



In order to evaluate doppler meters, investigations were undertaken by the Council for Mineral Technology (Mintek) using six different doppler electrs. The tests were conducted on mine-sand Slurry at three relative densities : 1,040, 1,267 and 1,271 with flow rates over the range 1,5 m/s to 6 m/s.

The enjor results of the tests revealed that all flowmeters responded well to charges in flow and gave linear outputs to within 2 5 of the maximum. The time constants for dopplar meters is longer than that of magnatic flowmeters and that the slope of the response decreases as the density of the slurry increases (thought to be due to the velocity profile). (Mintek,1964)

### 4.3 MAGNETIC FLOWMETERS

Magnetic flowmaters ware specifically designed for the measurement of flowrates in slurries[Marks,1977;Springer,1980]. The magnetic flowmater is based on Faraday's law of electromegnetic induction. As the material flows through the magnetic field, which is sate up by axciting the mater's electromagnets, an 'lectric voltage, which is directly proportional to the liquid's velocity, is induced in the liquid which acts as the electrical conductor in.

U = B.D.v.

Where:

- U = The induced voltage (V)
- B = The magnetic field strength (Tesla)
- D = The distance between the electrodes (m)
- v = The velocity of the material (m/s)

The operation is shown in Figure 9 on page 24.

The induced voltage is perpendicular to both the conductive liquid and the magnetic field as tup by the mater's coils. The voltage is amesured for subsequent transmission to an applicable converter. The flux density of the magnetic field and the distance between the electrodes are constant; therefore the induced voltage is solarly a function of the material's velocity. The induced voltage is not affected by temperature, viscosity, or conductivity as long as the conductivity of the measured liquid is above a minimum threshold level (min. 0,05 microSiemens/cm). For reliable measurements, the pipe must be completely full of liquid. This is necessary as the electrode voltage is a weighted function of the induced voltage throughout the cross-section of the pipe. Secondly, the

Flow measurement



#### Figure 9. Magnetic Flowmeter Schematic

meter output is often taken as a volumetric flowrate which is only valid if the pipe is full.

To prevent the voltage being shorted out through the meter walls, the meter must be lined with an insulating lining. The magnetic flowmeter is available with different linings, different electrodes (including capacitive pick-up) and with either AC or DC exclision.

### 4.3.1 CHARACTERISTICS

The magnetic flowmeter has the following advantages:

- Linear indication of average flow velocity, thus ensuring superior accuracy to differential pressure meters.
- The meter is non-intrusive, therefore wear and prossure loss in the meter are the same as an equivalent length of straight pipe.
- The metering is independent of the nature of the flow (turbulent or leminar).
- The measurement is independant of temperature, pressure, viscosity, density, concentration, and direction of flow.

Flow measurement

s

- There is no preferred flow direction and even pulsating flows can be measured.
- 6. The primary head (detector) can be mounted in any position.
- 7. The method is independent of undissolved solids in the medium.
- The method can be used to measure chemically corrosive liquids by suitable choice of tube liners and electrodes.
- The output signal is a voltage which is readily converted to any desired electrical quantity.
- 10. Accuracy of up to 0.5% of rate down to 10% of full scale and thereafter an accuracy of 0.5% of full scale. The percent-of-rate accuracy means the meter remains accurate at low flow rates.
- A turn-down ratio (i.e. maximum to minimum flowrate) of 10 to 1 in a span of 40 to 1.

The disadvantages of the meter are:

- Pressure and temperature limitations imposed by the liner and electrode material.
- The fluid must have sufficient conductivity, i.e. typically greater than 20 µS/cm.
- 3. The fluid must not contain magnetic materials.
- 4. The cost of the meter is high.

#### 4.3.2 SINEWAVE OR PULSED DC EXCITATION

In order for a magnetic flowmetor to operate, a magnetic field must be established. The field is established by coils mounted on the meter body. An AC or pulsed DC voltage is used to excite the coils.

Sinewave or AC excitation of magnetic flowmoturs has been in use for more than twenty years. This method was originally used because of its manufacturing simplicity. The method how/var has several disadvantages: (Marks,1977)Springer,1980]

- AC meters have a high power consumption which results from the reactance of the magnetic fiels coils. The maters are thus large and bulky.
- 2. AC meters have a quadrature voltage component induced onto the electrodes as a result of transformer coupling between the excitation coils and the path formed by a line joining the electrodes and the signal cables massning from the electrodes. This voltage is 90 degrees out of phase with the velocity induced voltage. The measured

Flow measurement

25

signal is thus the vector sum of the desired signal and the quadrature component. Elimination of this signal is required.

- 3. If the conductivity of the fluid is not constant along the cross-section and along the length of the primary head, eddy currents result which vary with time. Therefore, interference voltages arise which cannot be fully suppressed in the meter.
- 4. Fouling of the electrodes increases the internal resistance of the signal circuit and the capacitive coupling between the magnetic coils and the signal circuitry. Futhermore, the signal circuitry becomes more sensitive to capacitive transmission by interference from nearby power lines.
- Interference voltages may be coupled into the signal circuit from nearby cables carrying high currents.
- 6. Occasionaly, such as in buried pipelines near railway networks, stray currents of unknown phase orientation are carried by the liquid flowing in the pipe and by the pipe itself. These currents can give rise to interference voltages.
- Interference voltages can be produced as a result of inadequate earthing which allows stray currents to induce voltages in the meter body.
- In AC magnetic-inductive flow measurement, interference voltages can only be eliminated with great difficulty if at all.
- The AC meter requires zero point setting which is done during calibration.

The pulsed DC magnetic flowmeter was developed to overcome the liditations of the AC excited mater. A close examination will reveal that almost every source of error in the AC meter is a result of magnetic (transformer) or capacaritym coupling resulting from the use of AC fields. To eliminate the AF sources of error the DC flowmeter was developed. The use of pulsed DC has two gavantages spart from eliminating problems resulting from AC excitation:

- The periodic reversal of the field (typically 6 to 11 times a second) prevents polarization of the material in the pipe and reduces buildup on the electrodes.
- Every time the field is reversed, the signal voltage is measured at the zero crossing point, thus automatic zero compensation takes place at twice the pulse rate.

A futher advantage of the pulsed DC meter, is that many meters now incorporate a microprocessor, which allows for utpile satting of span, range and output scaling, usually by means of BCD switches. This results from the fact that all signal conditioning is performed mathematically by the microprocessor. To change the range only the constants in the scaling formula has to be changed. A meter using an analogue computer (operational mapifisms) would require reachibration of the scaling potentiometers.

Flow measurement



NOTE Despite their advantages, pulsed DC meters are no longer specified for slurry applications! After, their introduction users noted inadequate performance in slurry applications. The reason most often cited is, that as the slurry particles collide with the electrodes they induce noise which results in incorrect readings. This has been attributed to poor electrode design and at least one emenfactures will still specify pulsed DC meters for slurry applications. The induced noise is not a problem in &C meters because of the higher effective sampling rate.

### 4.3.3 SELECTION OF INSULATING LINER

An important point in the selection of magnetic flowmetars is the selection of the liner. This wards with every application and determines the temperature and pressure rating of the meter as well as its resistance to corrosion and abrasication and and the temperature characteristics of the liner material. As a general guideline the characteristics shown in Figure 10 apply. (Springer: 1960)

Liner Type Ch	Corrosion aracteristic	Abrasion Characteristic	Temperature Nax.(Celsius)	Pressure Max. (NPa)
Hard Rubber	Fair - Excellent	Fair	90	10
Soft Rubber	Varies	Very Good	40	6,4
Neoprene	Varies	Excellent	80	25
Polyurethane	Varies	Excellent	40	25
Teflon (PTFE)	Excellent	Fair	180	4

#### 4.3.4 SELECTION OF ELECTRODES

Figure 10. Megnetic Flowmeter Liner Characteristics

Several types of slectrodes are available, differing mainly in derivation resistance. Hany of the electrodes are a satural several available turar development. In normal circumstances the standard electrodes supplied with the standard electrodes of the standard electrodes sup-

Flow measurement

27

2

Z

Normally, given the development of the pulsed DC meter, this meter should always be chosen above that of the AC meter. Its advantages, simplicity and ease of use, make it the overiding choice wherever applicable.

NOTE Despite their advantages, pulsad DC meters are no longer specified for slurry applicational After, their introduction users noted inadequate performance ir slurry applications. The reason most oftem cited is, that as the slurry particles collide with the electrodes they induce noise which results in incorrect readings. This has been attributed to poor electrode design and at least one manufactures will still specify pulsed DC exters for slurry applications. The induced noise is not a problem in AC meters because of the higher effective sampling rate.

### 4.3.3 SELECTION OF INSULATING LINER

An important point in the selection of magnetic flowmaters is the selection of the liner. This naturally varies with every application and determines the temporature and pressures rating of the matur as well as its resistence to corrosion and abrasion. These ratings result from the characteristics of the liner material. As a general guideline the characteristics shown in Figure 10 apply. [Springer, 1980]

Liner Type G	Corrosion haracteristic	Abrasion Characteristic	Temperature Max.(Celsius)	Pressure Max. (MPs)
Hard Rubber	Fair - Excellent	Fair	90	10
Soft Rubber	Varies	Very Good	40	6,4
Neoprene	Varies	Excellent	80	25
Polyurethane	Varies	Excellent	40	25
Teflon (PTFE)	Excellent	Fair	180	4

Figure 10. Magnetic Flowmeter Liner Characteristics

### 4.3.4 SELECTION OF ELECTRODES

Several types of alectrodes are avsileble, differing mainly in abresion resistance. Many of the alectrodes are a natural avolution of manufacturers development. In normal circumstances the standard electrodes supbied with the meter will be sufficient. For axtreme circumstances of

Flow measurement
electrode fouling, ultrasonic cleaning of electrodes is available as an option, or removable electrodes can be obtained. A recent development is the use of capacitive pickup, where the electrodes forming the capacitor plates are imbedded bahind the liner and thus cannot be fouled. Capacitive electrodes glass elev operation with liquids of conductivities is low as 5  $\mu$ S/cm. This is possible as a result of a larger electrode area. Capacitive electrodes are however exponsive and the situation max warrant their use. Available electrode materials include Hastelloy C, Stainles Steel, Titanium and Platinum-Tridium.

### 4.3.5 INSTALLATION RECOMMENDATIONS

In order for a magnetic flowmeter to operate correctly it must be correctly installed. Incorrect installation can lead to inscentrate readings and/or unreliable operation. When installing magnetic flowmeters the following guidelines should be followed to ensure reliable operation.

- The flowmeter can be installed in any position (electrode axis approximately horizontal).
- The pipe must be filled at all times as the output, is often taken to be the volumetric flowrate (velocity times cross-sectional area), which assumes a constant cross-sectional area.
- Inlet/outlet section must be greater than or equal to 5 times (for inlet) or 2 times (for outlet) the diameter of the meter measured from the electrode axis.
- Slide valves or control flaps must not be mounted directly on the primary head. (See point 3)
- Shut-off and control valves should be fitted downstream of the primary head.
- 6. For AC meters a shut-valve must be installed for zero-point setting.
- Signal transmission cables must be as short as possible and should not run alongside heavy-current lines so as to minimise interference.
- The meter must be correctly and adequately grounded to prevent interference of the signals. (See manufacturers specifications).

## 4.4 FLOWMETER EVALUATION

The dopplar and the magnetic flowmater both have their respective advantages. The main advantage of the dopplar meter is that it is a clamp-on device and thus is easy to install and use. However, the magnetic flowmater is a proven device and is widely used. The main argument egainst the dopplar mater is its high probability of failure in this application.

Flow measurement

28

Ġ2

Bespite the Hints, study the suitability of the doppler meter to high density (passe) survise and their associated low velocitiss (a system charactoristic) is still unknown. The major disadvantage of the magnetic flowmeter is that the required pressure rating makes it a non-stock item, thus increasing both it's cost and availability. Despite this, the magnucie flowmeter is considered to be the most suitable meter for the project. An AG flowmeter must be used until pulsed LC meters have proved to be reliable in slurry application.

This concludes the review of instrumentation which meets the requirements for mass flowrate and concentration measurement of high concentration pasts slurries.

The instrumentation can be summarised as follows:

Mass Flow Measurement

Nagnetic Flowmeter - For measuring volumetric flowrate

Nuclear Density Gauge - For measuring SG

Mass Flow Calculator - For calculating mass flow

Moisture Content Measurement

Nuclear Density Gauge - For measuring concentration

The same nuclear density gauge is used for both mass flow and concontration measurement. However, many end-users still shy away from nuclear instruments, largely due to misunderstanding, unfamiliarity, the health hazard and the associated legal requirements for their use. It was also found that few users (and product representives) understand the principles, operation and application of muclear instruments. As nuclear density gauges are a integral part of the proposed solution, an investigation into the principles, operation, health hazards, legal requirements and applications of nuclear based instrumentation was undertaken.

The objective of this study, which is presented in the following chapters, was to review nuclear based instrumentation and to develop an understanding of how to choose and apply such instruments.

Flow measurement

ā,

ø

# 5.0 NUCLEAR RADIATION APPLICABLE TO DENSITY MEASUREMENT

# 5.1 INTRODUCTION

This chapter reviews the concepts of stonic and nuclear physics relavant to radiation measurements. The field is however a very broad one and the exphasis is on the concepts relating to process control applications. The intention of this chapter is not to present a detailed discussion of the subjects, but rather to outline the principles inherent is nuclear process control instrumentation. The emphasis is on the nuclear behaviour and theory relevant to level, moisture and density measurement. For an in-depth discussion of the theory, principles and spectrum of nuclear bysiss the reader is referred to the Fourlandis[1963].

# 5.2 ATOMS

To the best of our knowledge boday, every stre unsists of a central positively charged nucleus around which negative intertons revolve in stable orbits. Considered as a sphere, the stom hrs a redius of the order of 0.1 mm and the nucleus a radius of the order of positive charges of the nucleus; thus the stom is electrically neutral in the normal state.

The number of positive elementary charges in the nucleus is called the actoric number and is indicated by 2. The attack number identifies the chemical element. All atoms of an element have the same chemical properties.

The stomic electrons move around the nucleus as a result of the attractive electrostatic Coulomb force batween the positive nucleus and the negative charge of the electron. According to classical electrodynamics, the revolving electrons ought to continuously radiate part of their energy, follow a spiral orbit, and eventually be ceptured by the nucleus. Obviously, this does not heppen: atoms exist and are stable. Therefore, a new theory governing orbital mechanics and called quantum mechanics was developed.

The available experimental evidence points toward the following facts regarding the motion of atomic electrons:

Nuclear Radiation Applicable to Density Messurement

- Bound stomic electrons revolve around the nucleus in stable orbits without radisting energy. Every orbit corresponds to a certain electron energy and is called an energy state.
- Only certain orbits (only certain energies) are allowed. That is, the energy states of the bound electrons form a discrete spectrum. This phenomenon is called quantization.
- 3. If an electron moves from an orbit (state) of energy  $\mathbf{E}_{i}$  to a state of lower energy  $\mathbf{E}_{f}$ , then (and only then) electromagnetic radiation, an x-ray, is emitted with frequency w such that

 $v \approx (E_z - E_z)/h \qquad (5.1)$ 

where h is Planck's constant.

The energy of the x-ray depends on the atomic number:

$$E_X = h_U = E_z - E_z = k(Z-a)^2$$
 (5.2)

where k and s are constants. Every atom emits characteristic x-rays with discrete energies that identify the stom like fingerprints. X-ray energies range from a few eV for the light elements to a few hundreds of keV for the heaviest elements.

- 4. A bound electron may receive energy and move from a state of energy  $E_1$  to a state of higher energy  $E_2$ . This phenomemon is called excitation of the state. An excited atom moves preferentially to the lowest possible energy state, therefore in times of the order of 10 ns, the electron valit recurs to  $E_1$  and sn x-ray will be emetted.
- 5. An atomic electron may receive enough energy to leave the etom and brome a free entricle. This phenomenon is called ionization, and the positive entry left behind is called an ion. The energy necessary to cause ionization is called the ionization potential. The ionization potential is not the same for all electrons of the same stom because the electrons move at different distances from the nucleurs. The closer the electron is to the nucleus, the more tightly bound it is and the greater its joinstein optential becomes.
- 6. When two or more atoms join togethar to four a molecule, their common electrons are bound to the molecule. The energy spectrum of a molecule is also discrete but more complicated than that of an etca. The discrete and unique spectrum of molecules and atoms is used in X-ray Fluorescence to identify the substance.

### 5.3 NUCLEI

At the present time, all experimental evidence indicates that nuclei consist of neutrons and protons, which are particles collectively known as nucleons. Some of the properties of a neutron, proton and an electron are listed in Figure 11 on page 32 for comparison.

Nuclear Radiation Applicable to Density Measurement

Electron Neutron Froton Rest mass 9.109558x10<sup>-31</sup> 1.672614x10<sup>-27</sup> 1.674919x10<sup>-27</sup> = 939,549 ≈ 0.511 = 938,256 MeV = 0.000549 = 1.008665= 1.007276u Charge n +0 -0

Figure 11. Neutron-Proton Properties.

A Nucleus consists of A particles,

```
A = N + Z
```

where ass number A mess number N = number of neutrons A = number of protons = element stomic number A nuclear species X is indicated as A x Z

where X = chemical symbol of the element.

The following definitions apply:

- 1. Isobars are nuclides that have the same A.
- Isotopes are nuclides that have the same 2. They are nuclides of the same chemical element. They have the same chemical but slightly different physical properties, due to their difference in mass. The nuclear properties chemne drastically from isotope to isotope.
- Isotones are nuclides that have the same N, i.e. same number of neutrons.
- 4. Isomers are two different energy states of the same nucleus.

The different atomic species are the result of different combinations of one type of particle - the electron. There are 32 antural element. Since 1940, 15 more have been artificially produced for a total of 107. The different nuclides, on the other head, are the result of different combinations of two kinds of particles, neutrons and protons, and so there are many more possibilities. There are more than 700 known nuclides.

Nuclear Radiation Applicable to Density Measurement

Experiments have determined that nuclei are almost spherical, with a volume proportional to the mass number (Å), thus all nuclei have similar densities, and an average radius approximately equal to

$$R = 1.3 \times 10^{-15} A^{1/3}$$
 in meters, (5.3)

The mess of the nucleus with mass number A and stomic number Z, indicated as Mn(A,Z), is equal to

$$Mn(A,Z) = Z,Mp + N,Mn - B(A,Z)c^{2}$$
(5.4)

where Mp = proton mass Mn = neutron mass B(A,Z) = binding energy

The unit used for the measurement of nuclear mass is equal to 1/12 th of the mass of the Carbon-12 isotope. Its symbol is u (formerly amu for atomic mass unit):

lu=(1/12) (mass of <sup>12</sup>C) = 1.660531×10<sup>-27</sup> kg = 931.478 MeV

In many experiments, what is normally measured is the atomic mass and not the nuclear mass. The atomic mass is obtained by adding the mass of all the atomic electrons. The mass may be given in one of the following three ways:

## 1. Units of u.

- Kilograms.
- Energy units (MeV or J, in view of the equivalence of mass and energy).

The meaning of B(A,Z) may be expressed in two equivalent ways:

- The binding energy B(A,Z) of a nucleus is equal to the mass transformed into energy when the Z protons and the N=A-Z neutrons combined to form the nucleus. An amount of energy equal to the binding energy was released when the nucleus was formed. Or
- 2. The binding energy  $B(\Lambda,Z)$  is equal to the energy necessary to break the nucleus apart into its constituents, Z free protons and N free neutrons.

The energy necessary to remove one particle from the nucleus is the separation or binding energy for that particle for that particular nuclide. A "particle" may be enautron, proton, an alpha particle, a deutron, etc. The separation or binding energy of a nuclear particle is analogous to the ionization potential of an electron. If a particle enters the nucleus, an amount of energy equal to its separation energy is released. The separation or binding energy is equal to the difference in binding energy is equilated by the original nucleus and the resulting nucleus after the "particle" has been removed.

Nuclear Radiation Applicable to Density Measurement

## 5.4 NUCLEAR ENERGY LEVELS

Neutrons and protons are held togethar in the nucleus by nuclear forces. Although the exact nature of nuclear forces is not known, scientistic have successfully predicted many characteristics of nuclear behaviour by assuming a certain form for the force and constructing nuclear models based on that form. The success of these models is measured by how well their predicted results agree with the experiment. Many nuclear models have been proposed, each of them explaining certain features of the nucleus; but as of today, no model exists that explains all the facts about all the Known nuclides.

All muclear models assume that the nucleus like the atom, can exist only in certain discrete energy states. Depending on the model, the energy states may be assigned to the nucleons -neutrons and protons- or the nucleus as a whole. The present discussion of nuclear energy levels will be based on the second approach.

The lowest possible energy state of a nucleus is called the ground state, in Figure 12 on page 35, the ground state is shown as having negative energy to indicate a bound state. The ground state and all the excited states below the zero energy level are called the bound states. If the nucleus finds itself in any of the bound states, it de-excites after a time of the irder of 1 to 100 ps by dropping to a lower state. De-excitation is accompanied by the emission of a photon with energy equal to the difference between the initial and final states. Energy tates located above the zero energy level are called virtual energy levels. If the nucleus obtains enorgh energy to be raised to a virtual level, it may de-excite sither by ialling to one of the bound levels or by emitting a nucleon.

Studies of the energy levels of all the known nuclides reveal the following:

- The distance botween nuclear energy levels is of the order of keV to Mev. By contrast, the distance between stomic energy levels is of the order of eV. Thus photon emission from atomic de-excitation (X-rays) are of a much lower frequency than that of nuclear de-excitation (gamma-rays).
- The distance between energy lovels decreases as the excitation energy decreases. For very high excitation energies, the density of the levels becomes so high that it is difficult to distinguish individual energy levels.
- As the mass number A increases, the number of levels increases; i.e. heavier nuclei have more energy levels than lighter nuclei (in general - there may be exceptions).
- As A increases, the energy of the first excited state decreases (again, in general - exceptions exist).

Nuclear Radiation Applicable to Density Measurement

34

....





# 5.5 NUCLEAR DISINTEGRATIONS

This section outlines the three main modes of "dcay: alpha, beta and gamma, and the variations which can occur. The type of docay of relavance in process control instruments is gamma dcay and the emphasis will be on this mode. Some naturally radioactive elements have more than one dcay mode as shown in Figure 13 on page 36 (PDI.1983). The most commonly used elements in process control are Cohalt-60, Averticiar-221, and Caesium-137. The thickness gauge applications are prevalent in the paper and pulp industry.

## 5.5.1 GAMMA DECAY

In gamma decay, a nucleus goes from an excited state to a state of lower energy and the energy difference bouwsen the two states is released in the form of a photon (an electromagnetic wave). Gamma docay is represented by

 $^{\Lambda}x^{*} \rightarrow ^{\Lambda}x + x$ 

where the \* indicates an excited nucleus.

Nuclear Radiation Applicable to Density Measurement

Radiosotope	radia- tion type	radiation energy (NeV)	n Half life (years)	Application	Typical source activities (millicuries)
Iron 55	X-ray	0.006	27	thickness gauges	50 - 100
Nickel 63	Beta	0.066	100	thickness gauges	100
Krypton 85	Beta	0.158 0.672	10.73	thickness gauges	10 - 2000
Promethium 147	beta	0.103 .	2.62	thickness gauges	50 - 400
Strontium 90	Beta	2.274	28.6	thickness gauges	10 - 100
Thallium 204	Beta	0.763	3.78	thickness gauges	20 - 200
Americium 241	Alpha	5.5	433	smoke detectors	few micro
Can-1- 193	Gamma	0.060	20 17	dencine services	500
Caesium 157	Gamua	0.002	50.17	density gauges	100 - 100
CODAIT OU	Gamma	1.33)	5.21	level gauges	100
Polonium 210	Alpha	5.305	138 days	static eliminator	Up to 1000
Carbon 14	Beta	0.15	5760	radiography	200
Raduim 226	Beta	0.227		thickness gauges	10 micro
	Gamma	4.761	1600	chickness gauges	200
	Alpha	4,598		smoke detectors	0.5 micro

# Figure 13. Radioactive Isotopes.

 $A^{i}$ 

The second se

Sometimes the excitation energy of the nucleus is given to an atomic electron instead of being released in the form of a photon. This type of nuclear transition is called an internal conversion (IC), and the ejected atomic electron is called an internal conversion electron.

When internal conversion occurs, there is a probability that an alectron from the K shell. L shell, or another shell, may be emitted. Therefore, a nucleus that undergoes internal conversion is a source of groups of mono-energetic electrons.

The total probability for gamma decay is the sum of t., probability for internal conversion doesy  $(\lambda_g)$  and the probability a photon will be emitted  $(\lambda_p)$ .

For most nuclei  $\lambda_{a}^{=0}$ , but there is no gamma-decaying nucleus for which  $\lambda_{a}^{=0}$ . This means radioisotopes that internally convert, emit gammas, electrons form dety-rays. The transfill from an elactron from a higher shall that "falls into" the shell vacated by the internal conversion electron.

Three frequently used isotopes that undergo internal conversion are  ${\rm Sn}{}^{-113},$  Cs-137 and Bi-207. These isotopes are very useful for instrument calibration.

Nuclear Radiation Applicable to Density Measurement

# 5.5.2 ALPHA DECAY

Alpha particles are Helium nuclei, i.e. Helium atoms stripped of their electrons. Alpha decay is represented by the equation

 $\begin{array}{c} A_X \rightarrow A^{-4}X + {}^{4}He \\ Z & Z^{-2} & 2 \end{array}$ 

Note that after the alpha particle is esmitted, the parent nucleus is transformed into a new idenent (daughter nucleus). In meny cases, the daughter nucleus is left in an excited state and alpha decay is thus often accompanied by gamma decay.

## 5.5.3 BETA DECAY

In bats decay, a nucleus emite an electron or a positron and is transformed into a new alement. In addition to the electron or positron, a neutral particle with rest mass zero (neutrino) is also emitted. There are two types of bets decay,  $\beta^{*}$  and  $\beta^{+}$ . In  $\beta^{-}$  an electron is emitted while in  $\beta^{\dagger}$  a positron is emitted. A free positron will eventually combine with an atomic electron, the two annihilate and two 0.511 MeV ennihilation games are emitted.

Beta particles are emitted with a continuous range of velocities, from zero up to a maximium which depends on the nucleus. The decay spectrum for beta decay is thus continuous.

In some cases it is possible for the nucleus to capture an electron and in this case a neutrino is emitted. This transformation is known as Electron Capture. When an electron is captured a vacancy is left which may be filled by an electron from a higher energy state and a x-ray will be emitted.

### 5.5.4 COMPLEX DECAY SCHEMES

In many nuclei more than one mode of decay bip tosifive. For information about particles emitted, active the state of the

Nuclear Radiation Applicable to Density Measurement

# 5.6 THE RADIOACTIVE DECAY ! AW

Radioactive decay is the spontaneous change of a nucleus. The change may result in a new nuclide or simply a change in the energy of the nuclide. If there is a cartain amount of radiolsotops on hend, there is no certainty that in the next second "so many nuclei will decay" or "none will docay". One can talk of the probability that a nucleus will decay in a cartain time.

The probability that a given nucleus will decay per unit time is called the decay constant and is indicated by the letter  $\lambda$ . For a certain species,  $\lambda$  is

1. The same for all nuclei.

2. Constant, independent of the number of nuclei present.

3. Independant of the age of the nucleus.

Consider a certain mass m of a certain radioisotope with decay constant  $\lambda.$  The number of atoms (or nuclei) in the mass m is equal to

N = m.Na/A

(5.5)

 $\mathcal{Q}'$ 

÷ 2

Grand and a start start strains

Where  $Na = 6.023 \times 10^{23} = Avogadro's number A = atomic weight of the isotops$ 

This number of atoms decreases with time, due to the decay according to

Decrease per unit time = decay per unit time

or mathematically

 $\frac{-dN(t)}{dt} = \lambda N(t)$ (5.6)

The solution of this equation is

 $N(t) = N(0)e^{-\lambda t}$ (5.7)

where N(0) = number of atoms at t=0.

The average lifetime t of the nucleus is given by

 $t = 1/\lambda$  (5.8)

One concept used extensively with radioisotopes is the half life T, defined as the time it takes for half of a certain number of nuclei to decay. Thus,

Nuclear Radiation Applicable to Density Measurement

 $\frac{N(T)}{N(0)} = \frac{1}{2} = e^{-\lambda T}$ (5.9)

which then gives the relationship between  $\lambda$  and T

$$T = \ln 2/\lambda$$
 (5.10)

For a sample of N(t) nuclei at time t, each having decay constant  $\lambda$ , the expected number of nuclei decaying per unit time is

 $A(t) = \lambda N(t)$ (5.11)

where A(t) = activity of the sample at time t.

The units of activity are the Becquerel (Bq), equal to 1 decay/s, or the Curie (Ci) equal to ?  $7 \times 10^{18} Bq.$  The Bequerel is the SI unit defined in 1977.

The term specific activity is used frequently. It may have one of the two following meanings.

1. For solids:

SA = <u>activity</u> (Bq/kg or Ci/g) mass

2. For gases or liquids:

SA = <u>activity</u> (Bq/m<sup>3</sup> or Ci/cm<sup>3</sup>) volume

There are isotopes that decay in more than one mode. The total probability of decay is then the gun of the individual probabilities of decay for each mode. The term partial half-life is sometimes used to indicate a different decay mode.

Sometimes the daughter of a radioactive nucleus may also be radioactive and decay to a third radioactive nucleus. Thus a radioactive chain is generated. A well known example is that of Uranium-238, which through

combined alphe and beta decays ends up as an isotope of lead. The general equation giving the number of atoms of the ith isotope at time t in terms of the decay constants of all the other isotopes in the chain was developed by Gsteman[1510]. This equation is not important in this discussion as the isotopes ancountered in process control do not form a series.

Nuclear Radiation Applicable to Density Measurement

39

. 66

1.5

'o

# 5.7 CONCLUSION

sť.

This chapter discussed the basic theory of nuclear physics and the different types of radiation. In order to fully understand how radiation can be utilised, the following chapter discusses energy loss and penetration of radiation. A knowledge of how different materials absorb radiation ellows certain properties of that material to be doubeed.

Nuclear Radiation Applicable to Density Measurement

40

÷Ö;

1

0

Ē,

6.0 ENERGY LOSS AND PENETRATION OF RADIATION THROUGH MATTER

## 6.1 INTRODUCTION

This chapter discusses the eachemisms by which iorizing radiation interacts, and hence loses energy as it moves through matter. The study of this subject is extremely important for radiation measurements, because the detection of radiation is based on its interactions with, and energy deposited in, the material of which the detector is made. Therefore in order to build (or understand and choose) detectors and interpret the results of the measurement, it is important to know how radiation interacts and what the consequences are of the various interaction to this extensive subject. The range of energies considered is shown in Figure 147(2004).

Particle	Energy (HeV)
alpha (u)	20
beta (B)	10
gamma (V)	20
neutrons (n)	15
Heavy ions	100

## Figure 14. Maximum Energy Calculated

For the discussion that follows, ionizing radiation is divided into three groups:

- Charged paticles: electrons (e<sup>-</sup>), positrons (e<sup>+</sup>), protons (p), deutrons (d), alphas (α) and heavy ions (A>4).
- 2. Photons: Gammas (8) or x-rays.
- 3. Neutrons.

The division is convenient because each group his its own characteristics and can be studied separately. Emphasis will be given to the last two as these are the most important in instrumentation. The discussion on charged particles will centre on explaining why these have limited ep. Losbility in instrumentation explainings.

Energy Loss and Penetration of Radiation through Matter

# 6.2 MECHANISMS OF CHARGED-PARTICLE ENERGY LOSS

Charged particles travelling through matter loss energy as a result of:

- 1. Coulomb interactions with electrons and nuclei.
- 2. Emission of electromagnetic radiation (bremsstrahlung).
- 3. Nuclear interactions
- 4. Emission of Cerenkov radiation.

When a charged particle travels through a cortain msterial it may interact with either the electrons, or the nucleus, of the actoms of the material. Since the cross-sectional area of the atom is approximately 10<sup>9</sup> times the cross-sectional area of the nucleus, interaction with the electrons is much more likely. Therefore, for the charged particles with kinetic energies concidered hare, nuclear interactions may be neglected.

Cerenkov radiation constitutes a very small fraction of the energy loss, and is visible electromegnetic radiation emitted by particles travelling in a medium, with speed larger than the speed of light.

As a result of energy losses, the particle finally stops after travelling a finite distance, colled the range. The probability of a charged particle going through a piece of material without interaction is practically zero. It is this fact that limits the application of charged particle radiation in process instrumentation.

## 6.2.1 COULOMB INTERACTIONS

A coulomb force F is exerted on an electron by a charged particle at a distance r from it. The action of the force over a period of time, may result in the transfer of energy irom the moving charged particle to the bound electron. Ionization or axcitation of the electron may then occur: Ionization occurs when the electron obtains enough energy to leave the atom and it then behaves like any other charged particle. Excitation occurs when the electron ownes to a higher energy level. In a short period of time the electron will move to a lower anergy level, if there is one available, and da x-ray will be united.

## 6.2.2 EMISSION OF ELECTROMAGNETIC RADIATION

Every free charged particle that accelerates or decelerates loses part of its kinetic energy by emitting electromagnetic rediation. This radi-

Energy Loss and Penetration of Radiation through Matter

ation is called breasstrahlung, which in German means braking radiation. Breasstrahlung is not monoenergetic, but consists of photons of energies from zero up to a maximum equal to the kinetic energy of the particle.

It can be shown that:

- For two particles travelling in the same medium, the lighter particle will emit a much greater amount of bremsstrahlung than the heavier particle (other things being equal).
- More bremsstlahlung is emitted if a particle travels in a medium with higher atomic number 7 than in one with a lower atomic number.

For low energy radiation, bremsstrahlung might be important for electrons only in high-Z materials like lead (Z=82).

## 6.2.3 STOPPING POWER DUE TO IONIZATION AND EXCITATION

A charged particle moving through a material exerts Coulomb forces on many stome simplemenously. Every stom has many electrons with different ionization and excitation potentials. As a result of this, the moving charged particle interacts with millions of electrons. Each interaction has a certain probability of courtence and energy loss. It is thus only possible to calculate an average energy loss per unit trevelled (dE/qZ). Voing to their different masses the calculation differs alignity for electrons and positrons than that for the heavier charged particles like protons, deutrons gata daylae particles.

This is because an incoming electron or "peitron may lose all its energy in a single collision with an atomic electron as they are of similar mass. They may also be assily scattered, and as a result their trajectory is a zigragg one. Heavier particles on the other hand loss smaller amounts of energy and their trajectory is almost a straight one. For the derivation of the average energy loss per unit distance travelled (dE/dx). (see Evans[1972], Sagre[1968] or Rey et. al. (1968)).

Many different names have been used for the quantity dE/dx: names like energy loss, differential energy loss, or stopping power. In this text the term stopping power will be used.

It should be noted that the stopping power [Tsoulfanadis, 1984]

1. Is independent of the particle,

2. Is proportional to the square of the particle charge,

Energy Loss and Penetration of Radiation through Matter

3. Depends on the speed of the particle,

4. is proportional to the density of the material.

## 6.2.4 RANGE OF CHARGED PARTICLES

A charged particle moving through a certain material loses its kinetic energy through interactions with the electrons and muclei of the material. Eventually the particle will stop, pick up the necessary number of electrons and become neutral.

The pathlength S, is the total distance travelled by the particle (sig zag path). The thickness of material that just stops a particle of kinetic energy T, eass H and charge z is called the range of the particle in the material. The range is equal to the displacement of the particle and is an average quantity.

Range can be measured in meters, or, if normalised by the meterial density in  $kg/m^2$ . The range measured in  $kg/m^2$  is independent of the state (density) of the metter, i.e. it is the same whether it moves in ice, steam or water.

Examples of particle ranges are:

à

2 94 A

3 NeV alpha particle in gold = 4µm.

2. 5 MeV deutron in air = 0.21m.

3. 1 MeV electron in gold = 113µm.

### 6.3 INTERACTIONS OF GAMMA AND X-RAYS WITH MATTER

X-rays and gamma rays are electromagnetic radiation. Considered as particles, when they travel with the velocity of light c, their rest mass is zero and so is their charge. A common name for x-rays and gamma (Y) rays as particles is photons. The relationship between energy, frequency and wevelanceth is:

 $E = hv = h \cdot c/\lambda$ 

(6.1)

44

There is no clear distinction between x-rays and gamma rays. The verm  $\times$ -rays is applied generally to photons with  $\mathbb{R} < 1$  MeV. Gammas are photons with  $\mathbb{E} > 1$  MeV. In what follows, the terms photons, gamma and x-rays will be used interchangeably.

Energy Loss and Penetration of Radiation through Natter

X-rays are generally produced by stomic transitions such as excitation and ionization. Gamma rays are emitted in nuclear transitions. Photons are also produced as bremsstrahlung, by accelerating or decelerating charged particles. X-rays and gamma rays emitted by atoms and nuclei are monoenergetic. Bremsstrahlung has a continuous energy spectrum.

Both x-rays and gamma rays are used in the process control industry. X-rays are used in x-ray flourescence to identify various substances and to examine process material (e.g. pipes) for defacts. Gamma rays are widely used in instrumentation and are the centre of discussion here. Gamma rays are used for level, density, thickness end weight measurement.

There is a long list of possible interaction of photons with matter, but only the three most important ones will be discussed here: the photoelectric effect, Compton scattering and pair production.

## 6.3.1 THE PHOTOELECTRIC EFFECT

The photoelectric offset is an interaction between a photon and a bounatomic electron. As a result of the interaction, the photon disappears and one of the atomic electrons is ejected as a free electron called a photoelectron.

The probability of this interaction occurring is called the photoelectric cross section or photoelectric coefficient and is designated  $\tau(m^{-1})$ ,

where

 $\tau$  = probability for photoelectric effect to occur per unit distance travelled by the photon.

The photoelectric effect decreases as  $E_{\rm y}$  increases and increases as Z increases. Thus it is more important, i.e. more probable for high Z materials. It is also more important for  $E_{\rm y}=10$  keV than  $E_{\rm y}=500$  keV for the same material.

### 6.3.2 COMPTON SCATTERING OR COMPTON EFFECT

The Compton affect is a collision between a photon and a free electron. Of course, under normal circumstances, all the electrons in a medium are not free but bound, however, if the energy of the photon is of the order of keV, while the electron bynding energy is of the order of eV, the electron may be considered free.

Energy Loss and Penetration of Radiation through Matter

45

\* \* \_

ż

The photon does not disappear after Compton scattering, only its direction and energy change. The photon energy is reduced by the anount given to the electron. The minimum energy of the scattered photon is greater than zero. Therefore, in Compton scattering, it is impossible for all the energy of the photon to be given to the electron. The energy of the electron will be dissipated in the material in a distance equal to the range of the electron. The scattered photon may escape.

The probability that Compton scatterin, will occur is called the Compton coefficient or the Compton cross section and is designated  $\sigma(m^{-1})$ ,

where o = probability for Compton scattering to occur per unit distance.

The probability for Compton scattering to occur is almost independent of the atomic number of the material.  $\sigma$  decreases as  $E_{\varphi}$  increases and is almost constant as 2 increases but does devrease slighty for high Z.

6.3.3 PAIR N

Pair production is an interaction between a photon and a nucleus. As a result of the interaction, the photon disappears and an electron-positron pair results, although the nucleus does not undergo any change as a result of the interaction, its presence is necessary for pair production to occur.

The kinetic energy of the electron-positron is equal to the photon energy minus 1.02 WeV, which is meassary for the production of the two cast enserve. Note that this means only gamma rays of energy greater than 1.022 NeW rmc capable of pair production. The electron end positron share, for all practical purpores, the available kinetic energy. Pair production eliminates the original photon, but two photons are insportant in constructing shielding against tham as woll as for the detection of gammas.

The probability for pair production to occur is called the pair production coefficient or cross section and is designated  $k(m^{-1})$ ,

where k = the probability for pair production to occur per unit distance travelled.

h has a threshold at 1.022 NeV and increases with  $E_{\chi}$  and Z. Of the three coefficients discussed, k is the only one increasing with the energy of the p-ton.

Energy Loss and Penetration of Radiation through Matter

## 6.3.4 TOTAL ATTENUATION COEFFICIENT

When a photon travels through matter, it may interact through any of the three interactions discussed earlier. (For pair production,  $E_{\chi}>1.022$  MeV). There are other interactions, but they are not mentioned here as they are not important in the detection of gammas.

Figure 15 (from Examp[1972]) shows the relative importance of the three interactions as  $E_q$  and Z change. Consider a ,hoton with b = 0.1 MeV. If this particle travels in carbon (Z=6), the Compton effect is the predominant effect by which this photon interacts. If the same photon travels in indim (=253), the photoelectric effect dominates.





The total probability for interaction  $\mu$  -called the total linear attenuation coefficient, is equal to the sum of the three probabilities.

 $\mu(m^{-1}) = \tau + \sigma + \chi$  (6.2)

Physically  $\mu$  is the probability of interaction par distance. There are tables that give  $\mu$  for all the obsents for many photon energies. Meat of the tables provide  $\mu$  in units of  $m^2/k_B$ . (or  $cm^2/k_B$ ) because in these units the density of the material dees not have to be specified. If  $\mu$  is given in  $m^2/k_B$  (or  $cm^2/k_B$ ) it is called the tatal mass attenuation coefficient. The relationship batween linear and mass coefficient is

 $\mu(m^2/kg) \approx \mu(m^{-1})/\rho(kg/m^2)$  (6.3)

Energy Loss and Penetration of Radiation through Matter

The total mass attenuation coefficient shows a minimum because as  ${\rm E}_\chi$  in-

creases,  $\tau$  decreases, k increases and  $\sigma$  does not change appreciably. However, the minimum of  $\mu$  does not fall at the same energy for all elements. For lead,  $\mu$  is minimized at  $E_q\sim3.5$  MeV; for aluminum at 20 MeV and for MaJ at 5 MeV.

If a parallel beem of monoentergetic gamma rays goes through a material of thickness t, the fraction of the beem that traverses the medium without any interaction is equal to  $e^{UE}$ . The probability that a photon will go through thickness t without an interaction is:

Probability of

traversing = <u>number transmitted</u> = e<sup>-µt</sup> (6.4) thickness t number incident

The average distance between two successive interactions, called the mean free path  $(\lambda)$ , is the inverse of the total linear attenuation coefficient.

The total mass attenuation coefficient for a compound or a mixture is:

$$\mu_{c}(m^{2}/kg) = \Sigma W_{i} \mu_{i}(m^{2}/kg)$$

where

 $\mu_{\mu}$  = total mass attenuation coefficient for a compound or mixture

W, = weight fraction of ith element in compound

 $\mu_{a}$  = total mass attenuation coefficient for ith element.

# 6.3.5 PHOTON ENERGY ABSORPTION COEFFICIENT

When a photon has an interaction, only part of its energy is absorbed by the a photon has an interaction took place. Snergy given by the photon to electrons and posicrons is considered absorbed at the point of interaction because the range of tness charged particles is short. However, x-rays, Compton tonitered photons or annihilation gammas may escipa. The fraction of photon energy that oscepses is important than one wants to calculate heat generated due to gamma energy deposited in any materials is calculated with the help of energy absorption coefficient defined in the following way.

The gamma energy obsorption coefficient is in general that part of the total attenuation coefficient that when multiplied by the gamma energy, will give the energy deposited at the point of interaction. The energy absorption coefficient  $\mu_a$  is:

 $\mu_{a} = \tau + Tav.\sigma/E_{\chi} + k \qquad (6.6)$ 

Energy Loss and Penetration of Radiation through Matter

48

(6.5)

Ħ

The total mass attenuation coefficient shows a minimum because as E, in-

creases,  $\tau$  decreases, k increases and  $\sigma$  does not change appreciably. However, the minimum of  $\mu$  does not fall at the same energy for all elements. For lead,  $\mu$  is minimized at  $E_{q}\sim$ 3.5 MeV; for sluwinus at 20 MeV and for NaI at 5 MeV.

If a parallel beam of monoenargetic gamma rays goes through a material of thickness t. the fraction of the beam that traverses the medium without any interaction is equal to  $e^{\mu t}$ . The probability that a photon will go through thickness t without an interaction is:

Probability of

traversing = <u>number transmitted</u>=  $e^{-\mu t}$  (6.4) thickness t number incident

The average distance between two successive interactions, called the mean free path  $(\lambda)$ , is the inverse of the total linear attenuation coefficient.

The total mass attenuation coefficient for a compound or a mixture is:

 $\mu_{c}(m^{2}/kg) = \Sigma W_{i} \mu_{i}(m^{2}/kg)$ 

(6.5)

where

 $\mu_{e}$  = total mass attenuation coefficient for a compound or mixtur.

W, = weight fraction of ith element in compound

 $\mu_{i}$  = total mass attenuation coefficient for ith element.

# 6.3.5 PHOTON ENERGY ABSORPTION COLFFICIENT

When a photon has an intoraction, only part of its energy is absorbed by the medium at the pint whare the intoraction took place. Energy given by the photon to alectrons and positrons is considered absorbed at the point of intraction because the range of these charged particles is abort. However, x-rays, Compton scattered photons or annihilation gammas may escape. The fraction of photon anergy that scapes is important when one wants to calculate heat gammated due to gamma absorption in shielding materials or gamma radiation does to themans. The gamma anergy deposited in any materiel is calculated with the help of energy absorption coefficiant defined in the following way.

The gamma energy absorption coefficient is in general that part of the total stanzation coefficient that when multiplied by the gamma energy, will give the energy deposited at the point of interaction. The energy absorption coefficient  $\mu_a$  is:

 $\mu_{\sigma} = \tau + Tav.\sigma/E_{\mu} + k \qquad (6.6)$ 

Energy Loss and Penetration of Radiation through Matter

where Tav is the average energy of the Compton electron and  $\mu_{\rm d}$  may be a linear or mass energy absorption coefficient depending on the units.

In writing Eq. 6.6 it is assumed that:

- If photoelectric effect or pair production takes place, all the energy of the gamma is deposited there.
- If Compton scattering occurs, only the energy of the electron is absorbed. The Compton scattered gamma escapes.

In the case of photoalactric effect, assumption (1) is good. For pair production, however, it is quantianable because only the energy E - 1.022 MeV is given to the electron-positron pair. The rest of the energy, equal to 1.022 MeV, is taken by the two ennihilation games and is not deposited in the medium. There are cases when Eq. 6.21 is modified to account for this effect [Mabbel1, 1969].

#### 6.4 INTERACTIONS OF NEUTRONS WITH MATTER

Neurons with protons, are the constituents of muclei. Since a neuron has no charge, it interacts with nuclei only through muclear forces. When it approaches a nucleus, it does not have to go through a Coulomb barrier, as a charged particle does. As a result, the probability (cross section) for muclear interactions is higher for neurons than for charged particies. This section discusses the important characteristics of neuron interactions, with emphasis given to neutron cross sections and calculation of interaction rates. Neutron radiation is commonly used in process instrumentation to determine the soisture content of bulk solids. (See Chamber 2).

## 6.4.1 TYPES OF NEUTRON INTERACTIONS

The interactions of neutrons with nuclei are divided into two categories.

#### SCATTERING

In this type of interaction, the neutron interacts with a nucleus, but both particles reappear after the reaction.Scattaring may be classic or inelastic. In elastic scattaring, the total kinotic energy of the two particles is conserved and simply distributed between the particles. In inelastic acattering, part of the kinetic energy of the neutron is given to the nucleus as excitation energy. After the collision, the excited nucleus will decay by emitting one or more gemma rays.

## ABSORPTION

Energy Loss and Penetration of Radiation through Matter



If the interaction is an absorption, the neutron disappears, but one or more particles appear after the reaction takes place.

# 6.4.2 NEUTRON CROSS SECTIONS

Consider a monoenergetic parallel heam of neutrons hitting a thin target (i.e. one that does not appreciably attenuate the beam). The number of reactions per second, R, may be written as:

 $R = I(n/m^{2}s)[N(nuclei/m^{3})] [a(m^{2})][t(m)][\sigma(m^{2})]$ (6.7)

The parameter  $\sigma$ , called the cross section, has the following meaning:

 o(m<sup>2</sup>) = probability that an interaction will occur per target nucleus per neutron per m hitting the target

The unit of  $\sigma$  is the barn (b).

 $1 b = 10^{-24} cm^2 = 10^{-28} m^2$ 

Neutron cross sections depend strongly on the energy of the neutron as well as the atomic weight and number of the target nucleus. The cross section  $\sigma(b)$  is called the microscopic cross section. Another form of cross section frequently applied is the mecroscopic cross section  $\xi(\alpha^{-1})$  and is analogues to the linear attenuistion coefficient of generarrays.

# 6,4.3 THE NEUTRON FLUX

In most cases the neutron source consists of nutrons that travel in all directions and not in the same direction as assumed till now. It is thus necessary to modify Eq. 6.7. Assume that at some point in space the neutron density is n(neutrons/m<sup>2</sup>). If a target is placed at that point, the interaction rate R (reactions/m<sup>2</sup> will be:

$$R = n(neutrons/m^{3})[v(m/s)][\Sigma(m^{-1})]$$
(6.8)

The product nv, represents the total pathlength travellad per second by all the neutrons in 1 m<sup>-1</sup>, and is called the neutron flux §.

Anothe: quantity related to the flux and used in radiation exposure calculations is the neutron fluence F, defined by

Energy Loss and Peuetration of Radiation through Matter

50

N

## 6.4.2 NEUTRON CROSS SECTIONS

Consider a monoenergetic parallel besm of neutrons hitting a thin target (i.e. one that does not appreciably attenuate the beam). The number of reactions per second R, may be written as:

 $R = I(n/m^{2}s)[N(nuclei/m^{2})] [a(m^{2})][t(m)][\sigma(m^{2})]$ (6.7)

The parameter o, called the cross section, has the following meaning:

### o(m<sup>2</sup>) = probability that an interaction will occur per target nucleus per neutron per m hitting the target

The unit of  $\sigma$  is the barn (b).

$$1 b = 10^{-24} cm^2 = 10^{-28} m^2$$

Nextron cross sections depend strongly on the energy of the nextron as well as the storic weight and number of the target nucleus. The cross section  $\sigma(b)$  is called the mirroscopic cross section. Another form of cross section frequently applied is the macroscopic cross section  $\{\mathbf{r}_{n}^{(a)}\}$  and is malosue to the linear attemption coefficient of names rays.

### 6.4.3 THE NEUTRON FLUX

In most cases the neutron source consists of neutrons that travel in all directions and not in the same direction as assumed till now. It is thus necessary to modify Eq. 6.7. Assume that at some point in space the neutron density is n(neutrons/m<sup>2</sup>). If a target is placed at that point, the interaction rate B (reactions/m<sup>2</sup>) will be:

$$R = n(neutrons/m^{2})[v(m/s)][\Sigma(m^{-1})]$$
(6.8)

The product nv, represents the total pathlength travelled per second by all the neutrons in 1 m<sup>-1</sup>, and is called the neutron flux  $\hat{P}$ .

Another quantity related to the flux and used in radiation exposure calculations is the neutron fluence F, defined by

50

Energy Loss and Penetration of Radiation through Matter

# $F(n/m^2) = \frac{1}{2}t$

where t is the exposure time to flux 4.

# 6.5 CONCLUSION

This chapter discussed energy loss and penetration of radiation in matter. It was shown that charged particles have a finite range and are quickly absorbed by the material. Photoms, however, have only a statistical probability of absorption. The probability depends on the properties of the material. An important property affecting absorption is the density of the material. By determining the number of photoms absorbed, a measure of the density of the material can be obtained. The density of the material is also important when building detectors. The higher the density of the detector material, the more photoms will be absorbed and the higher the detector efficiency.

In order to detect the number of photons passing through a material, and hence infer some of the material properties, a detector is required. Two common detector families are discussed in the following two chapters.

Energy Loss and Penetration of Radiation through Matter

51

# 7.0 GAS-FILLED DETECTORS

# 7.1 INTRODUCTION

In this section the basic operation of gas-filled detectors is outlined. Although, gas-filled detectors heve, until now been the mainstay of nuclear based process instrumentation, they are being replaced by scintillation detectors.

Gas-filled detectors can be either gas-filled or gas-flowing, however, only gas-filled detectors will be discussed, as only these are used in process instrumentation.

Gas-filled detectors operate by utilizing the ionization produced by radiation as it passes through a gas. Typically, such a detector consists of two electrodes with a certain potential applied across them. The space between the electrodes is filled with a gas. lonizing radiation passing through the gas generates electronic pairs. These charge carriers move under the influence of the electric field, inducing a current on the electrodes which may be measured (Figure 16 on page 53).

Either the current is measured directly, in which case the detector is called a current or integrating chamber, or alternatively, the charge may be transformed into a pilse by inserting a resistor into the circuit. The voltage pulse developed across the resistor is then measured and amplified. In this case the detector is called a pulse chamber.

For most gases, the average energy to produce an electron-iron pair is about 30 eV. If a 3 MeV alpha or bata particle deposite its energy in the counter, it will produce, on average

(3 MeV)/(30 eV) = 100 000 electron-ion pairs [Tsoulfanidis, 1983]

A typical gas-filled detector has a capacitance of about 50 pF, and the charge will be collected in a time of the order of 1µs. If all the charge is collected, the voltage and current expected are of the order

 $V = Q/C \sim (10^{\circ} x \ 1.6 \times 10^{-19} c/el)/(50 \text{ pF}) = 0.5 \text{ mV}$ i = Q/t ~ (10<sup>5</sup> x \ 1.6 \times 10^{-19} \text{ A})/10^{-6} = 1.6 \times 10^{-8} \text{ A}

Gas-filled Detectors

3

Electrons move to the positive plate and ions to the negative plate under the same force P=Ee, where D = the electric field intensity. The acceleration, a=P(m, is nowever quite different as the ion mass is approximately 1000 times the electron mass. Thus the time to collect the electron is about hys, and the time to collect the ion about ms.



Figure 16. A typical gas-filled detector

7.2 RELATIONSHIP BETWEEN HIGH VOLTAGE AND CHARGE COLLECTED

If, in an experiment, the HV applied to the counter is steadily increased, the charge collected per unit time changes as shown in Figure 17 on page 54. The curve is divided into 5 regions. (Tsoulkanidis, 1963)

### REGION I

When the voltage is very low, the electric field is weak, electrons and ions move relatively slowly, and have a considerable recombination rate. As V increases, the field becomes stronger, the carriers move faster and the recombination rate decreases until it becomes zero. Then, all the

Gas-filled Detectors

53





charge created is being collected  $(V=V_{\underline{1}})$ . Region I is called the recombination region.

## REGION II

In region II, the charge collected stays constant despite a change in the voltage because the recombination rate is zero and no new charge is produced. No charge multiplication takes place. The output signal is proportional to the margy dissipated in the detactor; therefore measurement of particle emergy is possible. Convorsity, low energy particles will produce small output signals. Thus fontastion counters can be used for particle identification. This region is called the ionization region and is the region in which the ionization chamber, used for density measurement, is operated. The applied voltage is usually less than a 1000 V.

# REGION III

In this region, the collected charge starts increasing because the alectrons produes secondary ionization that results in charge multiplication. The electric field is so strong, in a certain fraction of the volume, that electrons from primary ionization acquire enough energy between collisions co produce additional ionization divided by the primary factor (i.e. the ratio of the total ionization divided by the primary

Gas-filled Detectors

ŧ

ſ.

ionization) is for a given voltage, independent of the primary ionization. The counter output is thus proportional to the energy dissipated in the counter; therefore particle identification and energy measurement is possible. The higher the anergy of the particle the higher the output signal. This region is called the proportional region, but is not important in process instrumentation. Proportional counters may be used for the detection of any charged signal. The applied voltage ranges between 800 and 200 V.

#### REGION IV

In this region, the electric field in the counter is so strong that a single electron-ion pair generated in the chamber is enough to initiate an avalanche of electron-ion pairs. This avalanche will produce a strong signal with shape and height independent of the primary ionization and the type of the particle. Region IV is colled the Geiger-Waller (GH) region. Detectors operating in this region are called Geiger-Waller counters and are used for detection of radiation thus making them ideal for point level measurement. The advantage of GM counters is that the's their relatively long dead time (200 to 300µs). The applied voltage ranges between 1000 to 3000 V.

## REGION V

If the applied voltage is raised beyond the value  $V_{\gamma\gamma\gamma}$ , a single ionizing

event inistes a continous discharge in the gas and the device is not a counter anymore. A commercial counter operating in this region will probably be destroyed.

In operation as a detector in a density gauge the gas-filled detector must operate in a tither region II or in region IV. Operation in region II is however preferable, as the long dead times characteristic of region IV will lead to many photoms not being detected. Operation in region IV is preferable for on/off type measurements i.e. in determining if redistion is present on root.

## 7.3 DIFFERENT TYPES OF GAS-FILLED COUNTERS

## 7.3.1 CONSTRUCTION

Gas counters may be constructed in any of three basic geometries [Tsoulfanidis,1983]: parallel plate, cylindrical or spherical. In a parallel-plate chamber the electric field is uniform, with strength equal to

Gas-filled Detectors

### E = Vo/d

where d is the plate separation.

In the cylindrical chamber, the voltage is applied to a very thin wire, stretched axially at the centre of the cylinder. The cylinder wall is usually grounded. The electric field in this case is:

E(r) = Vo/(ln(b/a).r)

where

- a = central wire radius
- b = counter radius
- r = distance from the centre of the counter

Thus, very strong fields can be maintained inside a cylindrical countar close to the central wire. This is the usual construction for gas-filled detectors.

In a spherical counter, the voltage is applied to a small sphere located at the centre of the counter. The counter wall is usually grounded. The electric field is:

 $E(r) = Vo.ab/((b - a)r^2)$ 

where a, b and r have the same meaning as in the cylindrical case. Strong fields may be produced in the spherical counter but this geometry is not favoured because of construction difficulties.

A counter filled with a gas at a certain pressure may operate in any of the regions discussed previously, depending on the following parameters:

- 1. Counter size.
- 2. Size of wire in cylindrical counters.
- Gas type.
- Gas pressure.

5. Level of high voltage.

Normally gas counters are manufactured to operate in one region or 'y, The manufacturer has selected a combination of the above variables that results in a ionization, proportional or 6% counter. It is nowadays usual for manufacturers of process instrumentation to include the high voltage supply in the deteator casing and it is thus not variable as would be the case for counters for general use.

Gas-filled Detectors

8

## 7.3.2 IONIZATION CHAMBERS

Ionization chambers are used in process instrumentation to count the number of particles resching the datector. From a knowledge of the source strength the density of the meterial can then be obtained. Although it to obtain the to obtain the meterial can the particles using a family chamber, for density measurement one is only interested in the number of particles reaching the detector.

The ionization chamber may be used to produce either pulses or current. To produce a pulse a resistance is connected across the chamber which forms a capacitor. The electronion pairs created by the ionization of the gas, produce a small time-varying voltage superimposed on the NV of the applied field. However, because of the low drift velocity of the ions, the length of the period of the resulting pulse, (of the order of smol and the product of the resulting pulse, (of the order of smol result in incoming photons nor being detected. Thus, to ensure discrimimation between various particles, the pulse period is truncated with a RN anode (of the order of us). Isoulfandia [1983] gives calculations of pulse formation for a persile plate geneery while Frames and Cochrane [1963], and Kewalski[1970] give datailed calculations of the pulse shapes for the three geneeries of gas-filled chambers.

An ionization chamber of the current type measures the average ionization produced by incoming particles. This is achieved by measuring directly the current generated in the chember. The proper operating voltage of a current ionization chember is that for which all the ionization produced by the incoming relation is measured. If this is the case, a slight increase in the voltage will have a negligible effect on the current. This voltage is called the scurrein voltage and the corresponding current, the saturation current. The value of the seturation current depends on the intensity and the type of source.

It should be noted that the ionization chember has an inherent RC time constant. For the pulse chember the C corresponds to the chember and the R to the resistor connected across the chember. For the current type chember the R results from the input resistance of the measuring instrument or samplifier. Thus the ionization chember has a limited response time.

The advantages of the ionization chamber are its:

- 1. Mechanical stability
- 2. Simplicty in construction and
- 3. High reliability.

Gas-filled Detectors

The disadvantages of the ionization chamber are:

- The detection efficiency (i.e the ratio of incoming photons detected to the total number of incoming photons) is low (about 1%). Consequently large radiation sources have to employed which sutematically increases the health hazard. The detection efficiency can be raised by increasing the gas pressure (hence its density) or adding extra capacitor plates. Both alternatives however increase the price of the chember considerably. Another consequence of the low detection efficiency is the necessity of a wide radiation beam in order to detect as many seman quanta as possible.
- The ionization chamber does not discriminate batween types of radiation, and is therefore sensitive to backround radiation, which can be alleviated by careful and efficient screening.
- 3. The ion current is directly proportional to the gas pressure which is temporatura dependant. The high ohmic resistors required as a result of the low ionization currents (in the piccamp range) are also highly respondent compondant. Consequently to maintain accuracy it is often necessary to mount the chamber and/or the resistor in a temporature controlled oven.

# 7.3.3 GEIGER-MULLER (GM) COUNTERS

# 7.3.3.1 Operation and Quenching of the Discharge

The GM counter is usually cylindrical in shape. The electric field close to the very thin central wire is so strong that the gas multiplication factor M is extremely high. In a GM counter, a single primary electron-ion pair triggers a great number of successive avalanches. Therefore the output signal is independent of the primary ionization.

The operation of the GM counter is as follows. When the electrons are accelerated in the strong field crounding the wire, they produce, in atoms and molecules of the gas. These excited atoms and molecules produce photons when they deexcite. The photons, in turn, produce photoelectrons in other parts of the counter. Thus, the avalanche, which was originally located close to the wire, spreads uickly to the counter volume. During all this time the electrons are continuously collected by the anode wire. while the much slower moving positive ions are still in the counter and form a positive sheath around the anode. When the electrons have been collected, this positive sheath, acting as an electrostatic screen, reducas the field to such an extent that the discharge should stop. However, this is not the case because the positive ions eject electrons when they strike the cathode, and since by that time the field has been restored to its original high value, a new avalanche starts and the process just described is repeated. Clearly some means is needed by which the discharge is permanently stopped or "quenched". There are two methods of quenching the discharge.

Gas-filled Detectors

In external quenching, the operating voltage of the counter is decreased after the start of the discharge until the ions reach the acthods. to a value for which thergas multiplication factor is negligible. The decrease is achieved by a properly shown RC circuit as shown in Figure 18. The resistance R is as shigh that the voltage drop across it due to the current generated by the discharge  $\left( i_{\rm d} \right)$  raduces the voltage of the counter below

the threshold needed for the discharge to start. The net voltage is Vo -  $i_{\rm d}R.$  The time constant RC, where C represents the capacitance between

anode and cathode, is much longer than the time needed for the collection of ions. As a result, the counter is inoperative for an unacceptably long period of time, i.e. the dead time is too long.



## Figure 18. External quenching circuit for GM counter

The self- .ching mathed is accomplished by adding to the main gas of the cour \* small amount of a polyatemic organic gas or d halogen gas. The organ jas molacules, when ionized, lose thair emergy by dissociation rather than by photoelactric processes. Thus, the number of photoelectrons is greatly radued. In ddition or that, when the organic ions strike the surface of the cathods, they dissociate instead of causing the ejection of new ions. Therefore, new avalandhes do not start.

GM counters using an inorganic gas , a quanching agent have a finite lifetime because of the dissociation of urganic molecules, Usually, the GM counters last for a 10<sup>4</sup> to 10<sup>5</sup> court. The lifetime can be considerably increased if a halogen gas is used or a quanching sgent. The helgen molecules has dissociated but there  $\gamma$  , pertain degree of regeneration of the molecules which gravity per the lifetime.

Gas-filled Detectors

# 7.3.3.2 The Pulse Shape and the Dead Time of a GM Counter

For GM counter the output signal is a result of the sum of all the contributions from all the positive ion avalanches produced throughout the volumes of the counter. The pulse however rises very slowly. The shape and height of GM is counter pulses are not very important because the pulse is only used to signal the presence of the pan icle and nothing else. Howwer, how non pulse effects the formation of enother is important.

During the formation of a pulse the electric field in the counter is greatly reduced because of the presence of positive ions around the anode. If a particle arrives during that period, no pulse will be formed because the counter is insensitive. The insensitivity lasts for a time called the dead time of the counter. Then the detector slowly recovers with the pulse height growing exponentially during the recovery period. This is illustrated in Figure 19 on page 11 which shows the change of voltage and pulse for a typical GN counter. Typical values of dead time are from 100 to 300µs. If the dead time is 100µs and the counting rate is 500 counts/s, there is gring to be a 55 loss of counts due to dead time.

Because of the dead time, GM counters are used as comparators in lavel switches, i.e. they are only used to indicate the presence of radiation. GM counters are not employed for counting purposes in process instrumentation.

## Gas-filled Detectors





## 7.4 CONCLUSION

三日二日

In order to perform any sort of measurement of nuclear radiation a detector is required. This chapter discussed gas-filled detectors. Two types of gas-filled detectors are available: the Geiger-Muller counter and the Ioninazion chamber.

The GW counter has a long dead tame and is thus inappropiate in accurate counting applications. GM counter are applied to stutions such as laval measurement, where it is only necessary to detect the absence or presence of radiation. When the material in the bunker is below the laval of the measuring path, a high means of radiation (31) reach the detector and a high count rate will be detected. When the material reaches the measuring level, the rediation hear will be strongly attenuated and a much lower count rate will result. It is usual for manufacturers to include the discrimination logic into the detector butishing and only to provide a single binary output. Geiger-Woller detectors are thus not suitable for slurry density measurement.

Gas-filled Detectors

1.6
The ionization cheaker has been the predominant detector in industrial applications but is gradually been superseded by the scinitilation counter. Its mair "dvantages are its simplicity and reliability. However, because it is "read on a gas, which has a low density, it has a low efficiency (typically 1%). This results in a much larger source than would be required for " more efficient detector. In order to compensate for the low efficiency can draise the number of photons reaching the detector a divergent radistion beam is used. This compounds the increased health hazard already presented by the larger source.

The other family of detectors suitable for slurry density measurement are scintillator based detectors. These are discussed in the following chapter.

Gas-filled Detectors

62

() ()

1

\*6

## 8.0 SCINTILLATION DETECTORS

# 8.1 INTRODUCTION

Scintiliators are materials -solids, liquids, gerse-that produce sparks or scintiliations of light when ionizing radiation passes through them. The first solid material to be used as a particle detector was a scintiliator. It was used by Ruberford, in 1930, in his alpha scattering experiments. In his setup, alpha particles his a sinc sulphide screen, which were counted with the aid of a microscope. Thus was a very inafficient, inaccurate and the consuming process, which resulted in the sethod being abandomed for about 30 years. It was remaved when the devent of electronics made porsible amplification of the light orduced in the scintiliator. Despite this, and despite their devents of constantion -chambers, (40%, versus 1% typical conversion affiriscory and insensitivity to backround rediation); it was only recently that they became robust and reliable enough to be employed in an industrial enviroment. [Boringer: 1979]

The assumt of light produced in the scintillator is very small. It must be amplified before it can be recorded as a pulse or in any other way. The amplification or multiplication of the scintillator's light is achieved with a device known as a photomultiplier tube (or phototube). Its name denotes its function: it accepts as small amount of light, samplifies it many times and delyvers a strong pulse at the output. Amplifications of the order of 10<sup>6</sup> are common.

The operation of the scintillation counter may be divided into two broad steps:

- Absorption of incident radiation energy by the scintillator and production of photons in the visible part of the electromagnetic spectrum.
- Amplification of the light by the photomultiplier tube and production of the output pulse.

The different .ypes of scintillators may be divided into three groups [Tsoulfanidis, 1983]:

- 1. Inorganic scintillators
- 2. Organic sciutillators
- 3. Gaseous scintillators

However only inorganic scintillators (crystal scintillators) are used for detectors in industrial instruments and only these will be discussed.

Scintillation Detectors

# 8.2 INORGANIC (CRYSTAL) SCINTILLATORS

Most of the inorganic scintillators are crystals of the alkali iodidas, that contain a small concentration of an impur'.y. Examples are Nal(T1), Cs(T1), Cs(TA), Lil(Fu) and Cs $_{\beta}$ (Eu). The element in parentheses is the

impurity or activator. Although the activator has a relativly small concentration it is the agent responsible for the luminescence of the crystal

# 8.2.1 THE MECHANISM OF THE SCINTILLATION PROCESS

The luminescence of inorganic scintillators can be understood in terms of the allowed and forbidden energy bands of a crystal. The electronic energy states of an atom are discrete energy levels, which in an energy level diagram are represented as discrete lines. In a crystal, the allowed energy states widen into bands (Figure 20 on page 65). In the ground state of the crystal, the uppermost allowed band that contains electrons is completely filled. This is called the valence band. The next allowed band is empty (in the ground state) and is called the conduction band. An electron may obtain enough energy from incident radiation to move from the valence to the conduction band. Once there, the electron is free to move anywhere in the lattice. The removed electron leaves behind a hole in the valence band which can also move. Sometimes, the energy given to an electron is not enough to raise it to the conduction hand, instead it remains bound to the hole in the velence band. The electron hole pair is then called an exciton. The exciton states form a thin band with the upper level coinciding with the lower level c, the conduction band.

In addition to the exciton band, energy stats may be created between valance and conduction bans because of crystal impurities or imperfections. Particularly important are the states created by the activator atenses such as thallium. The activator atom are wrist in the ground state or one of its excited states. Elevation to an excited state may be the result of photon absorption, or the capture of an exciton, or the successive capture of an electron and a hola. The transition of an impurity atom from an excited to a ground state (if allowed), results in the emission of a photon. If this photon has a weelength an the visible part of the electromagnetic spectrule to the scintification. Thus, production of a scintillation is the result of the occurrence of these avents:

1. Ionizing radiation passes through the crystal.

2. Electrons are raised to the conduction band.

3. Holes are created in the valence band.

Scintillation Detectors



#### Figure 20. Allowed and forbidden energy of a crystal

4. Excitons are formed.

- Activation centers are raised to the excited states by absorbing electrons, holes and excitons.
- De-excitation is followed by emission of a photon.

The light emitted by a scinitilitary is primarily the result of transitions, of the activator atoms and not of the crystal. Since most of the incident energy goes to the lattice of the crystal (eventually becoming heat), the appearance of luminescence produced by the activator atoms means that energy is transferred from the host crystal to the impurity.

The magnitude of the light output and the wavelength of the emitted light are two of the most important properties of any scinitilator. The light output affects the number of photoelectrons generated at the input of the phototube, which in turn affects the puts height produced at the counting system. Information about the wavelength is necessary in order to match the scinitilator with the proper photomultiplier tube. The light output also depends on the temperature of the crystal. Emission spactra and temperature response are given in Tsoulfandis[198].

Scintillation Detectors

65

iy P

# 8.2.2 TIME DEPENDANCE OF PHOTON EMMISSION

Since the photons are emitted as a result of decays of excited states, the time of their emission depends on the decay constants of the different states involved. Experiments show that the emission of light follows an exponential decay law of the form

 $N(t) = Noe^{-t/T}$ 

(8.1)

where N(t) = number of photons emitted at time t T = decay time of the scintillator

Most of the excited states of the scintillator have essentially the same lifetime T. There are however some states with longer lifetimes contributing a slow component in the dacay of the scintillator known as afterglow, which may be important in certain measurements where the phototube output is integrated.

In a counting system using a scintillator, the light produced by the crystal is amplified by a phot-multiplier tube and is transformed into an electric current having an exponential behaviour given by  $\mathbb{F}_Q$ , 8.1. The current is fed into a parallel RC circuit and a voltage pulse is produced of the form

$$V(t) = V_{e}(e^{-t/RG} - e^{-t/T})$$
 (8.2)

In practice, The value of RC is cho-an much greater than T. Thus for short times -i.e. T << RC, which is the timespan of interest- Eq.8.2 becomes

 $V(t) = V_{m}(1 - e^{-t/T})$  (8.2a)

Thus the rate at which the pulse rises is determined by the decay time T.

# 8.2.3 IMPORTANT PROPERTIES OF CERTAIN INORGANIC SCINTILLATORS

## Nal(T1)

This is the most commonly used scintilistor for gamma rays and is the basis for almost all scintilistion detectors in industrial density gauges. Its rolativaly high density  $(3.67 \times 10^4 \text{ kg/m}^2)$  and high atomic number combined with its large volume make it s gamma ray detector with a very high afficiancy (40 to 80%). Although semiconductor detectors have better energy resolution, they cannot replace NaI(T1) in experiments where large detector volumes are needed.

Scintillation Detectors

The emission spectry of NaI(71) peaks at 410 nm and the light conversion efficiency is the highest of all the inorganic scintillators. It does however have many undesirable properties. It is brittle and sensitive to temperature gradients and thermal shocks. It is also so hygroscopic that it must be kept encapsulated at all times. NaI(71) always contains a small amount of potessium, which creates a certain backround because of its radioactive  ${}^{40}$ K.

CsI(T1)

CsI(T1) has a higher density (4.5Lt0<sup>3</sup>Kg/a<sup>3</sup>) and higher atomic number than NaI(T1); therefore it afficiancy for gamma detection is higher. The light conversion efficiency is however only about 45% of NaI(T1) at room temperature. At light individues (7KK), pure CsI(T1) has a light output equal to that of NaI(T1) at room temperature. The emission spectrum extends from 420 nm to about 600 nm.

CsI(T1) is not hyproscopic. Being softer and more plastic than NaI(T1), it can withstand severe shocks, acceleration and withstain as well as large temperature gradients and suddam temperature changes. These properties make it suitable for space arguments. Finally, (SsI(T1) does not contain potessium. It would also appear to be very suitable for industrial applications.

CsI(Na)

The density and atomic number of CsI(Na) are about the same as those of CsI(T1). The light conversion efficiency is about 85% of that of NaI(T1). Its emission spectrum extends from 320 nm to 540 nm. CsI(Na) is slightly hygroscopic.

CaF<sub>2</sub>(Eu)

 $CaP_{\chi}(Eu)$  consists of low-atomic number materials and this makes it an efficient detector for beta particles and x-rays. Thus its application in density gauges is limited.

Lil(Eu)

This is an efficient thermal-neutron detector and has no application for density gauges.

Scintillation Detectors

#### 8.3 THE PHOTOMULTIPLIER TUBE

# 8.3.1 GENERAL DESCRIPTION

The photomultiplier tube or phototube is an integral part of a scintilation counter. Without its amplification, a scintillator is useless as a radiation detector. The photomultiplier is essentially a fast amplifier, which in picoseconds amplifies an incident pulse of visible light by a factor of a million.

The photomultiplier consists of an evented glass tube with a photoschole at its entrance and several dymodes in the interior (Figure 21 on page 69). The photons produced in the scintillator enter the phototube and hit the phototubed, which is made of a material (Ca-Sb) that emits electrons when light strikes it. The electrons emitted by the photocathode are guided, with the help of an electric field, towards the first dymode, which is costed with a substance (Ca-Sb or Ag-Mg) that emits secondary electrons, if electrons in process is repeated for each successive dymode. Typical commercial phototubes may have up to 15 dymodes.

The electrons produced in the phototube are directed from one dynode to the next by an electric field established by applying a successively increasing positive voltage to each dynode. The voltage difference between two successive dynodes is of the order of 80-120 V.

A very important parameter of every photosultiplier tube is the spectral sensitivity of its photocathole, which for best results should match the scintilator spectrum. Another important parameter is the magnitude of the dark current. The dark current consists mainly of alectrons amitted by the cathode after thermal energy is absorbed. This process is called thermionic emission and a 50-em diameter photocuthode may release as many as 10° electrons/s at room temperature. Obviously, the magnitude of the dark current is important in cases where the rediction aware is very weak. Both the dark current and spectral response should be considered when a phototube is purchased. Recall that the electrons are guided from one dynode to the next by an electric field. If a magnetic field is next dynode. To roduce the magnetic affect, the phototube can be surrounded with a cylindrized sheet of p-metal.

## 8.3.2 ELECTRON MULTIPLICATION IN A PHOTOMULTIPLIER

The electron multiplication (N) in a photomultiplier can be written as [Tsoulfanadis, 1983]:

Scintillation Detectors



Figure 21. Schwastic diagram of photomultiplier interior  

$$M = (\theta_1 \epsilon_1)(\theta_2 \epsilon_2) \dots (\theta_n \epsilon_n)$$
 (8.3)  
where  
 $n = number of dynodes$   
 $\epsilon_1 = number of electrons collected by (if dynode
 $\theta_1 = number of electrons emitted by (if dynode
number of electrons implinging upon ith dynode
If  $\theta_1$  and  $\epsilon_1$  are constant for all dynodes, then$$ 

 $M = (\theta \epsilon)^n$ 

2

Scintillation Detectors

69

(8.4)

The quantity  $\epsilon$  depends on the geometry. The quantity  $\theta$  depends on the voltage betw: r successive dynodes and on the material of which the dynode is made. The 'vendance of  $\theta$  on voltage is of the form

$$\theta = kV^{\alpha}$$

(8.5)

(8.6)

where

 $\mathtt{V}\approx\mathtt{V}_{\underline{i}}$  -  $\mathtt{V}_{\underline{i}+\underline{i}}$  = potential difference between any two successive dynodes.

k, a = constants (the value of a is about 0.7)

Thus the multiplication M bacomes

$$M = \epsilon^{n} (kV^{n})^{n} \simeq CV^{n}$$

where  $C = (zk)^n = constant$ , independent of voltage.

Eq. 8.6 indicates that the value of M increases with the voltage V and the number of stages n. The number of dynodes is initial, because as n increases, the charge density between two dynodes distorts the electric field and hindars the emission of electrons. From the previous dynode with the lower voltage. If one takes n=10 and Ur=4, a typical value, M becomes evual to 10<sup>6</sup>.

To apply the electric field to the dynodes, a power supply provides a voltage adequate for all the dynodes. A voltage divider, usually an intagral part of the preamplifier, distributes the voltage to the individual dynodes. When reference is made to the phototube voltage, one means the total voltage applied.

# 8.4 DEAD TIME OF SCINTILLATION COUNTERS

The dead time or resolving time, is the minimum time that can elapse after the arrival of two successive particles and still result in two separate pulses. For a scintillation counter, this time is equal to the sum of three time intervals:

- Time it takes to produce the scintillation, essentially equal to the decay time of the scintillator.
- Time it takes for electron multiplication in the phototube, of the order of 20-40 ns.
- Time it takes to amplify the signal and record it by a scalar. The recovery time of commercial scalars is of the order of lus. The time taken for amplification and discrimination is negligible.

Scintillation Detectors

The quantity  $\epsilon$  depends on the geometry. The quantity  $\theta$  depends on the voltage between successive dynodes and on the material of which the dynode is made. The dependance of  $\theta$  on voltage is of the form

$$\theta = k y^{a}$$
 (8.5)

where

 $V = V_i - V_{i-1}$  = potential difference between any two successive dynodes. k, a = constants (the value of a is about 0.7)

Thus the multiplication M becomes

.

$$M = \epsilon^{n} (kV^{n})^{n} = CV^{n}$$
(8.6)

where  $C = (\epsilon k)^n = constant$ , independent of voltage.

Eq. 8.6 indicates that the value of M increases with the voltage V and the number of stages n. The number of dynodes is limited, because as n increases, the charge density between two dynodes distorts the electric field and hinders the emission of electrons from the previous dynode with the lower voltage. If one takes n=10 and  $\theta_{\Sigma}=4$ , a typical value, M becomes equal to 10".

To apply the electric field to the dynodes, a power supply provides a voltage adequate for all the dynodes. A voltage divider, usually an integral part of the presentifier, discributes the voltage to the individual dynodes. When reference is made to the phototube voltage, one means the total voltage applied.

#### 8.4 DEAD TIME OF SCINTILLATION COUNTERS

The dead time or resolving time, is the minimum time that can elapse after the arrival of two successive particles and still result in two separate pulses. For a scintillation counter, this time is equal to the sum of three time intervals:

- 1. Time it takes to produce the scintillation, essentially equal to the decay time of the scintillator.
- 2. Time it takes for electron multiplication in the phototube, of the order of 20-40 ns.
- 3. Time it takes to emplify the signal and record it by a scalar. The recovery time of commercial scalars is of the order of lus. The time taken for amplification and discrimination is negligible.

Scintillation Detectors

By adding the three ab. components, the resulting dead time of a scintillation counter is of the order of tens to hundreds of microseconds. Thus scinitilators are detectors with fast responses.

# 8.5 SOURCES OF BACKROUND IN SCINTILLATION COUNTERS

The major backround source is the dark current, which is the leakage current flowing in the photomultiplier when no scintillations are being produced. Other backround sources are naturally ocurring radioisotopes, cosmic rays and phosphorescing substances.

It is however possible to disuriminate between these sources and gamma ray radiation, thus backround radiation is not much of a problem for industrial density genues using scintillation detectors.

# 8.6 INDUSTRIAL SCINTILLATION DETECTORS

Although the scintillation counter is the oldest method of *radiation de*taction, it is only since the development of the photomultiplier tube that it has become accepted in industry and today it is no doubt the most important detector in use.

The primary advantue of the scintillation conter is its high efficiency from the primary advantue of the scintillation of the scintary of the scintary distribution of the scintary and the scintary allowing the scintary end backword scintary allowing the scintary and the scintary scintary and the scintary and the scintary and the scintary scintary and the scinta

The limitations of the scintilation counter is that since the output pulse voltage depends on the interaction voltage a single precision voltage stabilizer must be used. Also, the surrounding temperature could influance the tube emplification. The scintillation crystal is highly hygroscopic and care must be taken, not to accidentally open the atmospheric tight enclosure.

There are still widespread rumours about the mechanical weakness of the scintillation counter and its sensitivity to vibration. It may be stated however, that scintillation counters are amployed in rockets in the military and space fields, where they are subjected to extremes of vibration and acceleration.

#### Scintillation Detectors

#### 8.7 CONCLUSION

Until new schrtilation commers have not beam considered reliable and scable enough for industrial applications. These problems have now beam overcome and it is expected that the scintillation counter will become the predominant detector for counting applications. The sain advantage of the scintillation counter is its high efficiency, between 40 and 80%, allowing a corresponding reduction in source size when compared to the ionization chamber. The higher efficiency also allows a colliarated beam to be used which further reduces the health hazard. A further advantage is the much higher contribution counter which allows for less deviation in the measurement mean or a faster response for the same deviation.

It should therefore be clear that given a choice between the ionization chamber and scintillation counter, the scintillation counter should be ured. The smaller source size and collisated beam which result from its use minimizes the health hazard which increases the acceptability of a nuclear density gauge.

When any instrument is used there is a question of the validity and accuracy of the measurement. As rediation is a statistical phenomem, the ouput of a radiation delector will have some statistical variance. Stalistics and errors in nuclear measurement are discussed in the following chapter.

Scintillation Detectors

# 9.0 STATISTICS AND ERRORS IN NUCLEAR MEASUREMENTS

# 9.1 INTRODUCTION

In this previous chapters it was shown that nuclear disintegrations are statistical induce. This was also and the senitor of the senitor of the senitor of the measurements of nuclear disintegrations, some knowledge of statistics is the senitor of the senitor of the senitor of the statistics and the rather, is meant to illustrate comparison of the statistical neture of radiation and how they can be compensated for the statistical neture of the senitor of the senitor of the statistical neture of the statistics of the statistical neture of the senitor of the senitor of the statistical neture of the statistical neture of the statistical neture of the senitor of the senitor of the statistical neture of the senitor of th

# 9.2 THE ERRORS, ACCURACY AND PRECISION OF MEASUREMENTS

A measurement is an attempt to determine the value of a certain parameter or quantity. In attempting any measurement the following two axioms regarding the result of the measurement should be kept in mind.

Axiom 1 No measurement yields a result without an error.

Axiom 2 The result of a measurement is almost worthless unless the error associated with that result is also reported.

The term error is used to define the following concept:

Error = (messured quantity) - (true value)

or Error = estimated uncertainty of the measured quantity

The result may be reported in two ways:

- 1. R  $\pm$  E; In which case E is the absolute error. (R and E have the same units).
- 2.  $R \pm e_{s}^{*}$ ; Where  $e \approx (E/R)100 \approx relative error (dimensionless).$

In most cases, the relative error rather than the absolute error is reported. The important this growthat that the the sean that the correct result is bracksted between R - E and R + E. It only means that there is a probability that the correct result has a value between R - E and R + E. The most common probability values used are the standard error and the probable error.

Statistics and Errors in Nuclear Measurements

# THE STANDARD ERROR

If the result of a measurement error is reported as R  $\pm$  Es and Es is the standard error, then there is a 68.3% chance for the true result to have a value between R - Es and R + Es.

# THE PROBABLE ERROR

By definition, the probable error is equally likely to be exceeded or not. Therefore, if the result of a measurement is  $R \pm Ep$  and Ep is the probable error, then there is a 50% chance for the result to be between R - Ep and R + Ep.

Other errors used include the hime-tenths, minety-five hundredths and ninety-nine hundredths error. The type of error used should always be quoted with the result.

Related to the error of a ensurement are the terms accuracy and precision. The accuracy of an experiment is, how close the result of the measurement is to the true value of the measured quantity, and is the difference between the real value and the measured value. The precision of the measurement, is on the other hand, related to the number of significant figures representing the result.

Limitations in the accuracy and precision of measurements result from many causes. Among the most important are:

- 1. Incorrectly calibrated instruments.
- 2. Algebraic or reading errors of the observer.
- Uncontrolled changes in environmental conditions, such as temperature, pressure and humidity.
- Inability to construct arbitrarily small measuring meter-sticks, rods, pointers, clocks, lenses, etc.
- A natural limit of sensitivity for any real measuring instrument detecting individual effects of atoms, electrons, molecules and protons.
- Imperfect method of measurement in most cases.
- Unknown exact initial state of the system. Dr, even if the initial state is known, it is impossible to follow the evolution of the system.
- Statistical nature of some processes, e.g. radioactive decay. There
  is a probability that a radioactive atom will decay in the next 10s
  and this is a "uch information as one can report on the matter.

Statistics and Errors in Nuclear Measurements

Nost measurements of applied nuclear reinnes involve the counting of nuclear events. Events of this nature follow a statistical law -the Poisson statistica- which is simple and easy to apply. In many cases, the inherent statistical error, due to the fundamental fluctuations in the rate of occurrence of the events, is the principle error. It becomes important to 'a able to predict or calculate the error. Generally speaking, the stabulity of the counting apparatus is such that the fundamental statistical error will dominate if it is greater then 1%.

# 9.3 THE NORMAL DISTRIBUTION

Before proceeding with the discussion of Poisson scatistics, first consider the normal distribution to which a large class of physical mearurements conform. Most measurements of quantities which have a continuously variable magnitude appear to fit the normal distribution, which states that the probability df that the quantity will lie between x and x + dx is [Title 1965]:

$$dP = (1/s/2\pi)/e^{-(\chi-\pi)^2/2s^2} dx \qquad (9.1)$$

where m is the mean value of the distribution and s is the standard deviation, a measure of the width of the distribution factor. A plot of y = dP/dx vs. x is the well known "bell-shaped curve".

Measurements give a set of values of x which are regarded as members of the population described by the distribution. The choice of distribution for the description is arbitrary. There is no assurance that a given series of measurements will actually fit a normal distribution or any other distribution. One of the most important applications of statistics to measurements is, in fact, the investigation of the question of whether a particular set of measurements (in the easumed distribution.

)rdinaily the value of interast is the quantity m. To obtain an estimate of m, the observations, x, are averaged to obtain what is called the observed mean, x'. If there are anough measurements, an estimation of the atandard deviation (s) from the set of observed value x  $\cdot$  x', can be obtained it can then be decided whether the measurements fit the assumed distribution. In the case of the normal distribution it is generally measures of base above and the "goodness of fit". However, it is sometimes necessary to result of sets of data.

The test estimate of s is given by [Tittle, 1965]

 $s = \frac{1}{(n-1)} \sum_{i=1}^{n} \sum_{i=1}^{n} (x_{i} - x^{i})^{2}$ (9.2)

where n is the number of observations. In this equation the factor n - 1 in the denominator, rather than n, results in the use of  $x^{\rm I}$  instead of m

Statistics and Errors in Nuclear Measurements

in the numerator. The true mean m is never known exactly. Consider, as an example the data in Figure 22 on page 76.

Analysis of Geiger-Mueller Counter Data.

Test	x	x - x'	(x - x') <sup>2</sup>
1	209	~18	324
2	217	-10	100
3	248	21	441
4	235	8	64
5	224	- 3	9
6	223	- 4	16
7	233	6	36
Totals	1589	0	990

Figure 22. Example of Standard Deviation Computation

The mean for the above set of data is

 $x' \approx 1589/7 = 227$ .

The standard deviation for the above set of data is

 $s = \sqrt{((990)/(7-1))} = 12.8.$ 

The question of interest is: How well does the observed mean represent the true mean? The standard deviation of the mean  $S_{\rm m}^{},$  is given by the relation

S <sub>m</sub> = s/√n	(9.3)
S = s/√n	(9.3)

or  $12.8/\sqrt{7} = 4.9$  in the example. The result would therefore be quoted as

m = 1589 ± 4.9 (9.4)

where the quoted error is the standard deviation or standard error.

# 9.4 THE POISSON DISTRIBUTION

Poisson's distibution describes most of the counting observations made in experimental and applied nuclear physics, and is given by [Title]e65]:

 $Px = m^{X} \cdot e^{-m}/x_{1}$  (9.5)

Statistics and Errors in Nuclear Measurements

where Px is the probability of observing x events when the average for a large number of tries is m events. In this distribution, x and m are integers - one point of difference from the normal distribution, where the variable is continuous. If m is a fairly large number, say 100 or larger, the Poisson distribution is adequately approximated by a special normal distribution

$$dP \approx (1/\sqrt{2\pi}) \int e^{-(\chi - m)^2/2m} dx$$
 (9.6)

. . . . . where by comparison with Eq. 9.1 the standard deviation is

s ≃ √m (9.7)

The standard deviation of a normal distribution approximating a Poisson distribution is therefore specified when m is specified. In most cases Eq. 9.6 is an adequate representation of the Poisson distribution, even if m is as low as 20. Therefore the relation between the standard deviation s and the probable error r of the normal distribution, where

may be used.

If the data in Figure 22 on page 76 represents seven counts made in a set of uniform time intervals, the predicted Poisson standard deviation would be  $m = \sqrt{(1589/7)} = 15.1$ . A simple test of counting data is to compute s from the residuals and from m and see if they agree. If they disagree appreciably, this test does not tell anything about the significance of the disagreement. The chi-square test, discussed next, is recommended as a better test.

# 9.5 THE CHI-SOUARE TEST OF GOODNESS OF FIT

Pearson's chi-square test determines the probability P that repetition of the observations would show greater deviations from the assumed distribution than those observed in the first trial. The quantity chi-square is defined as follows:

$$\chi^{2} = \frac{(\text{observed value})i - (\text{expected value})i}{(\text{expected value})i}^{2}$$
(9.9)

The date should be subdivided into at least five classifications (i), each containing at least five events. In the counting application, the expected value is the average count observed.

The number of degrees of freedom F is the number of groups of data minus the number of imposed conditions. For the Poisson distribution

F = n - 1 (9,10)

Statistics and Errors in Nuclear Measurements

where n is number of observations. The imposed condition is the specification of m. In the case of the normal distribution, an additional specification is that of s, hence F=n-2.

From a knowledge of F and X<sup>2</sup>, computed from the experimental data, the value of P is 0.5. A greater value indicates that the fluctuations are less than one would expect from the assumed distribution. If P lies between 0, is not 0.9, it is generally assumed the distribution corresponds to the observed one, but if P is less than 0.0 cor greater than 0.96, the distribution corresponds to the observed me, but if P is less than 0.0 cor greater than 0.96, the summed the distribution corresponds to the observed me, but if P is less than 0.0 cor greater than 0.96, the summed the distribution of the division of the contract of the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me, but if P is less than 0.0 corresponds to the observed me that the observed me, but if P is less than 0.0 corresponds to the observed me that 0.0 corresponds to the observed me that 0.0 corresponds the observed me that 0.0 corresponds the observed me that 0.0 corresponds to the observed me that 0.0 corresponds the observed me that 0.0 corresponds to the observed me that 0.0 corresponds to the observed me that 0.0 corresponds the observed me that 0.0 corresponds the observed me that 0.0 corresponds to that 0.0 correspo

It is just as bad to get a set of data that is "too consistent" as it is to get inconsistent data. This may indicate for example that spurious pulses of a wurferm rate are mixed in with the desired pulses. To much inconsistency (a shall value of P) usually means instability of some component such as the high voltage supply. It can also simply mean an unlucky run, which should be kept in mind if the apparatus performs satisfactorily after one seeming malfunction.

In the example of Pearson's chi-square test in Figure 22 on page 76, F=6 and P=0.6, which is a little consistent but quite acceptable.

## 9.6 CHAUVENET'S CRITERION FOR REJECTING A READING

Occasionally a reading is obtained which, although velid in the normal sense that no malfunction occurred and the reading may well be a manber of the correct population, deviates so much from the sean that it adversally affects the observed mean. The criterion established by Ghauvenet states that such a reading is to be rejected if it has a deviation greater than that corresponding to the l/2n probability limit (the l - 1/2n error). For example in a sorties of 10 readings, if  $x \sim x$  exceeds 1.56 (the 0.95 error), that reading should be rejected, and the mean should be recomputed with the reading monited from consideration.

# 9.7 PROPAGATION OF ERRORS

When two quantities are combined by addition or subtraction, the error of the sum or difference is not the simple sum of the arrors of the individual quantities, but is less than the because of the probability that the errors may partially cancel. The law of addition of independent errors is

 $s = \sqrt{(s_1^2 + s_2^2)}$  (9.11)

Statistics and Errors in Nuclear Measurements

where s is the quoted error and  $s_1$  and  $s_2$  are the individual errors.

When quantities are multiplied or divided, the fractional or per cant errors add quadratically. For example if 100  $\pm$  5 is divided by 20  $\pm$  2 the result is

 $5 \pm \sqrt{(5\%^2 + 10\%^2)}$ = 5 ± 11.2% = 5 ± 0.56

# 9.8 DEAD TIME LOSSES

Type I Systems - Type I systems are paralyzable systems. In a paralyzable system, a photon entering the system during the dead time of the provious photon detacted will retrigger the dead time. (c.f. retriggeble and non-restriggeble emonstables.) These systems count only intervals which are longer than the dead time T. These systems are not used in industrial applications.

Type II Systems - These are nonparalyzable systems; i.e. an impulse arriving during a dead interval commot institute another dead interval. Self-quenching GM counters as well as ionization and scinitilation counters are members of this group. The true impulse rate N is given in terms of the observed rate n and the dead time T by

N = n/(1 - nT)

## 9.9 BACKROUND COUNTING TIMES

If it is possible to count the backround radiation, it is possible to find an optimum division of time between backround and sample counting. This however does not apply to industrial applications where it is not possible to perform just backround counting in an on-line instrument.

# 9.10 THE CASE OF THE DECAYING SOURCE

The methods described here assume the source disintegrations do not decrease appraciably during the counting period, i.e. the counting period is very much less than the half lifs, as is usually the case.

Statistics and Errors in Nuclear Measurements

79

(9.12)

# 9.11 CONCLUSION

Statistics are an essential part of any nuclear measurement and an understanding of statistics is essential when analysing radiation measurement. It was shown that the stavistical error can be minimized by obtaining as high a count as possible and by averagin: the count over as large a data set as possible. The size of the count and the data set must however be weighed against the response time of the meter.

This concludes the review of the basic theory behind radiation measurement and its application to density measurement. In order to apply nuclear instruments an appreciation of the health hazards and associated legal requirements is essential. These aspects are discussed in the following chapter.

#### Statistics and Errors in Nuclear Measurements

## 10.0 HEALTH PHYSICS AND LEGAL REGULATIONS

#### 10.1 INTRODUCTION

Any instrument or piece of equipment can constitute a s "ious hazard if it is not properly utilized, and nuclear based instrumentsion is no different. If the correct precautions are observed there is no reason why muclear based instrumentsion should not be as safe, or even asfer than much of the equipment commonly encountered in any industrial environment. It is the potential health hazard which causes many to shy away from nuclear instruments, and it is indeed, this health hazard which leads to the use of muclear based instruments being regulated by law.

The intention of this chapter is to outline the potential threat any nuclear source imposes, to provide advice on the protection of users of nuclear sources from ionizing radiation and to review the legal requirements which must be met when nuclear sources are used.

Nuch of the meterial is taken from a guidance produced by the Paper and Board Industry (UK) Advisory Committee (IAC) to assist British shills in meeting their iegal obligations and to avoid health risks. This material is not legally binding but is highly recommended [PSI,1983]. Fo further information, Tsoulfanidis[1983] gives a thorough discussion on the calsulation of dome rates.

#### 10.2 HEALTH PHYSICS

Health Physics is the discipline that consists of all the activities related to the protection of individuals and the general public from potentially harmful effects of ionizing radiation. Ionizing radiation comes from two sources:

- Natural or backround radiation that is emitted by radioisotopes which exist on or inside the earth, as well as radiation incident upon the earth from outer space. Humans have been exposed to this natural radiation as long as they have lived on planet.
- Human-made radiation which is emitted by all the radioisotopes that have been produced through muclear reactions (mainly fission), as well as radiation produced by machines used in medical installations (e.g. x-ray machines) or in scientific laboratories (e.g. accelerators).

Health Physics and Legal Regulations

Health physics is concerned with the protection of people from radiation. Since the backround radiation has always been on our planet at about the same level everywhere, there is not much a health physicist can do to protect individuals or populations from backround radiation. Hence, health physics is, scsentially, concerned with human-made radiation.

#### 10.3 EFFECT OF EXPOSURE

Exposure to ionizing radiation can result in interforance with the structure of access because of ionization. The extent of this depends on the dose received, the type of radiation and the energy of radiation. In the case of humans, such exposure may result in some atoms of the matter contained within human cells behaving in an abnormal manner. This can result in complete destruction of these cells or interformet with their normal method of reproduction. These effects are tareed scanatic, where there is interforence with normal cell reproduction, the effect normally long terms and can result in the development of cancers.

When exposure to radiation results in cells not reproducing themselves in an identical form, because the generic material is altered, the damage can result in the dawaiopument of horeditory defects in subsequent generations. This is termed the genetic effect. The somatic effect therefore concerns individuals, whereas the genetic effect is of concern to acciety as a whole.

Gamma and X-ray radiation of sufficient energy can penetrate and pass through the body when it is exposed to such radiation sources. The hazard is thus both external and internal and can be sometic, and if the reproductive organs are involved a genetic risk.

Beta radiation has relatively little penetrating power and the hazard from such radiation external to the body is only to the surface areas and is a somatic risk only. Ingestion, inholation or absorption of the bata-emitting substance itself into the body becomes an internal exposure hazard and can be a somatic and, in some cases a genetic risk.

Most alpha radiation is stopped completely by the skin and therefore presents little or no external radiation hazard and no external sometic risk. As for beta-emitting patielns, ingestion, inhelation or absorption of the alpha-emitting substance has both a sometic and genetic risk.

It is prudent to assume that there is some risk from any done of rediction above the zero level and that risks to helch vary with the dose of radiation received. It is recognised that express continiously at the rate of 50 milliStovert par year (Sievert (Sv) is a measure which quantifies the equivalent absorbed dose.) throughout the whole of a parson's working life would lead to an increased probability of the development of canvers.

Health Physics and Legal Regulations

Legislation in Great Eritain has followed the recommendation of the International Commission on Radiological Protectical, Persons likely to receive more than 15 milliSiverts per year from their work exposure mend to be designed as classified workras and are then subject to modical supervision and dose monitoring throughout their working life in such employment.

## 10.4 PRINCIPLES OF PROTECTION

# 10.4.1 BASIC CONCEPTS

The basic concept of radiological protection, and hence the regulatory control, is to keep the radiation dose as low as reasonably practicable. Mechods of achieving this protection include minimum sources, collimation, maximum working distance from source and appropriate shielding.

The radiation source should be of the lowest activity and energy necessary for the intended purpose taking into account the performance required and the need to avoid replacing sources to often. The latter can result in unnecessary frequent exposure.

A radiation source emits radiation in all directions and so the source holder should be designed to ensure that radiation is hielded off except in the direction of the "window". This radiation should be adequately collimated, to ensure that it is directed perpindicular to the window as useful been and aligned so that it does not spreed appreciably. Note that when ionization chambers are used as detectors, it is normal to have several degrees divergence in the beam.

The radiation source should be mounted as far awy as reasonably practicable from any moreal working area, and wherver possible the beam diracted away from that area. The radiation dose (Tsoulfanidis[1983]) varies with the distance from the source according the inverse square law. Thus, if the distance is doubled, the dose rate is reduced by a factor of four.

Madical supervision is required in circumstances where the foregoing methods of protection cannot continue to be provided during maintenance, repair or replacement of the radiation source equipment. Persons who might as a result be exposed during that work to a radiation dree of more than 7.5 microSivert per hour read to be made classified workers.

Health Physics and Legal Regulations

Classification requires the person to undergo specific medical examinations and to wear some form of personal dose meter. A health register and dose record have to be meintained for each classified person and be available for inspection by that person. The classified person may be subject to an annual medical examination dependent upon the redistion dose being received. No classified person should receive more than 50 milliSivert in a year and vryking methods and procedures meed to be examiniti and resportside whenever a classified person reaches a 15 milliSivert total dose.

#### 10.4.2 WHAT AN EMPLOYER SHOULD DO

The Paper and Board Industry (UK) Advisory Committee (IAC) recommends that the employer should ensure that:

- A suitable person is appointed and is readily available to advise on radiological protection and relevant legal requirements.
- The most suitable source of ionizing radiations for the purpose required is used and that the minimum activity commensurate with satisfactory economic performance is employed.
- The source of ionizing radiation is sited at a suitable position so as to restrict the potential of exposure to as few persons as it is reasonable to achieve.
- Any source of ionizing radiation is installed with adequate shielding so that during normal use no persons are exposed to radiation.
- 5. The equipment is provided with suitable warning signals and notices, interlocked barriers and shielding as appropiate and that these are all regularly meintained in an efficient state.
- 6. Those persons who might be exposed to ionizing radiations during maintenance work are classified and have been subject to medical examination, provided with suitable dosemeters and that records of medical examinations and any radiation doses received are maintained.
- Persons conducting work are provided with such shielding as is reasonably practicable for their protection during that work. They should also have been informed of any risk involved, the precautions to be observed and the methods by which the work should best be carride out.
- All persons operating .ny machine or equipment in which a source of ionizing radiation is installed have been informed of the precautions taken and instructed not to misuse or interfere with the equipment.
- 9. Suitable monitoring equipment is provided or is readily available.
- 10. The location of each source of ionizing radiation, the type of radiation, and its activity is always known by keeping an appropriate up to gate record and the operation of a check procedure.

Realth Physics and Legal Regulations

- 11. The use and disuse of ionizing radiations is notified to the enforcing authority and any loss, leakage or damage of a source and any incident where a person may have been unnecessarily exposed is notified to the authority and properly investigated.
- 12. Contingency plans have been made and are familiar to all persons who need to know, to catter for accident situations such as fire, explosion or impact that may have interfered with the integrity of the source and any safety equipment.
- Persons other than h. employees are not put to any risk to their health and safety as a result of his use of ionizing radiations.

#### 10.4.3 WHAT AN EMPLOYEE SHOULD DO

The Paper and Boatd Industry (UK) Advisory Committee (IAC) recommends that the employee should:

- Not interfere or misuse any safety devices provided for his protection against exposure to ionizing radiations.
- Report immediately to his employer any failure of an interlock or warning signal, or damage to any shield or barrier.
- Wear and use where necessary any safety equipment provided for his protection.
- 4. Operate and use the equipment only as instructed.

#### 10.4.4 WHAT AN SUPPLIER SHOULD DO

The Paper and Board Industry (UK) Advisory Committee (IAC) recommends that the supplier, agent or installer should:

- Determine the use for which the equipment is intended and the position at which it is intended to be used.
- Provide the most suitable source of ionizing redistions for the required purpose.
- Provide and/or install the equipment with appropriate shielding, barriers, interlocking arrangements, warning signals and notices.
- Ascertain where necessary that the equipment has been properly installed.
- Provide information and instructions so that the equipment may be properly used without risk to ionizing radiations.

Health Physics and Legal Regulations

- 11. The use and disuse of ionizing radiations is notified to the enforcing authority and any loss, leakage or damage of a source and any incident where a person may have been unnecessarily exposed is notified to the authority and proper'y investigated.
- 12. Contingency plans have been made and are familiar to all persons who need to know, to cater for accident situations such as fire, explosion or impact that may have interfered with the integrity of the source and any safety equipment.
- Persons other than his employees are not put to any risk to their health and safety as a result of his use of ionizing radiations.

#### 10.4.3 WHAT AN EMPLOYEE SHOULD DO

The Paper and Board Industry (UK) Advisory Committee (IAC) recommends that . the employee should:

- Not interfere or misuse any safety devices provided for his protection against exposure to ionizing radiations.
- Report immediately to his employer any failure of an interlock or warning signal, or damage to any shield or barrier.
- Wear and use where necessary any safety equipment provided for his protection.
- 4. Operate and use the equipment only as instructed.

#### 10.4.4 WHAT AN SUPPLIER SHOULD DO

The Faper and Board Industry (UX) Advisory Committee (IAC) recommends that the supplier, agent or installer should:

- Determine the use for which the equipment is intended and the position at which it is intended to be used.
- Provide the most suitable source of ionizing rudiations for the required purpose.
- Provide and/or install the equipment with appropriate shielding, barriers, interlocking arrangements, warning signals and notices.
- Ascertain where necessary that the equipment has been properly installed.
- Provide information and instructions so that the equipment may be properly used without risk to ionizing radiations.

Health Physics and Legal Regulations

# 10.5 LEGAL REQUIREMENTS

Conditions for the acquisition, possession, disposal, importation, export, use and conveyance of radioactive nuclides in South Africa are governed at present by the Atomic Energy Act, 1967 (Act 90 of 1967).

The latest conditions governed by the act are contained in the Government Gazette, No. 3105, of 28 November 1980. It should be noted that the Act is subject to change and as such it is the onus of the user of radioactive nuclides to be familiar with the latest regulations.

#### The major points of note are:

(Note that these are only major points, and it advisable to consult the latest regulations before purchasing or using radioactive material.)

- 1. A permit must be obtained to store or use radioactive materials.
- 2. Suitable (as determined by the act) storage must be arranged.
- A responsible person and alternat, "ith duties laid down in the sct, must be appointed. This person shall if required by the Board, submit himself for an examination or test > the Board.
- The concentrations of radioactivity in cir and water to which persons are exposed, shall not exceed the value of determined from time to time by the board.
- Applications for authority to use radioactive material (sealed sources) for industrial vadiography must be submitted to the Board on the designated form.
- 6. The holder of the authority shall ensure that permanently built-in sealed sources in apparetus such as oliminators of static electricity, level gauges, density maters and thickness gauges are tested for leaks at intervals not exceeding twenty-four months and all other sealed sources tested at intervals not exceeding six months. Pertiulars of such tests shall be enterval in the source register.

# 10.6 EMERGENCY PROCEDURES

The employer should make such plans as are necessary, and ensure that the approprise staff are made sware of them, and also have any equipment necessary to cater for a fire, explosion or impact accident involving a source of ionizing radiation. These plans should be in the form of written procedures, regularly updated.

The following matters need to be considered in such plans;

Health Physics and Legal Regulations

- 1. Exclusion from the area concerned of all persons except classified workers and persons otherwise authorised.
- Notification to the person appointed to advise on radiological protection or any other organization retained to advise.
- Monitoring to determine the extent of the radiation or contaminated area.
- Provision of emergency shielding in the form of sand bags, lead sheet or bags of lead shot.
- Identification of persons who might have been exposed to radiation or who might have ingested, inhaled or otherwise absorbed radioactive material.
- Medical examination of and the provision of biological samples from those concerned if necessary.
- 7. Notification of the incident to the enforcing suthority.

# 10.7 CONCLUSION

An approved the health hazards and set legal required consisting the application of the health hazards and set legal required consisting the application of these instruments. Nuclear based instrumentation is subplicatological regulation in South Africa and the necessary procedures must be followed before using such instruments. The two facts of wost concern to the proper using such instruments must be obtained to possess a nuclear source and that a responsible person and alternate must be sppoints to oversee their use.

The fact that these instruments are regulated by law, may seem to imply that these instruments are more dangerons than other equipment commonly found in an industrial environment. This is in fact not the case, and the regulations are primarily simed at monitoring the locations of nuclear sources in the country and ensuring their safe disposal. The regulations and safety procedures are no more onerous than those applying to many other dangerous chemicals.

Chapter's Five to Nine have discussed the theory of nuclear radiation applicable to density measurement in a general sense. The following chapter discusses the application of this theory in coursercial instruments.

#### Health Physics and Legal Regulations

# 11.0 GAMMA RAY INSTRUMENTS IN AN INDUSTRIAL ENVIROMENT

# 11.1 INTRODUCTION

In this chapter gamma ray instruments in an industrial environment are discussed. The intention is to relate the material in the earlier chapters, i.e. the survey and selection of instruments for the slurry monitoring system, to the theoretical material given in the latter chapters together.

When an instrumentation engineer purchases a density or level gauge, he typically obtains a complete unit. It is now comeon for manufacturers to include the high voltage supply and preamplifier for the detector into the detector housing. The supplier will supply a radiation source of a given type and strength based on the buyers data. Thus it is difficult to know the tradeoffs make in the selection of the instrument components and the suitability of the instrument to different process conditions, e.g. a change in process material.

To enable the user of nuclear gauges to gain a batter understanding of the above, this chapter discusses applications of nuclear gauges and gives some insight into the tradeoffs made by the supplier when specifying an instrument to suit a customers application.

# 11.2 INDUSTRIAL INSTRUMENTS

Industrial instrumentation based on gamma radiation is becoming widely accepted in several industries such as mining, petrolecum and the chemical industry. Their primary adventages are that they are

- 1. Non-contacting.
- 2. Maintenance free.
- Reliable.
- 4. Accurate.
- 5. Simple to calibrate.
- 6. Clamp-on in many cases.

Despite their disadvantages (mainly concerning the health hazard if improperly used, and the associated logal regulations), they will continue to be widely used and will come to dominate their sphere of application.

Gamma Ray Instruments in an Industrial Enviroment

All nuclear instruments are based on the attenuation of the gamma-ray photons as they pass through the material. The attenuation is governed by the equation

1 = Io.exp(-U.L)

(11.1)

where Io is the number of geneme photons emitted by the source and I the number of geneme photons reaching the detector. I is the pathlength of the geneme rays in the meterial. U is the linear attenuation coefficient and is the product of material density and the mass absorption coefficient. U is also the probability of a geneme photon passing through the material. Thus for any number of geneme photons (Ic) emitted by the source, there is always some probability U, governed by the type of material, that I geneme photons will pass through the material. (See Chapter 6.) Thus any change in the material or its density or the pathlength can be datacted. Therefore the following measurements (semogat charter) are possible:

1. The density of the material if the pathlength is fixed.

2. The thickness of the material if the density is fixed.

3. The solids content of a slurry, as density varies with concentration.

# 11.3 FACTORS INFLUENCING DENSITY MEASUREMENT

Although only the factors influencing density measurement will be discussed, the principles apply to all the industrial applications. When a user buys a nuclear density gauge, the supplier will request cartain information and will supply an instrument to meat the specified requirements. In most cases all the user them has to do is install the meter and calibrate it. The intertion of this section is to indicate the tredeoffs made in specifying an instrument. This will provide a means of judging the difference between two meters specified by different suppliers, or to judge the difference in two products from the same menufacturer.

In a nuclear density gauge the following factors affect the performance of the meter:

- The type of source which affects the penetration capability and the life of the source (half-life).
- 2. The type of detector which influences the source strength.
- 3. The health hazard which limits the source strength.
- 4. The secondary electronics which influences the parformance of the meter.

Bensity measurement in slurry pipelines is based on Eq. 11.1 with L being fixed and U varying with the density. In order to determine the density the following must be known:

The value of Io.

Samma Ray Instruments in an Industrial Enviroment

2. The attenuation (i.e. I/To) as a function of density.

First consider the detected gamma photons I, i.e. the photons pessing through the material. The value of I has a statistical variation owing to the statistical nature of the mattriced photons [o. ]n order to obtain an accurate and repetable measurement it is necessary to minimize the relative arror of I.

In the chapter on statistics (chapter 9), it was shown that the standard deviation for a Poisson distribution, (which governs radioactive decay) is

 $s = \sqrt{m}$  (11.2)

where m is the mean. It was also shown for a set of measurements, the standard deviation of the observed mean Sm, is given by the relation

$$Sm = s/\sqrt{n}$$
 (11.3)

where s is the standard deviation of the set and n is the number of measurements in the set. Therefore the relative error e is:

e = E/Iav	(11.4)

here	B ≕ s/√n	(11	.3	٢

but s = Iav (11.2)

therefore

 $e_{n}^{*} = (1/\sqrt{(lav,n)})100$  (11.5)

Eq. 11.5 shows that the relative error can be reduced in two ways;

1. By increasing the average count (Iav).

2. By increasing the size of the data set (n).

The sverage count depends on the count rate and the period of counting. It has an upper limit dependent the overflow value of the counter. The count rate is limited by the maximum count rate of the detector, which is determined by the detector dead time, and the source size which has an upper limit determined by the health heard, i.e. larger sources require thicker shielding. The count period and the data set size is limited by the response the of the meter, i.e. the longer the count period the gradient the data set the longer the capation of the set of the set of the data set the longer the response time of the meter.

The mean count value is also influenced by the range of measurement required. To increase the sensitivity of the meter the difference between the maximum and minimum count values over the given density range must be as large as possible.

The source size which determines Io is influenced by four main factors and is given by the relation

Gamma Ray Instruments in an Industrial Enviroment

# Io = I/(A.n.SA.HL)

## where

- A is the absorption factor. For a required detected count value, the detected count value must be increased by the absorption factor to obtain the source count.
- n is the efficiency of the detector. For any number of photons striking the detactor only a certain fraction will be detacted. To componante for this the source must be increased proportionally. This effect can be reduced by using a detector with the highest possible efficiency.
- SA is the solid angle factor. Since, the source emits rediation in all directions, the source is shielded in all directions except the direction of the detector in order to prevent stray rediation and uninsies the health heard. For high efficiency detectors such as scintilistion counters the emitted radiation is collimated into a single beam. For low efficiency detectors such as ionization chamhers the beam is allowed to diverge slightly to increase the number of detected photons.
- HL is the half life factor. Because the number of disintegrations emitted by the source halves over the hulf-life poriod, the size of the source must be increased initially to maintain the desired count rate at the end of its useful life. For example, assume the life of the meter is five years. For a Gobalt-60 source which has five year half-life the size of the source evald have to be doubled to ensure the required count after five years. For Gespue-137 which has a 30 year half-life the size would only have to increased by a factor of 1.12.

The attenuation as a function of density is given by Eq.11. At ocalculate the density directly from this equation would require a knowledge of U and the absolute value of 10. Because these quantities are never known explicitly. Here, are "aclculated" by calibrating the setur against one or several known reference values. In some cases the calibration curve is computer generated by the manufacturer according to the client spacifications. The source decay must also be compensated for. This either done automatically or by periodic recoellbration.

# 11.4 SPECIFYING A NUCLEAR DENSITY GAUGE

A nuclear density gauge comprises of a source, a detector, a mounting brackst and associated electronics. A particular gauge will use a predetermined detector, mounting brackat and electronics, thus the specification of the gauge holds down to the selection of type end size of source.

The majority of manufacturers provide mounting brackets of various sizes allowing the user to clamp the density gauge on to correspondingly sized

Gamma Ray Instruments in an Industrial Environent

(11.6)

pipes. One manufacturer, namely Krohne, mounts the density gauge onto a section of pipe which must then be inserted into the system.

## 11.4.1 THE NUCLEAR SOURCE

The specifications of the source size must be done by the supplier based on the client's data. The primary objective is to minimize the source size and thorby the health hazard. The source size is a tradeoff between the health hazard and the accuracy, rangeability and life-time of the meter. The source size and hence the health hazard, can be reduced by using scintilitation counters rather that doniestion chambers as detectors. In order to increase the accuracy as high a mean count as possible as posrible must be obtained, but the variation in the count must be largecrease the lifetime of the mean the halt-life of the source must be as long as possible. A longer half-life of also rasults in a smaller initial source size as the size is chosen to give a sufficient count rate at the end of the meter life.

The type of source used is dependent on its penatration shifty, and its helf-life. Three sources are commonly used. Americine-241 which has a low penstration capability (0.660 HeV radiation), but s long half-life (433 years). Cassimi-137 which has a moderate penetruing capability (0.662 HeV radiation) and half-life (50 years) and Cobalt-50 which has a high penetration capability (1.17 - 1.33 MeV radiation) but short half-life (5 years). Amercium-241 is saidom used as it does not have sufficient in level gauge applications. Cobalt-60 is not busually used in density gauges because of its short half-life, but is most commonly used in level gauge applications which have long measuring paths and where the short half-life is not a limiting factor. It may be used in density gauge applications, but may be used in lavel measurement for low donsity materials and short measuring paths. The half-life of Casejum-137 (37 years) results in a 3% error after one year if the doary is not compensated for.

To calculate the source size from the clients data, manufacturers use expirical formulae based on the above aspects, the characteristics of thair own hardware and the clients specifications. The formulae used differ from manufacturer to annufacturer and basically just consists of an equation that works and results in a suitable specification. To is thus not possible to give an quantion which can be used to check the source size against any application. The formulae used by a specific manufacturer are usually given in the manual accompaning the density gauge. If it is desired to charge the application, these formulae can be used to check the suitability of the source to the new application.

Gamma Ray Instruments in an Industrial Environent

## 11.4.2 THE DETECTOR

As was previously discussed (See Chapter 7 and Chapter 8) three types of detectors are used.

1. The Geiger-Mueller counter,

The GM counter has a long dead time and is thus inappropriate in accurate counting applications. GM counters are applied to situations such as level measurement, where it is only necessary to detect the absence or presence of radiation.

2. The Ionization Chamber.

The ionization chambers main advantages are its simplicity and reliability. However because it has a low efficiency, typically 1%, it is gradually been supersaded by the scintillation counter.

3. The Scintillation Counter.

The scintillation counter is becoming the predominant detector for counting applications largely due to its higher efficiency (typically between 40% and 80%).

#### 11.4.3 ANALOGUE OR DIGITAL SIGNAL CONDITIONING

The associated signal conditioning electronics can be either analogue or digital. As opposed to provious analogue based systems, recently released density gauges are now all microprocessor based. The differences are best discussed by considering the application of each type in commercially available meters.

# 11.4.3.1 Analogue Signal Conditioning

In analogue signal conditioning the output is 'calculated' by means of analogue signal processing.

As an example of analogue signal conditioning, consider the signal convertor used in the Krohne Density Gauge[Krohne, 1984]. Figure 23 on page 94 shows the block wiring diagram for the Krohne DH60 Radiometric Density meter.

Gamma Ray Instruments in an Industrial Environment



2 Massuing section 2 Massuing section 3 Scintillation counter 4 High-voltage supply 5 Control angibiler 6 Readver 7 Putes shaper 8 Integrator 9 Zaro suppression 10 Current output 11 Camperature compensation unit 12 Inesistance thermometer P1 100 13 Linearization unit\*

1 Radioactive source

15 Indicating instruments (optional)

#### Figure 23. Krohne DH80 Block Wiring Diagram

The measuring head (C) contains the precision high-tension generator (4) for the supply of the photomultiplier (3) and signal amplificar (5), which compensates for temperature and ageing effects upon the counter tube. In addition, the multiplier pulses are normalized in the control amplifier and conducted at low immediance to the subsequent signal converter.

In the signal convertor a Schmitt-Trigger (5) regenerates the incoming pulses. The resulting fast trining negos trigger a monostable multivibrator (7) whose pulse width can be adjusted to compensate for the isotope decay or temperature affects (11). The integrator (5) connected to the monostable multiple on the second scheme and the second scheme an

The non-linearity of the pulse stat change with density, while becomes predominent when the product of the density change times measuring length becomes very large, is solved by the pulge in linearisation (13) mirroit. This circuit simulates the error function and, by means of compensation, reduces the measuring error to a value within the measuring tolerance.

Gamma Ray Instruments in an Industrial Enviroment

Important disadvantages of analogue electronics include the following

- A major limitation is that the signal convertor requires frequent calibration as a result of component drift with time.
- 2. Calibration is also required to change the engineering units of the output.
- 3. All i \_\_\_\_\_\_ limits, celibration values and time decay, linearization and te. trature compensation values are set by potentiomaters. This method is susceptable to celibration errors and component drift.

# 11.4.3.2 Dir + Signal Conditioning

Unlike analogue systems, microprocessor based systems mathmetically calculate the specific gravity using the theoretical equation. As the system is microprocessor based, it allows for greater flexibility and the incorporation of settra features.

As an example of digital signal conditioning consider the signal convertor used in the MCI Digital Density Gauge[MCI,1984] The MCI signal convertor consists of two modules.

The Detector Hieroproc-seor Module comprises of a scintillation detector to convert gamma rays into witchical pulses, and a microprocessor with imput circuits to recaive the alectrical pulses, calculate S0 or percent solids and provide the required outputs to an isolated 4-20 and current loop (and other outputs). The operating program is stored in ROM and opcrating constants are as: nod in non-voltile RAM.

A communications module comprising of a keyboard and display to allow the operator to communicate with the microprocessor is provided. All communications with the processor are in key-coded instructions, entered through the keyboard. This module is normally located inside the detector/microprocessor woduls, but it may be remote in a separate enclosure.

The MCI density gauge has the following inputs and outputs :

- A LCD display which can be set to display count rate, SG or percent solids, or mass flowrate.
- A fully isolated 4-20 mA output selectable between SG and percent solids.
- 3. A fully isolated 4-20 mA output for solids mass flowrate.

Gamma Ray Instruments in an Industrial Environment
A serial current loop to drive an external device. This output is selectable between SG/percent solids and mass flowrate.

5. A 4-20 mA input from an external flowmere:

The following calculations are performed during operation of the density gauge.

Calculation of Specific Gravity

The basic gamma ray absorption equition is -

I=lo.exp(-U.SG.L)

(11.7)

'n

where

I = strength of radiation beam emerging from the fluid.

Io = strength of radiation beam entering fluid.

U ≈ Mass attenuation coefficient for fluid.

SG = SG of the fluid.

 $L \approx$  path length of gamma rays in the fluid.

The HOI gauge does not measure the number of pulses entering the detector over a fixed time pariod, but rather measures the rite at which pulses are detected. Thus, Jo is the count rate with no fluid present and I the count rate with the fluid present. To measure the court rate, the user anters the desired count (100 000 is recommended) to ensure an ecceptable statistical detected in the gauge them measures the time taken to reach us specified value and calculates the count rate which is then used in subsequent calculations. The problem with this technique is that the period between measurements is not fixed. As the fluid density increases fewer genera rays will reach the detector and a longer time period will be required to accumilate the desired count. The MOI gauge also implements a noving average over the last four measures not. This further reduces the standard deviation and also serves to swoth out random fluctuations in the solury density.

It was shown in section 3 of this chapter that to calculate the specific gravity directly from the gamma ray sebsorption equivation requires a Prowledge of Io. In the NCI gauge, Io is not avesured directly, but inferred from another measurement. It is known that the S0 of water is 1. Therefore, to calculate Io the pipe is filled with water and the count rate is neasured. Therefore for water, uring eqn. 1.7. We have

Cw = Cs.exp(-U.L)

(11.8)

where

Cw = count rate for pipe filled with water

Gamma Ray Instruments in an Industrial Enviroment

Cs = Count rate for empty pipe

Rearranging eqn. 11.8 we have -	
Cs = Cw.exp(U.L)	(11.9)
Using eqn. 11.7 the SG for a slurry is -	
Csl = Cs.exp(-U.SG.L)	(11.10)
where	

Cs1 = count rate for pipe filled with slurry

Substituting for Cs using Eqn. 11.9 we get -

Csl = Cw.exp(U,L).exp(-U,SG.L) (11.11)

Rearringing eqn. 11.11 and solving for SG we get -

$$S_{i}^{c} = 1 + (\ln C_W - \ln C_S 1)/OL$$
 (11.12)

where

Gw = count rate with pipe filled with water.

Csl = count rate with pipe filled with slurry.

U = Mass absorption coefficient for the slurry.

SG = SG of the slurry.

 $L \approx path$  length of gamma rays in the slurry.

Eqn. 11.12 is the equation implemented in the MCI density gauge.

In order to calculate the SG, the gauge must be standardised on a pipe filled with water. The count rate for this measurent is then taken to be a constant. The count rate can however change as a result of the source decay, dirt buildup on the walls (decreases L) or wearing of the walls (increases L). Therefore the gauge must be regularly re-standardised.

So that it is always possible to re-standardisa the gauge without having to re-fill the pipe with water, the concopt of standardising with the pipe empty has been introduced. An absorber is inserted into the bass in the source holder when the lever is in the STANNARDISF position, and the count rate for the empty pipe is measured. This is the AIX-STANNARD count rate. The microprocessor than implements the equation -

Cw = \$1.\$3/\$2

(11.13)

where

Cw = corrected water count rate.

Gauma Ray Instruments in an Industrial Environment

S1 = intitial measurement of water count rate.

S2 = initial measurement of air standard.

S3 = latest measurement of air standard.

In effect the water count rate is scaled according to the change in the air count rate. The initial air count rate and the initial water count rate are taken during installation.

The sutematic source decay also operates on the 53 value. This value is accreased by a small amount every three days. Whenever a new 53 value is externally entered the source decay factor is restored to one and the correction process starts anew.

#### Calculation of Percent Solids

The microprocessor is programmed to output percent solids on request.

$$PS = 100.SS.(SG - 1)/(SG.(SS - 1))$$
 (11.14)

where

PS = percent solids by weight.

SS = specific gravity of dry solids.

SG = slurry specific gravity.

Calculation of Solids Mass Flow.

The microprocessor is programmed to calculate solid mass flow by accepting an analogue 4-20 mA input from an external flow meter.

SF = F.(SG - 1).SS/(SS - 1) (11.15)

where

SF = solids mass flow in tonnes/hr

SG = specific gravity of slurry.

SS = specific gravity of dry solids.

F = volume flow of slurry in cu.m/hr. (from external flowmeter.)

Microprocessor based signal conditioning has several advantages including the following.

Gamma Ray Instruments in an Industrial Environment

**1** . Mar

- Being digital the meter is free from drift associated with analogue components.
- 2. The theoretically exact density equation is implemented with no requirements for extras such as linearisers to provide for precise output. The constants in the formula are easily altered to allow for different settings. This would not require recalibration as would be the case in an analogue system.
- 3. It is simple to calibrate. In many cases the calibration can be predicted exactly in advance of installation as the value of the mass attemnation coefficient can be calculated from a knowledge of the process material and all other constants can be obtained from the customers requirements.
- 4. Additional foatures are easily included. It is common for measurfactures to include a digital display and mess flow computation capability into the meter. The data convertor can accept an analogue input signal from a filow meter, multiply the flow by SG, and output solids mess flow on a separate 4-20 ak isolated current loop.
- All ranging, calibration, display and automatic source decay compansation can be done in software, thereby eliminating complex and frequent calibration.
- 6. All constants, standards and control parameters can be set independently and changed at will, using single keystrokes on the module keyhed. Outputs of all variables can be actomatically ranged over pressel HI-LO values. For example, assume the SG ranges between 1.0 and 1.5. To scale the output so that 4 mA corresponds to a SG of 1.0 and 20 mA to SG of 1.5 the user enters the constants into ond and 20 mA to SG of 1.5 the user enters the constants into changed such that the upper SG limit is now 2.0, to rescale the output the user only mak to re-enter the upper constant. This is amajor advantage over an analogue instrument which would typically require costly and time consuming resolithration.
- The pulse counting period is easily altered. This allows counting statistics can be controlled by requiring a preset number of counts to be accumulated before such calculation of SG.
- 6. Previous counts can be stored and thus a data set developed. This allows random short term changes in the slurry density to be smoothed out, by outputting a moving average over the data set and theraby increasing the accuracy of the meter.
- The dynamic range of the detector is extended by using the microprocessor to correct for the dead time.
- 10. The gauge can be operated as a digital ratemeter. To do this the gauge is put into the ratemeter mode and a count limit entered. The time to reach the count limit is measured and the count rate is calculated and displayea.
- A test signal can be generated internally for use in checking operations.

Gamma Ray Instruments in an Industrial Environent

- The printed circuit boards can be designed for fault finding using signature analysis techniques.
- It is possible to incorporate PID control with the parameters being set by a supervisory computer.

#### 11.5 CONSIDERATIONS IN DENSITY GAUGE APPLICATIONS

The following guidelines must be borne in mind when using density gauges. [Williams, 1979]

1. Air in line.

Air bubbles are a nuclear gauge's worst enewy as air represents an uncontrolled second variable with dramtically different gravity (0.001 SGU when related to water). Extreme care must be taken to eliminate air. This is importent when using contrifugal pumps which oftem suck in air through the seals. This problem can oftem be eliminated by directing a continuous stream of water onto the pump shaft seal.

2. Full pipe flow.

The pipe must be full of process material. Therefore a desirable location is under a positive head or where flow is upward.

3. Stratification.

If settling of the slurry may occur best results are obtained by directing the radiation bear vertically instead of across the pipe, or installing at a pump discherge.

4. Sampling means.

Since sampling is often required for calibration, intelligent installation of sampling valve(s) is hacessary in the vicinity of the gauge. If stratification can occur, multiple taps are desirable.

5. Bypass valving for agglomerating streams.

If the slurry has sticky properties permitting build-up, a bypass must be installed to allow periodic cleaning as build-up on the pipe will cause gauge calibration changes and drift.

6. Vibration

Gamma Ray Instruments in an Industrial Enviroment

When the gauge is located very close to a heavily vibrating pump or compressor, manufacturers may recommend neoprene jacketing or the like to reduce hemmering. However vibration must be severe to require this.

7. Corrosion in pipe.

Changes in pipe wall will cause gauge drift, since the density of metal is great with respect to the liquid. Periodic recelibration will remove this affect as a source of error.

8. Temperature compensation.

Temperature compensation can be provided, by means of an external temperature sensor, if the density varies greatly with temperature.

9. Sampling techniques.

If calibration by sampling is necessary it must be ansured to at the sampling cup is properly cleaned, the operator does not jur so the value thready not getting the thicker particles, or overfl the cup. Typically 5 to 10 samples per point must be taken to ensure a representative measurement.

10. Effect of solid's specific gravity on change

If the gauge is calibrated in "solids the essumption is made that the specific gravity of the solids is reasonably constant.

If the SG varies from 3.5 to 4.2 SGU the percent solids indication will rise 4% of full scale when the gauge is calibrated 40% solids as zero, and 70\% solids as full scale. If the gauge is to be used for mass flow this effect can be ignored.

11. Mass flow calibrations.

The span of the density gauge is usually calibrated to be 1.00 for 0% gauge output so the mass flow multiplier will be 0 for 0 solids. Linearity correction is also usually required as there is often a wide span and therefore non-linearity.

#### 11.6 CONCLUSION

This chapter discussed the application of nuclear physics and radiation in commercial nuclear density gauges.

Three sources are available for use, with Ceasium-137 being most commonly used as it provides the best balance between half-life and penetration

Gamma Ray Instruments in an Industrial Environment

~ #

When the gauge is located very close to a heavily vibrating pump or compressor, manufacturers may recommend neoprene jacksting or the like to reduce hammering. However vibration must be severe to require this.

7. Corrosion in pipe.

Changes in pipe well will caus, gauge drift, since the density of metal is great with respect to the liquid. Periodic recelibration will remove this effect as a source of error.

8. Temperature compensation.

Temperature compensation can be provided, by means of an external temperature sensor, if the density varies gre.tly with temperature.

9. Sampling techniques.

If calibration by sampling is necessary it must be ensured that the sempline properly cleaned, the operator does not just crack the v not getting the thicker particles, or overflows the cup. Co 10 samples per point must be taken to ensure a represe.

10. Effect of solid's specific gravity on change

If the gauge is calibrated in %solids the assumption is made that the specific gravity of the solids is reasonably constant.

If the SG varies from 3.5 to 4.2 SGU the percent solids indication will rise 4% of full scale when the gauge is calibrated 40% solids as zero, and 70% solids as tall scale. If the gauge is to be used for mass flow this effect can be ignored.

11. Mass flow calibrations.

The span of the density gauge is usually calibrated to be 1.00 for 0% gauge output so the mass flow multiplier will be 0 for 0 solids. Linearity correction is also urually required as there is often a wide span and therefore non-linearity.

#### 11.6 CONCLUSION

This chapter discussed the application of nuclear physics and radiation in commercial nuclear density gauges.

Three sources are (vailable for use, with Cassium-137 being most commonly used as it provides the best balance between half-life and penetration

Gamma Ray Instruments in an Industrial Enviroment

capability. The source size is subject to several factors and is usually calculated on saturfacturess propiatry expirical formulas. Two detectors are commonly used, the ionization chamber and the scintillation counter which is gradually supersceding the former. The prime advantage of the scintillation counter is that it allows a much smaller source size and a collimated beam to be used compared to instruments using ionization chambers.

The most striking difference in commercial instruments is in the signal conditioning electronics. Most commercial instruments are still based on analogue electronics, but mitroprocessor based systems are being introduced. The advantage of the latter is that all computations are performed methematically which allows for much greater fixethilty and versatility. The use of a microprocessor also allows for a better man methins interface with recalibration being performed via keypad rather than by means of potentiometers. As microprocessor based systems become more common, self-calibrating and self-diagnostic instruments will become the norm.

This concludes the review of the principles behind the instruments capable of meeting the project requirements. The following chapter discusses the evaluation of a mass flow system based on a magnetic flowmeter and a nuclear density gauge.

Gamma Ray Instruments in an Industrial Environent

#### 12.0 EVALUATION OF KROHNE MASS FLOW SYSTEM

#### 12.1 INTRODUCTION

In order to explore a commercially available nuclear-based wass flow system, this chapter discusses the Krohen Hass Flow System which wes ande available for evaluation. A brief description is given of the system, and this is followed by a descalled description of the test rig and methodology employed in evaluating the system. Finally the test results are analysed in detail.

#### 12.2 OPERATING PRINCIPLES

The Krohns Mass Flow system used in this project consists of an AC magnetic flowmeter and a nuclear density gauge. The source and the detector of the nuclear density gauge are mounted onto a section of pipe which has the same internal diameter as the magnetic flowmeter. The pipe section and the magnetic flowmeter are holced together.

The signal conditioning for both the magnetic flowmeter and the nuclear density gauge are performed in separate field housings. The primary and field housing of the magnetic flowmeter are connected via shielded cable as is the detector and field housing of the density gauge.

The systam was supplied in the following configuration:

- 1. A Nuclear Density Gauge comprising of
  - a. A rubber lined measuring section of diameter 150mm. The nuclear source and detector are mounted onto the measuring section, thus ensuring accurate alignment of the radiation beam.
  - b. A 10 milliCurie Caesium-137 source contained in a shutter.<sup>4</sup> housing. The shutter may be locked in either the open or closed position. In the open position the source is exposed and a collimated beam passes through the material in the pipe and onto the detector.
  - c. A Scintillation Countyr detecto. which allows for the very small source size. The photomultiplior tube and associated high voltage supply is built into the detector housing.
  - d. A data convector mounted in a field housing. The data convector supplies a regulated + 15 Volt supply to the detector and receives the output pulses from the detector. These pulses are then converted into a scaled 4 - 20mA current output. The system was calibrated such that wank corresponds to an SG of 1.5 and 20mA to

Evaluation of Krohne Mass Flow System

an SG of 2.0. (See Chapter 11 for a description of the signal conditioning electronics.) An ammeter with a LCD display is mounted on the field housing. The armster displays the loop current as a percentage of full scale where 20mA corresponds to 100% fullscale.

- e. A galvanic isolator to isolate the current loop from the receiving instruments. This is used primerily to increase the compliance of the current loop. The current loop on the converter can drive a maximum load of 1000 chms.
- 2. A Nagnetic Flowmeter comprising of
  - a. A X2000 Fused aluminium oxide measuring head employing AC excitation and a pipe diameter of 150mm. The fused aliminim oxide liming is a recent Krohne development end is extramely abraeion and comportature (up to 180 degree Celsius) resistant. The X2000 head has a separate mains connection for the AC excitation voltage. The measuring head used in the evaluation system had a is bar maximum pressure rating, as this is the highest pressure rating of took icema.
  - b. A field housing type TIV 50/F 1 containing the data converter electronics. The data converter is connected to the measuring section by a shielded cable. The data converter receives the induced signal voltage from the electrodes, as well as an in-phase reference voltage obtained from the excitation voltage via a small transformer. The signal voltage is then scaled to a 4 - 20mA output where 4mA corresponds to a flow of Om/s and 20mA to a flow of 1 m/s. The converter has a range potentiometer which liews the maximum flowrate scale to be changed. Changing the maximum flowrate e.g. to 3m/s does not require recalibration of the meter. An analogue ammeter is mounted on the converter which measures and displays the loop current as a percentage of full scale where 20mA corresponds to 100% (full scale), and 4mA corresponds to 20%. Two jacks are provided to insert an external emmeter into the current loop to allow for zero setting. The meter provided with the converter is not suitable for the setting of the live zero (4mA).
- 3. An Analogue Mass Flow Multiplier.

This receives a 4 - 20mA current from both the flowmeter and the density gauge and outputs their product as a scaled 4 - 20mA signal corresponding to the dry solids mass flowrate.

i.e.  $I_{mf} = K.I_d.I_f$ 

where

 $I_{mf} = Output current corresponding to Dry solids mass flowrate$ 

I = Current input from density gauge (Percent solids)

I = Current intput from flometer (Volumetric flowrate)

Evaluation of Krohne Mass Flow System

K = constant equal to  $SG_{dry solids} \cdot K_1$  $K_1 = Output current scale factor$ 

The complete system block diagram is shown in Figure 24.



Figure 24. Mass Flow System Block Diagram

#### 12.3 THE TEST RIG

The test rig is shown in Figure 25 on page 106.

The system piping is connected in a loop configuration. A double acting piston pump with a pump capacity of 45 m/hr and maximum head of 75 her is fad from a local storage tank. The meterial is pumped through the pipe (approximate length of 200 meters), up a vertical section of pipe and into a hopper mounted on loadcalls. The hopper is mounted directly above and feeds into the local storage tank of the pump. The mass flow system is mounted onto the vertical section of pipe. The flowmeter zero is set by scorping the jumps as no curoff valve ves installed into the pipe.

Evaluation of Krohne Mass Flow System



#### Figure 25. Schematic of Test Rig

A pressure translucer was installed to monitor the pressure fluctuations. As the flowmeter has a lower pressure rating (16 ber) than the pump head, it is mocessary to install the flowmeter at the end of the pipe Section. In an actual installation, the flowmeter must be sufficiently rated to be installed immediately after the pump to prevent time delays in the control system.

The outputs from the pressure transducer, l-adcells, magnetic flowmeter and density gauge are connected to thert recorders for data acquisition purposes.

The density gauge and flowmeter outputs were also connected to a single chip BASIC microcomputer (the INTEL SOLAR). Interfacing was by means of as bit analogue to digital converter. The system was used to scale the current outputs of the density gauge and flowmeter, thus resulting in a considerable timm saving. The data logging program Also performed a moving average over the last one hundred flowmeter readings, thus allowing the average incorrect to be docummined.

Evaluation of Krohne Mass Flow System

## 12.4 TEST METHODOLOGY

The esphesis in the tests use on proving the principles of the instrumentation. Therefore, the primary consideration was the correlation between the measured values and the expected values. A good indication of the correlation as given by the linearity of the curve. Of secondary importance was the accuracy of the krohme instrumentation, the measured values must be compared to a traceable standard. It is also necessary that the accuracy of the data logging equipment exceeds that of the measured values must be compared to a traceable standard. It is also necessary that specified to 0.5% accuracy, (traceable to Durch Standards), whereas the resistors used in the test rig to convert the 4 - 20 mA current to a voltage term only of 1% accuracy. Thus, the accuracy of the instrumentation could only be determined to limits in the accuracy of the data logging equipment.

load cells were used to determine the slurry mass flowrate. The hopper was emptied into the local storage tank and then allowed to fill up. After a measured time period of the mates of the material is measured and the hopper was again emptied into the local storage tank and the cycle repeated. Therefore

# $Flowrate(m<sup>3</sup>/hr) = \frac{\Delta M(mV) \times Cal(kg/mV) \times 3600(sec/hr)}{\Delta T(sec) \times SG \times 1000(kg/m<sup>3</sup>)}$

Where the calibration factor for the locdcells was 1.606 kg/mV. Note, to assess the flowmeter, an accurate value for the SG must be obtained.

The specific gravity of the material is determined by taking samples of the material at regular intervals, after which the wet susples were weighed. The samples were then dried and reweighed. The SG is then given by the following formula:

 $SG_{siurry} = 1/(1 - C_W + C_W/SG_{dry solids})$ 

Where

SG<sub>slurry</sub> = Slurry SG

SG dry solids = Dry solids SG

C., = Concentration by weight

= (sample dry mass)/(sample wet mass)

The above measurements were then compared to the corresponding measurements taken from the magnetic flowmeter and nuclear density gauge.

The evaluation of the mass flow system was performed using a goldslime slurry, with a solids concentration between 50, and 80%, which corresponds to a SG range of 1.5 to 2.0. However, at the lower SG range, the slurry becomes a dilute slurry and is no longer in a pasts form.

Evaluation of Krohne Mass Flow System

#### 12.5 TEST RESULTS

The major results are the comparison of the measured flowrate to the actual flowrate (as determined by the load cells) and the measured SG to the actual SG as determined by weighing. The results of the tests are shown in the following graphs.

Figure 26 on page 110 shows the measured flowrate versus actual flowrate for several different slurry specific gravities. Irrespective of the solid concentration there is a strong correlation between the measured and actual values. The flowenter is accurate (as determined by the leadcalls and 80) to within 5% of the expected value, thus satisfying the project requirements. The metar does however consistently gives a slightly lower treading than the expected value. This is possibly due to the zero an accurate reading solutions, thus possibly indicating that the flowenter was only reading lower than expected during the goldbline tests. Tests over a longer time period will have up any consistent inaccuracy.

Figure 27 on page 111 shows the measured specific gravity versus actual specific gravity for several different flowards. Again, there is a strong correlation between the measured and actual values. The measurement of the S6 is also independent of the flowards (See Appendix C) at the higher S6 zenges. At the lower S6 ranges, the demity rises with increasing flowards. The reason for this is that the slowry is no longer a paste, but is now a dilute slurry and setting of the solids occurs at the lower flowards. The reason for this increased, more solids are carried in suspension and the density gauge correctly reads the corresponding rise in 56.

As it was necessary to calibrate the density gauge, the gauge accuracy is determined by the calibration values. As the same method was used to calibrate and evaluate the density gauge, a quantitative assessment of the accuracy cannot be given.

As seen in Figure 26 on page 110 and Figure 27 on page 111 the mass flow system meets the project requirements in terms of the accuracy of the measured values. However, there are two other factors which influence its suitability to the project.

- Figure 28 on page 112 above a chart recording of the pressure fluctuations in the pips. It can clearly be seen that the pressure is highly impulsive and the mass flow system must be capable of withstanding the impulsive pressures produced. The mass flow system must also be capable of withstanding the severe vibration resulting from the pressure impulses.
- 2. Figure 29 on page 113 shows the chart secordings of the flowmeter output at several different pump stroke rates. When the pump is operating at its lowest capacity the stroke rate is 5 strokes/min. At the maximum pump capacity, the stroke rate is 17 strokes/min. When pumping high concentration slurries, the slurry momentarily stops

Evaluation of Krohne Mass Flow System

flowing during the piston changeover pariot. This can clearly be seen on the chart recoder. (Note, the chart recorder reading so not fail to zero as the flowmeter has a 10 sec. time constant.) For the flowmeter to up the the series are adding tha time constant. The the increased. The time constant was be at least twice the duration of the pump stroke intervals at the slowest the sums speed.

#### 12.6 MASS FLOW SYSTEM EVALUATION

The mass flow system is clearly capable of providing a manurement of the slurry mass flow and solids concentration. There is a strong correlation between the resourced and expected values, clearly indicating the validity of the measurement technique. The only adjustments to the flowmeter was the setting of the magnetic flowmeter zero. This is however a periodic requirement and a characteristic of all present AC excited magnetic flowmeters.

During the evaluation of the flowmeter, the zero drifted noticeably between tests. During operation with the coll and flyash slurries, the zero, once set remained relatively stable. However, during the actual evaluation of the flowmeter using a goldslime slurry, the zero drifted with a change in concentration of the slurry. The drift was more pronounced when the slurry was in a dilute phase. The drift would lead to errors in the measurement of the flowrate if a wide variation in the concentration is required. However, thuther tests would be not cossary to determine if this is a charecteristic of all AG axcited magnetic flowmeters, or only a characteristic of the test model.

Although the system was supposed to be precelibrated, on taking delivery is was found that the density gauge was out of calibration. Calibration of the gauge on site is possible and was undertaken. However, the calibration process is extremely time consuming and error prone. This is particularly true at the higher SG range (a solids content in the region of 80%), where the slurry essumes a pasts form, as it becomes difficult to six in dry material and to obtain a representative sample. The difficulty in calibrating an analogue density gauge clearly outlines the advanteges to be gained from a microprocessor based system.

Calibration is futher complicated in high concentration slurry systems, as it is not possible to provide a bypass for the instrumentation, since solidification of the slurry in the dead leg can occur.

In order to employ the system in a working installation, the time constant of the flowmeter must be increased to at least twice the period of the lowest stroke rate. This is a consequence of the use of double acting piston pumps and the high viscosities of the high concentration slurry. Failure to include a sufficiently long time constant can result in fluc-

Evaluation of Krobne Mass Flow System



Evaluation of Krohne Mass Flow System





Figure 28. Slurry Pressure Recordings

tuations in the control system. (Note: This is only true if positive displacement pumps are used. Contrifugal pumps will give a fairly uniform flowrate.)

A more important reservation in the use of the system is the capability of the mass flow system to withstand the high pressures and vibration in the slurry pump system. During the test period of approximately 2 months duration the system showed no suscepticability to vibration, but the long term effect has not been determined. A more artitical consideration is the pressure rating of the mass flow system. The system must be capable of withstanding the high avarage pressures (up to 100 bar) and the periodic peak impulsive pressures. The system offered for evaluation uses a low

Evaluation of Krohne Mass Flow System

i



Figure 29. Slurry Velocity Profile

pressure system and it is still necessary to determine the capability of a rated meter to withstand the pressure impulses.

It can be noted that during the course of the system we well ation, the system wes used on cosl, flysh and goldslime slutries. The 86 of the slutries varied widely (up to 1.3 for cosl, 1.5 to 1.7 for flyssu, and 1.5 to 2.0 for goldslime), however the solids cantent always ranged between 50% and 55%, in slit cases the flowmater showed no deviation in its capability to accurately measure the flowmater. The density gauge, once correctly calibrated, correctly measured the 80 in the range 1.5 to 2.0.

The mass flow multiplier was not used and therefore not tested. This however has no bearing on the tests as the multiplier is an external device to the Krohne \_ruivment, and does not affect the measurement of the volumetric flowmater  $u_{\rm S}$  SG, i.e. the operation of the either magnetic flowmater or the der.its gauge.

#### Evaluation of Krohne Mass Flow System

#### 12.7 CONCLUSION

The Krohne wass flow system offered for evaluation met all the measurement accuracy requirements of the project. The system offered for evaluation incorporates a nuclear density gauge based on an analogue signal converter and thus does not offer the flexibilty of the newer microprocessor based systems.

Before the system can be installed in a working installation the following must be performed:

- The time constant on the magnetic flowmeter must be increased to compansate for the periodic fluctuations in the system resulting from the double acting piston pump. The time constant must be at least twice the period of the slowest atroke rate, or preferably field ... justable.
- The mass flow system must be capable of withstanding the severe vibration encountered in the pipe line.
- 3. The mass flow system must be capable of withstanding the high average pressure the pump is capable of producing (up to 100 bar), as well as the periodic impulsive pressures in the slurry system.

The above considerations are all applicable to the manufacture of the mass flow system and not to the principle or operation of the system. The tests have shown that AC excited magnetic flowmatters and a nuclear density gauge are any itable for the measurement of the mass flowrate and concentration of pasts slurries and can used if the above mentioned critaria are satisfied.

Ł.

Krohne Mass Flow System

#### 13.0 CONCLUSION

This report discussed the choice of instrumentation suitable for measuring the concentration and mass flowrate of high density 'past's lurriss. The slurry system itself is unique and still in the development stage. The uniquesses of the system results from the high density of the slurry and the use of pneumatically conveyed powder, in this case cament. Like many new techniques, the requirements and characteristics of the system fash outside the general lists of central hardware. The system fash outsides the general lists of central hardware. The system this results in the meter having to be specially made up which in turn results in high prices and long delivery times and lists the evailability of off-the-shelt replacements.

The most important point arising out of the project, is that an instrument is only as good as the principles on which it is based! Hamy principles are used in the design of related instrumentation, e.g. flowmeters, and thus the choice of instrument instruly requires a choice of measuring principle most suited to the application. Once a suitable principle has been chosen, a specific instrument modelying that principle can then be related or designed. A major matrix of this project concerned the survey reasiled in the oppoint flow of a superior flowmetic and a mulcal designed. The principle can be and stygenzy. The principles, characteristics and application of these instruments year than considered in fetal.

Magnetic flowmeters were specifically designed for measuring the flowrate of slurrise and no degradation in performance was found in high density slurry applications. In the project only an AC excited flowmeter was evaluated. Although DC meters have amay davintages, they suffer in performance when used in slurry applications, due to noise being induced when the slurry particles collide with the alectordes. However, high concentration slurries assume a homogeneous form and are usually pumped at low velocities. These factors may make it possible to use DC excited flowmeters for high concentration slurries, and this probability should be accounted on the state of the state of DC excited megnetic sciences. A magnetic flowmeter, namely the Kent Veriflux [Kent.1985], its - sciently bean developed which combines both AC and DC opecation. It is - aimed that this meter has the advantages of both types of excitation, i.e. autors are and explicit on in slurry explications.

Nucl: ar instruments are capable of a variety of measurements and can be extensively used in the process control industry. There is between a lack of knowledge relating to the principles, use and applications of nuclear instruments. To rectify this problem an in depth study into the theory boind nuclear 'varruments was undertaken. The latter chapters of this project report serve as a guide to the use and application of these instruments.

Conclusion

Nuclear instruments are now entering a second phase, and in line with other instrumentation are becoming more intelligent and easier to use. The use of advanced detectors are minimizing the source size and the health heard, thereby increasing their acceptability. The use of microprocessor based electronics has resulted in an integrated meter. This makes it possible to satisfy all the project requirements with a flowester and an intelligent density gauge. The newer density moters are not yet freely available in South Africa and so it is advised that they be eveluated in the mean time to allow for their repid introduction when their availability increases.

An important point arising out of the project is the signal conditioning electronics used in instrumentation. Hany downtages are to be gained by asking the electronics microprocessor based. The most important being the flexibility, ease of use and better manmachine interface. As microprocessor systems become more common, more advanced features such as auto-calibration, self-diagnosis and advanced signal processing can be incorporated. These factures will improve the instruments employing them, with a corresponding increase in performance of the associated control system.

This project concentrated on the instrumentation for the slurry. Before the backfill system can be implemented, instrumentation must be found to measure the mass flourate of pneumatically conveyed powders. A possible solution is to use a nuclear density gauge to determine the effective cross-sactional area of the powder, and a doppler or a vortex shedding flowmeter to determine the average volocity of the powder particles.

Conclusion

#### APPENDIX A. GLOSSARY OF NUCLEAR TERMS

- Alpha Particle Nuclear particle of mass 4 units carrying 2 units of positive charge. Will not penetrate skin, and ranges only a continetre or so in air. Emitted by radio's topes such as uranium 238, thorium 225, plutonium 210, etc.
- Beta Particle Nuclear particle of mass 1/1860 units carrying 1 unit of negative charge. Will panetrate skin and has a range of up to fm in air. Emitted by many radioisotopoes.
- Becquerel (Bq) The measure of quantity of radiosctivity. A substance has an activity of 1 Sq if 1 atom of it disintegrates every second. This is the new S1 unit gradually replacing the curie. 1 Bor - 2,7x10<sup>-11</sup>, curies.
- Bremsstrahidng Radiation produced by the rapid deceleration of a beta particle, such as when it is caused to bombard a heavy material. The radiation is identical to X-radiation. Bremsstrahing is therefore an additional heard which may arise when handiing beta-emitting radioisotopes.
- Contamination Radioactive materials released from their normal containment into the working environment. It may be present on working surfaces or airborne.
- Curie A measure of quantity of radiosctivity. 1 curie of radiosctivity is such that  $3.7 \times 10^{10}$  atoms of it are disintegrating every second. This unit is gradually being replaced by the becquerel.
- eV Electron volts. A measure of energy of a particle or radiation. 1 eV is the energy attained by an electron when accelerated between them.
  voltage of 1 volt between them.
- Element A substance which cannot be chemically decomposed into simpler substances.
- Electron Atomic particle of mass 1/1860 units carrying one unit of negative charge. Differs only from the bate particle in its origin, which is from outside the nucleus of the stom.

Appendix A. Glossary of Nuclear Terms

Nuclear rediation. Will penetrice considerable b\*-oknesses of lead or concrete and has a very long rang. In airs. Emitted by many redicisotopes in association with either an alpha or a beta particle. Gamma radiation is electromagnetic rediation with energy of the order of MaV.

Half-life The time in which half the amount of stoms present in a given amount of radioactive material disintegrats. The value of the half-life is characteristic of a particular radioisotope and varies from millions of a second to millions of years.

Gamma

Ionization The process by which an electron is stripped from its parent atom. When radiation strikes an electron it loses energy in fonizing that electron.

Neutron Nuclear particle of mass 1 unit carrying no charge.

- Proton Nuclear particle of mass 1 unit carrying 1 unit of positive charge.
- Rad The abbreviation for "radiation absorbed dose". A measure of energy deposited in a medium.
- Rem The abbreviation for "rad equivalent man". A measure of the damage caused by exposure to radiation, and is gradually being replaced by the sievert (1 rem = 0.01 Sv).
- Sievert (Sv) This is the new SI unit replacing the rem. It quantifies the equivalent absorbed dose.
- X-radiation Radiation produced when materials are hombarded with electrons. X-radiation is electromagnetic radiation with energy of the order of eV.

Appendix A. Glosssry of Nuclear Terms

# APPENDIX B. MASS ATTENUATION COEFFICIENTS

The following table of mass attenuation coefficients, in  $cm^2/g_1$  for gamma rays were interpolated from values given in hubbell[1969]. The coefficients are given for a gamma energy of 0,662 KeV, which is the energy of gamma rangemen aradistion for Caesium-137.

1H	0.153	2 <sup>6</sup> Fe	0.0732
*C	0.0777	2 <sup>8</sup> Cu	0.0722
7N	0.0775	<sup>42</sup> Pb	0.1093
*Ū	0.0776	<sup>82</sup> U	0.0759
1'Na 13 <u>47</u> 1°Si NaI	0.0743 0.0744 0.0774 0.0759	Air Water Concrete Pyrex Glass	0.0861 0.0777 0.0772

The total mass attenuation coefficient for a compound or a mixture is:

$$\mu_{c}(m^{2}/kg) \approx \Sigma W_{i}\mu_{i}(m^{2}/kg)$$

where

 $\mu_{\rm c}$  = total mass attenuation coefficient for a compound or mixture

W, = weight fraction of ith element in compound

 $\mu_{i}$  = total mass attenuation coefficient for ith element.

Appendix B. Mass Attenuation Coefficients

## APPENDIX C. MASS FLOW EVALUATION TEST RESULTS.

# C.1 COMPUTED RESULTS.

The following values are the calculated averages for each test.

Flowmeter setting = 1.768DH80 settings: Zero = 7.279; Range = 2.581; Isotope = 0

TEST 1

SG(meas) = 1.959 ; SG(exp) = 1.966 ; Cw = 78.9 %

FLOW(meas)	FLOW(exp)
15.03	16.21
26.90	25.89
36.35	36.07

TEST 2

SG(meas) = 1.889 ; SG(exp) ≈ 1.886 ; Cw ≈ 75.5 %

FLOW(meas)	FLOW(exp)
16.23	16.95
21.03	22.22
27.49	27.17
33.35	34.88
38.50	39.88
41.24	43.15
1 1	

Appendix C. Mass Flow Evaluation Test Results.

## TEST 3

SG(meas) = 1.777 ; SG(exp) = 1.768 ; Cw = 69.8 %

FLOW(meas)	FLOW(exp)
12.03	12.47
18.34	19.74
31.38	31.77
40.90	40.79

TEST 4

SG(meas) = 1.670; SG(exp) = 1.6735; Cw = 63.7 %

FLOW(meas)	FLOW(exp)
14.06	15.38
19.36	21.65
25.27	26.77
31.16	34.00
40.31	40.21
43.20	41.52

TEST 5

SG(meas) = 1.567 ; SG(exp) = 1.567; Cw = 58.0 %

FLOW(meas)	FLOW(exp)
11.30	11.32
18.33	19.00
30.50	31.73
38.45	37.70
43.15	43.10
43,20	43.05
1	1

## Appendix C. Mass Flow Evaluation Test Results.

# TEST 6

SG(meas) = 1.552 ; SG(exp) = 1.553 ; Cw = 57.2 %

Γ	FLOW(meas)	FLOW(exp)
Γ	17.40	17.86
1	24.40	24.83
1	28.66	31.26
1	34.40	36.29
1	40.21	40.96
	41.14	42.60
-		

WATER TEST

FLOW(meas)	FLOW(exp)
14.19	14.18
19.96	20.72
33.61	33.53
39.13	37.86

C.2 READINGS FOR EVALUATION OF MAGNETIC FLOWMETER.

Loadcell calibration factor = 1.606 kg/mV - 385 kg.

Flowmeter setting = 1.768 ;

 $\label{eq:Flowrate(m^3/hr) = } \frac{\Delta M(mV) \times Cal(kg/mV) \times 3600(sec/hr)}{\Delta T(sec) \times SG \times 1000(kg/m^3)}$ 

Appendix C. Nass Flow Evaluation Test Results.

# TEST 1

Speed 1

SG(meas) = 1.958

LC emp(mV)	308.9	354.9	377.9	380.7	
LC full(mv)	398.0	443.8	463.7	468.6	
Time (sec)	17	15	15	17	
Flw(exp).SG	30.302	34.266	33.071	29.894	
Flow(exp)	15.41	17.42	16.82	15.20	
Flow(meas)	15.1	14.8	15.2	15.03	

# Speed 2

SG(meas) = 1.9558

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	250.9 393.5 21 39.26	371.4 457.0 12 41.24	377.3 468.5 13 40.56		
Flow(exp) Flow(meas)	20.07 20.0	21.08 21.0	20.07 21.1	19.8	_:_

Speed 3

SG(meas) = 1.959

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	243.9 401.2 18 50.52	226.4 409.4 21 50.38	229.5 407.0 20 51.31	===	
Flow(exp) Flow(meas)	25.78 27.0	25.71 26.7	26.19 27.0	=:=	

## Appendix C. Mass Flow Evaluation Test Results.

SG(meas) = 1.955

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	284.6 428.9 13 64.18	265.4 420.2 15 59.67	253.5 404.6 14 62.40		
Flow(exp) Flow(meas)	32.82 32.2	30.52 30.1	31.91 31.1	=:=	

Speed 5

SG(meas) = 1.9485 slight air in pipe

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	291.5 414.4 10 71.06	281.6 407.8 11 66.33	318.4 407.4 7 73.51	76,83	
Flow(ex;) Flow(meas)	36.46 36,3	34.04 36.1	37.72 36.4	_:_	

Speed 6

SG(meas) = \_.....

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG			`: :		
Flow(exp) Flow(meas)	_: <u>-</u>	; ; ;		='=	<u>=:</u>

SG(meas) = 1,8921

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	230.1 325.7 18 30.71	142.3 331.3 16 32.16	244.3 340.4 17 32.68	244.0 342.9 18 32.77	; ;
Flow(exp)	16.23	16.99	17.27	17.31.	ΞΞ
Flow(meas)	16.2	16.3	16.1	16.3	

# Speed 2

SG(meas) = 1.8921

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	242.9 352.7 15 42.32	245.1 344.9 14 41.21	263.9 367.2 14 42.66		
Plow(exp) Flow(meas)	22.36 21.0	21.78 21.0	22.54 21.1_	=:=	

Speed 3

SG(meas) = 1.8897

LC emp(mV)	241.8	240.7	239.0	245.2_	
LC full(mv)	341.7	345.8	394.7	375.3	
Tira (sec)	12	12	17	14	
Flw(exp).SG	48.13	50.64	52.95	53.73	
Flow(exp)	25.46	26.79	28.02	28.43	_:-
Flow(meas)	27.4	27.5	27.65	27.5	

## Appendix C. Mass Flow Evaluation Test Results.

SG(meas) = 1.8897

LC emp(mV)	246.1	273.4	247.9	257.8	
LC full(mv)	383.2	398.1	384.6	383.6	
Time (sec)	12	11	12	11	
Flw(exp).SG	66.05	65.54	65.86	66.23	
Plow(exp)	94.95	34.68	34.85	33.04	=:=
Flow(meas)	33.2	33.5	33.4	33.3	

# Speed 5

SG(meas) = 1.8848

LC emp(mV)	238.5	254.0	247.9	245,9	
LC full(mv)	418.0	426.7	433.8	422,7	
Time (sec)	14	13	14	14	
Fiw(exp).SG	74.13	76.81	76.77	73,01	
Flow(exp)	39.33	40.75	40.73	38.73	
Flow(meas)	38.4	38.6	38.7	35.3	

Speed 6

SG(meas) = 1.8848

LC emp(mV)	244.2	249.6	246.0	258.5	
LC full(mv)	429.7	421.1	423.4	455.5	
Time (sec)	13	12	13	14	
Flw(exp).SG	82.5	82.63	78.9	81.36	
Flow(exp)	43.77	43.84	41.86	43.16	
Flow(meas)	41.1	41.3	41.35	41.2	

Appendix C. Mass Flow Evaluation Test Results.

m	٣,	07	• •	
ж.	ь,	2,		

SG(meas) = 1.7990

LC emp(mV) LC full(mV) Time (sec) Flw(exp).SG	217.2 298.8 22 21.44	222.6 296.5 19 22.49	219.9 300.8 21 22.27		
Flow(exp) Flow(meas)	12.12 12.1	12.72 12.0	12.59 12.0	12.0	

# Speed 2

SG(meas) = 1.8014

LC emp(mV)	223.2	223.4	226.3	224.9	
LC full(mv)	350.3	340.5	339.4	341.5	
Time (sec)	21	19	19	19	
Flw(exp).SG	34.99	35.63	34.42	35.48	
Flow(exp)	19.66	20.02	19.34	19.94	1 _:_
Flow(meas)	18.36	18.33	18.33		

## Speed 3

SG(meas) = 1.8014

LC emp(mV)	229,6	229,9	232.2	236.7	
LC full(mv)	341.8	356.1	355.5	350.2	
Time (sec)	14	16	15	15	
[Flw(exp).SG	46.34	45,60	47.52	46.06	
Flow(exp)	26.04	26.63	26.71	25.89	_:_
Flow(meas)	24.1	24.1	24.0	24.05	

# Appendix C. Mass Flow Evaluation Test Results.

SG(meas) = 1.7990

LC emp(mV)	238.6	234.8	237.5	234.2	
LC full(mv)	401.9	406.9	408.0	415.8	
Time (sec)	16	18	17	19	
Flw(exp).SG	59.01	55.28	57.99	55.26	
Flow(exp)	32.96	30.88	32.39	30.87	_:=
Flow(meas)	30.50	30.60	30.70	30.70	

Speed 5

SG(meas) = 1.7965

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	249.2 433.5 15 71.04	245.1 433.9 15 72.39	248.6 431.7 14 75.62		
Flow(exp) Flow(meas)	39.69 39.5	40.44 41.7	42.24 41.46	41.6	41.0

## TEST 4

Speed 1

SG(meas) = 1.6617

LC emp(mV)	221.2	223.3	220.7	236.8	
LC full(mv)	307.5	308.5	308.3	309.0	
Time (sec)	20	19	20	16	
Flw(exp).SG	24.95	25.93	25.32	26.09	
Flow(exp)	15.01	15.60	15.23	15.7_	_:_
Flow(meas)	14.13	14.03	14.13	`_	

# Speed 2

SG(meas) = 1.6666

LC emp(mV)	230.7	236.4	230.5	228.7	
LC full(mv)	313.3	516.0	312.2	311.8	
Time (sec)	13	14	13	14	
Flw(exp).SG	36.74	37.00	36.34	34.32	
Flow(exp)	22.04	22.20	21.80	20.59	
''low(meas)	19.53	19.23	19.33	19.33	

## Speed 3

SG(meas) = 1.6670

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	232.0 327.6 13 42.52	232,4 327.7 12 45.92	235,8 330,2 12 45,48	; 	; 
Flow(exp)	25.50	27,54	27.28	_:_	_:_
Flow(meas)	25.03	25,13	25.43	_:_	

# Appendix C. Mass Flow Evaluation Test Results.

130

19月底
S	peed	4
- 62	beer	

SGCREAS	h	77	1.6715
DOCHERO			2.0/13

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	239.1 346.6 11 56.50	234.2 381.6 15 56.81	241.6 409.8 17 57.20	Ì.	
Flow(exp) Flow(meas)	33.80 30.83	33.98 31.03	34.22 31.33	31.53	_:-

SG(meas) = 1.674

LC emp(mV)	244,4	247.0	251.1	239.8	
LC full(mv)	42230	429.1	418.6	421.9	
Time (sec)	16	15	14	16	
Flw(exp).SG	64.18	70.19	69.17	65.80	
Flow(exp)	38.33	41.92	41.32	39.30	
Flow(meas)	40.0_	40.6	40.1	40,5	

Speed 6

SG(meas) = 1.6789

LC emp(mV) LC tul1(mv) Time (sec) Flw(exp).SG	247.4 428.5 15 69.8	240.4 413.7 14 71.57	241.9 429.5 16 67.79	
Flow(exp) Flow(meas)	41.57 43.1	42.62 43.3	40.37 43.2	 ÷

TEST	5
------	---

SG(meas) ≈ 1.5686

LC emp(mV) LC full(mv) Time (sec) Flw(ext).SG	210.4 272.0 20 17.81	213.5 268.6 18 17.70	222.5 277.9 18 17.79	=	
Flow(exp) Flow(meas)	11.35 11.2	11.28 11.3	11.34 11.4	_:_	

#### Speed 2

SC(meas) = 1.5784

LC emp(nV) LC full(mv) Time (sec) Flw(exp).SG	221.0 308.1 17 29.62	224.3 317.5 18 29.94	218.3 319.5 19 30.79	
Flow(exp)	18.76	18.76	19.5	 ~:
Flow(meas)	18.4	18.3	16.3	_:

Speed 3

SG(meas) = 1.5906

LC sup(mV)	229.4	229.4	231.2	231.9	;
LC full(mv)	351.3	358.7	353.3	356.7	
Time (sec)	14	15	14	14	
Flw(exp).SG	50.13	49.64	50.42	51.54	
Flow(exp)	31.51	31,33	31.69	32,40	<u> </u>
Flow(meas)	30.5	30,5	30.5	30.5	

# Appendix C. Mass Flow Evaluation Test Results.

SG(meas) = 1.5955

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	235.3 380.0 14 59.34	236.4 392.7 15 60.24	233.0 370.0_ 13 60.93	; ;	ĬĿ
Flow(exp) Flow(meas)	37.19 38.5	37.75 38.4	38.18 38.47	=:=	

#### Speed 5

SG(meas) ≈ 1.605

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	244.8 475.4 19 70.17	242.0 411.9 14 70.16	243.4 429.5 16 67.25		
Flow(exp) Flow(meas)	43.71 43.0	43.71 43.2	41.90 43.2	<u> </u>	=:=

Speed 6

SG(meas) = 1.6028

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	243.5 414.6 14 70.66	241.8 414.7 15 55.64	241.5 447.4 17 70.03		` `-
Flow(exp) Flow(meas)	44.02 43.3	41.52 43.3	43.63 43.1	43.1	==

Appendix C. Mass Flow Evaluation Test Results.

# TEST 6

Speed 1

SG(meas) ≈ 1.5245

LC emp(mV)	219.1	222.4	302.2	228.3	
LC full(mV)	304.4	313.0	376.8	327.1	
Time (sec)	18	19	16	22	
Flw(exp).SG	27.40	27.57	26.96	27.02	
Flow(exp) Flow(meas)	17.97 17.4	18.08 17.4	17.68 17.4	17.72	z:=

#### Speed 2

SG(meas) = 1.5343

LC emp(mV)	225.7	233.8	227.3	238.3	
LC full(mv)	364.5	359.2	357.5	377.8	
Time (sec)	21	19	20	21	
Flw(exp).SG	38.21	38.16	37.64	38.41	
Flow(exp)	24.90	24.87	24.53	25.03	<u> </u>
Flow(meas)	24.4	24.4	24.4	24.4	

# Speed 3

SG(meas) = 1.539

LC emp(mV)	234.9	228.9	230.2	232.7	
LC full(mV)	415.1	427.2	412.6	420.1	
Time (sec)	21	24	23	22	
Flw(exp).SG	49.61	47,77	45.85	49.25	
Flow(exp)	32.23	31,03	29.79	32.00	_:_
Flow(meas)	26.5	28.6	28.6	28.8	

# Appendix C. Mass Flow Evaluation Test Results.

# SG(meas) = 1.5612

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	233.6 428.8 20 56.43	234.1 422.6 19 57.36	233.8 437.9 21 56.19	;- ;-	; 
Flow(exp) Flow(meas)	36.14 34.5	36.74 24.6	35.99 34.4	34.1	_:_

# Speed 5

SG(meas) = 1.5686

LC emp(mV) LC full(mv) Time (sec) Flw(exp).SG	238.4 412.8 16 63.0	239.8 431.2 17 65.09	242.1 432.4 17 64.72		; ;
Flow(exp) Flow(meas)	40.16 40.2	41.49 40.2	41.25 40.24	_:_	_:_

Speed 6

SG(meas) = 1.5759

LC emp(nV) Lo full(mv) Time (sec) Flw(exp).SG	236.2 422.4 16 67.28	242.0 441.3 17 67.78	237.9 421.6 16 66.38		`;_ ;_
Flow(exp) Flow(meas)	42.69 41.2	43.01 41.18	42.12 41.28	40.9	_:_

# Appendix C. Mass Flow Evaluation Test Results.

# WATER TEST

# Speed 1

LC emp(mV) LC full(mv) Time (sec)	199.4 272.0 30	200.4 284.2 32	192.9 290.3 42	:-	:_
Flow(exp) Flow(meas)	13.99 14.19	15.14 `	13.4 	_:_	_:_

#### Speed 2

LC emp(mV) LC full(mv) Time (sec)	187.0 295.9 30	218.1 316.9 28	193.2 315.4 34	 ;
Flow(exp) Flow(meas)	20.98 19.96	20.4	20.77	 Ξ:Ξ

# Speed 3

LC emp(mV) LC full(mv) Time (sec)	193.0 347.3 32	195,3 347,9 32		<u> </u>	:-
Flow(exp) Flow(meas)	27.87 21.25	27.57	_:_	_:_	_:_

# Appendix C. Mass Flow Evaluation Test Results.

LC emp(mV) LC full(mv) Time (sec)	198.5 374.1 30	195.1 379.0 32	``````````````````	;-	;;; ;;;
Flow(exp) Flow(meas)	33.84 33.61	33.22 `	Ë	=:=	:_

#### Speed 5

LC emp(mV) LC full(mv) Time (sec)	205.3 401.4 30	197.9 315.7 18	:=	:-	
Flow(exp) Flow(meas)	37.79 37.86	37.83	_:_	_:_	` `

# Appendix C. Mass Flow Evaluation Test Results.



Dry Solids SG = 2.65

DH80 settings: Zero = 2.279; Range = 2.581; Isotope = 0

$$SG_{slurry} = 1/(1 - C_W + C_W/SG_{dry solids})$$

Where

SG<sub>slurry</sub> ≈ Slurry SG SG<sub>dry solids</sub> = Dry solids SG C<sub>W</sub> = Concentration by weight

= (sample dry mass)/(sample wet mass)

Appendix C. Mass Flow Evaluation Test Results.

r	e		T	· 1
z,	Ð	ø		

Pan Mass(g)	3.3	3.2	3.3	165.3	167.9
SAM wet (g)	130,9	124.7	160.1	1360.9	1225.6
SAM dry (g)	103.3	99.3	127.1	1111.1	1004.6
Cw (%)	78.4	79.1	78.95	79.1	79.1
SG (exp)	1.953	1.97	1.9669	1.97	1.97
SG (meas)	1.958	1.957	1.962	1.958	1.962

TEST 2

Pan Mass(g) SAM wet (g) SAM dry (g) Cw (%)	3.2 127.1 96.7 75.5	183.8 139.5 75.4	3.2 127.1 96.7 75.5		
SG (exp) SG (meas)	1.8863 1.8897	1.8858 1.8872	1.8863 1.8897	<u>;</u>	

#### TEST 3

Pan Mass(g)	3.4	3.5	3.3	3.3	3.4
SAM wet (g)	85.2	111.4	105.4	122.1	106.9
SAM dry (g)	60.1	79.1	74.1	86.7	75.7
Cw (%)	69.3	70.1	69.34	70.2	69.85
SG (exp)	1.7592	1.774	1.7598	1.7765	1.7697
SG (meas)	1.775	1.775	1.779	1.779	1.7794

### TEST 4

 Pan Mass(g) SAM wet (g) SAM dzy (g) Cw (%)	3.2 79.7 51.4 63.0	3.2 72.3 47.8 64.64		
 SG (exp) SG (meas)	1.6455 1.6642	1.6735 1.6764	 -:	:=

Appendix C. Mass Flow Evaluation Test Results.

TE	ST	-5

Pan Mass(g) SAM wet (g) SAM dry (g) Cw (%)	3.3 53.4 32.4 58.08	171.4 1007.0 657.3 58.13	3.2 46.5 28.3 57.97	
SG (exp)	1.5665	1.567	1.565	 1.57
SG (meas)	1.5686	1.5661	1.5686	1.5661

TEST 6

Pan Mass(g)	3.3	172.0	169.9	166.4	
SAM wet (g)	44.5	242.7	966.8	1087.0	
SAM dry (g)	26.4	211.4	604.5	713.5	
Cw (%)	56.07	55.73	54.54	59.43	
SG (exp)	1.5363	1.5311	1.5142	1.5874	
SG (meas)	1.5245	1.539	1.5245	1.5784	

Appendix C. Mass Flow Evaluation Test Results.

#### REFERENCES

Baker, R.C. and Hemp, J. (1979) A Review of Concentration Meters for Granular Slurries, <u>Proceedings of</u> <u>Hydrotransport 6</u>, Paper B1, Sept. 1979, KENT, UK.

Bateman, H. (1910) Proc. Cambridge Phil. Soc. 15:423.

Blight, G.E., More o' Ferral, R.C. and Avalla, D.C. (1979) Cemented Tailings Fill for Mining Excavations.

C and I (1982) Using Gamma Rays to Measure Density, <u>Control and</u> <u>Instrumentation</u>, Vol.14, no.10, pp.53.

Chilton, A.SP. (1979) Nucl. Sci. Eng. 59:436

Crabtree, M. (1975) Industrial Applications of Moisture Content by Infragauging, <u>Pulse</u>, Mar. 1975, pp.25-28.

Evens, R.D. (1972) The Atomic Nucleus, New York: McGraw-Hill.

Franzen, W. and Cochran, L.W. (1962) <u>Pulse lonization Chambers</u> and Proportional Counters, Nuclear <u>Instruments and their Uses</u>, New York: Wiley.

Gumpert, R. and Pakulis, I. (1981) Three Moisture Messurement Techniques : A Comparison, <u>Intech</u>, Vol.28, no.8, pp.43. Hubbell, J.H. (1969) "Photon Cross Sections, Attenuation Coefficients, and Energy Absorption Coefficients from 10 keV to 100 GeV", NSRDB-NDS 29, Aug.1969

KENT (1985) Magnetic Flowmeter -An integrated package, <u>SA Measure-</u> ment and Control, Vol. 2, no. 5, pp.46-48.

Kowalski, E. (1970) <u>Nuclear Elec-</u> tronics, New York: Springer-Verlag.

Krohne (1984) DH80 Radiometric Density Meter, <u>Installation</u> and <u>Operating</u> <u>Instruction</u> <u>Manual</u>, Krohne.

Lederer, C.M. and Shirley, V.S. (eds.) (1978) <u>Table of Isotopes.</u> 7th ed. New York: Wiley-Interscience Publication.

Marks, B. (1977) The Evolution of the Magnetic Flowmeter, <u>Pulse</u>, Nov. 1977.

MCI 1984 Nodel 101 Digital Density Gauga, <u>Operators Manual</u>, Mineral Control Instrumentation Pty. Ltd., Australia.

Merrit, R. (1982) Flow Monitoring Updata, <u>Instruments and Control</u> Systems, Vol.55, no.2, pp.43.

Meyer, W. and Schilz, W. (1984) Microwave Measurement of Moisture, <u>Pulse</u>, Jan. 1984, pp.27-31.

Micro Motion (1983) Mass Flow Mater Remote Electronics Instruction Manual, <u>Micro Motion Technical</u> Manual

#### REFERENCES



Mintek (1983) The Use of Doppler Flowmeters for Measuring Slurry Flows, <u>Pulse</u>, Feb. 1983, pp.21-22.

Muller, P.H. (1983) Measuring Specific Gravity with DP Transmitters, <u>Pulsa</u>, Nov. 1983, pp.19.

Ormrod, G.T.W, (1984) "The Suitability of Doppler Flowmeters for use in the Minerals Processing Industry", <u>Mintek Report No. M68</u>.

PBI (1983) <u>Ionizing Radiations</u>, Paper and Board Industry (UK) Advisory Commíttes, Crown.

Roy, R.R. and Read, R.D. (1968) Interactions of Photons and Leptons with Matter, New York: Academic Press

Segre, E. (1968) <u>Nuclei and Parti-</u> cles, New Tork: W.A. Benjamin.

Slomiana, M. (1979) Using Differantial Preasure Sensors for Level, Density, Interface and Viscosity, <u>Intech</u>, Vol.26, no.9, pp.63-68. Springer, K. (1979) Density Meauurement using Gemma Radiation -Theory and Application, <u>The South African Mechanical Engineer</u>, Vol.29, no.12, pp.443-452.

Springer, K. (1980) The Electromagnetic Flowmeter - Its Theory and Application, <u>The South</u> <u>African Mechanical Engineer</u>, Vol.30, no.3, pp.66-83.

Tittle, C.W. (1964) How to Apply Statistics to Nuclear Neasurements, <u>Muclear-Chicago Technical</u> <u>Bulletin</u>, no.14.

Tsoulfanidis, N. (1983) <u>Measure-</u> ment and Detection of Radiation, New York: McGraw-Hill.

Weil Electronics. (1983) Nuclear Moisture Gauge, <u>Pulse</u>, Jan. 1983, pp.87.

Williams, J. (1979) Tips on Nuclear Sauging, <u>Instruments and Control</u> systems, Vol.52, no.1, pp.43.

REFERENCES

Author Clegg Michael Petrus Name of thesis Development Of A Slurry Monitoring System. 1985

**PUBLISHER:** University of the Witwatersrand, Johannesburg ©2013

# LEGAL NOTICES:

**Copyright Notice:** All materials on the University of the Witwatersrand, Johannesburg Library website are protected by South African copyright law and may not be distributed, transmitted, displayed, or otherwise published in any format, without the prior written permission of the copyright owner.

**Disclaimer and Terms of Use:** Provided that you maintain all copyright and other notices contained therein, you may download material (one machine readable copy and one print copy per page) for your personal and/or educational non-commercial use only.

The University of the Witwatersrand, Johannesburg, is not responsible for any errors or omissions and excludes any and all liability for any errors in or omissions from the information on the Library website.